

A Compound Novel Data-Driven and Reliability-Based Predictive Maintenance Framework for Ship Machinery Systems

Authored by Michail Fragkiskos Cheliotis

A thesis presented in fulfilment of the requirements for the degree of Doctor of Philosophy

Department of Naval Architecture, Ocean and Marine Engineering University of Strathclyde

Glasgow, UK

2020

Declaration

This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination which has led to the award of a degree.

The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyright Acts as qualified by University of Strathclyde Regulation 3.50. Due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

Signed: Michail Fragkiskos Cheliotis Date: 12/10/2020

"Don't try"

Charles Bukowski (1920 - 1994)

Acknowledgements

I would like to express my gratitude and appreciation to Dr. Iraklis Lazakis for supervising and funding this research study. I would also like to thank Dr. Gerasimos Theotokatos from the University of Strathclyde Maritime Safety and Research Center for also supervising this work and providing valuable input. Special thanks must be directed to INCASS EU FP7 Project and ISEMMS Innovate UK Project. Special thanks also go to all my colleagues Dr Konstantinos Dikis and Christos Gkerekos for their input, suggestions and support.

I would also like to formally acknowledge the support of my family, Angelos, Laya, Christina, my partner Konstantina and my friends Tasos, Petro, Eva, Giorgos, Marialena, Dimitra, Olga, Katerina and Andreas. I could not have completed this thesis without their support.

To friends and family, made, kept and lost

Research Output

Journal Papers

Michail Cheliotis, Iraklis Lazakis, and Gerasimos Theotokatos (Nov. 2020). "Machine learning and data-driven fault detection for ship systems operations". In: *Ocean Engineering* 216, p. 107968. ISSN: 00298018. DOI: 10.1016/j.oceaneng. 2020.107968. URL: https://linkinghub.elsevier.com/retrieve/pii/ S0029801820309203

Michail Cheliotis, Iraklis Lazaks and Angelos Cheliotis. (Dec. 2019). "Developing a Combined Bayesian and Machine Learning-Based Diagnostic Framework for Ship Systems Applications". In: *Ships and Offshore Structures*. **Under review**

Michail Cheliotis et al. (Sept. 2019). "A novel data condition and performance hybrid imputation method for energy efficient operations of marine systems". In: *Ocean Engineering* 188. ISSN: 00298018. DOI: 10.1016/j.oceaneng.2019. 106220

Conference Papers

Michail Cheliotis and Iraklis Lazakis (2018). "Ship machinery fuzzy condition based maintenance". In: *Smart Ship Technology 2018* Ed. by RINA. London: Royal Institute of Naval Architects. ISBN: 9781909024632

Michail Cheliotis and Iraklis Lazakis (2015). "Assessing key reliability and criticality aspects of ship systems for improved energy efficiency". In: *SSC 2015: Shipping in Changing Climates Conference 2015.* Glasgow: University of Strathclyde

Michail Cheliotis, Eirini Tryvyza, et al. (2016). "Project AUTONOMAD : The design of an unmanned coast guard vessel integrating the use of drones for emission regulation enforcement and territorial water protection in the Mediterranean". In: *MSO 2016: Maritime Safety and Operations.* Glasgow: University of Strathclyde

Invited Presentations

Michail Cheliotis (2017). "Maintenance and reliability assessment for enhanced safety and energy efficiency". In: *University of Strathclyde Annual SNAME Symposium*. Glasgow: University of Strathclyde

Michail Cheliotis (2018). "Ship Machinery Fuzzy Condition Based Maintenance". In: University of Strathclyde Annual SNAME Symposium. Glasgow: University of Strathclyde

Michail Cheliotis (2019). "Making Sense of Data for Maintenance in the Maritime Industry". In: University of Strathclyde Annual SNAME Symposium. Glasgow: University of Strathclyde

Contents

D	eclar	ation							i
\mathbf{A}	cknov	wledge	ements						iii
R	Research Output						iv		
\mathbf{Li}	st of	Figure	es					2	xvii
\mathbf{Li}	st of	Algor	ithms					x	viii
List of Tables					xxi				
\mathbf{A}	Acronyms xxi				xxii				
\mathbf{A}	bstra	nct						2	xxv
1	Intr	roducti	ion						1
	1.1	Chapt	er Overview		•			•	1
	1.2	Mariti	me Industry Basics		•			•	1
	1.3	Impac	t of Shipping		•			•	5
		1.3.1	Financial Impact		•			•	5
		1.3.2	Environmental Impact		•			•	6
		1.3.3	Safety Impact		•			•	6
	1.4	Resear	rch Direction						9

		1.4.1	Research Question	9
		1.4.2	Aim and Objectives	9
	1.5	Shippi	ing 4.0	10
	1.6	Mariti	ime Maintenance Regulatory and Industrial Framework	12
		1.6.1	The International Maritime Organisation (IMO)	12
			1.6.1.1 The International Safety Management (ISM) Code	12
			1.6.1.2 The Safety Management System (SMS)	13
		1.6.2	Classification Societies	14
		1.6.3	The Tanker Management Self Assessment (TMSA) $\ . \ . \ .$	15
	1.7	Thesis	S Layout	15
	1.8	Chapt	er Summary	17
2	Lite	erature	e Review	18
	2.1	Chapt	er Overview	18
	2.2	Maint	enance Introduction	19
	2.3	Maint	enance Concepts	20
		2.3.1	Corrective Maintenance	21
		2.3.2	Preventive Maintenance	23
		2.3.3	Predictive Maintenance	25
	2.4	Maint	enance Frameworks	27
		2.4.1	Total Productive Maintenance	27
		2.4.2	Business Centered Maintenance	29
		2.4.3	Reliability Centered Maintenance	30
		2.4.4	Condition Based Maintenance	32
		2.4.5	Comparison of Frameworks	36
	2.5	Predic	etive Maintenance Tools	38
		2.5.1	Reliability Assessment (RA) Tools	38
			2.5.1.1 Fault Tree Analysis (FTA)	39

			2.5.1.2	Bayesian Networks (BN)	42
			2.5.1.3	Failure Modes Event Analysis (FMEA) and Fail-	
				ure Modes Event and Criticality Analysis (FMECA)	44
			2.5.1.4	Event Tree Analysis (ETA)	46
			2.5.1.5	Comparison of RA tools	47
		2.5.2	Data Sci	ence (DS) Tools	50
			2.5.2.1	Supervised Learning	51
			2.5.2.2	Unsupervised learning	53
	2.6	Predic	tive Main	tenance Processes	54
		2.6.1	Critical	Equipment	54
		2.6.2	Data Pre	eparation	57
			2.6.2.1	Outlier Detection	57
			2.6.2.2	Imputation	59
		2.6.3	Fault De	etection (FD)	62
			2.6.3.1	FD Discussion	65
		2.6.4	Diagnost	ics	65
			2.6.4.1	Diagnostics Discussion	68
	2.7	Mariti	me Predio	ctive Maintenance	69
	2.8	Identif	ied Gaps		73
	2.9	Chapt	er Summa	ary	77
2	Mot	hodol			70
J	2 1	Chapt	or Overvi		70
	0.1 2.0	Novelt		τω	79
	0.2 2.2	Overvi	y iony of the	Noval Framowork	79 91
	0.0 24	Critics	lew of the	novel Flamework	80 01
	J.4	2/1			04 83
		0.4.1 249	Data Ca		00 85
		0.4.2	Data U0		00

	a 4 a	
	3.4.3	Ship Systems Analysis
	3.4.4	Fault Tree Analysis (FTA) 86
		3.4.4.1 Examined Gates and Events
		3.4.4.2 Qualitative Analysis
		3.4.4.3 Quantitative Analysis
		3.4.4.4 Importance Measures (IMs)
	3.4.5	Clustering Analysis
3.5	Data I	Preparation
	3.5.1	Overview
	3.5.2	Data Collection
	3.5.3	Preliminary Analysis
	3.5.4	Imputation Process
	3.5.5	Operational Analysis
3.6	Fault	Detection (FD) $\ldots \ldots 107$
	3.6.1	Overview
	3.6.2	Data Collection
	3.6.3	Data Checking
	3.6.4	Model Development
		3.6.4.1 Training and Validation
		3.6.4.2 Testing
	3.6.5	Fault Detection
3.7	Diagno	ostics $\ldots \ldots 121$
	3.7.1	Overview
	3.7.2	Data Collection
	3.7.3	Diagnostic Set-Up
		3.7.3.1 Fault Mapping 124
		3.7.3.2 FD Results Aggregation
		3.7.3.3 Network Set-up
		-

	3.8	Chapt	er Summa	ary	129
4	Cas	e Stud	lies		131
	4.1	Chapt	er Overvi	ew	131
	4.2	Critic	al Equipn	nent Selection	132
		4.2.1	Data Co	llection	133
		4.2.2	Ship Sys	stem Analysis	137
			4.2.2.1	Essential Machinery	138
			4.2.2.2	Fuel	139
			4.2.2.3	Lubricating Oil (LO)	139
			4.2.2.4	Cooling	139
			4.2.2.5	Various	140
		4.2.3	Fault Tr	ee Analysis	140
	4.3	Data I	Preparatio	on	157
		4.3.1	Data Co	llection	157
	4.4	Fault	Detection	(FD) and Diagnostics	160
		4.4.1	Data Co	llection	162
			4.4.1.1	Historic Process Monitoring (ProMon) Data	162
			4.4.1.2	Incoming Process Monitoring (ProMon) Data	163
			4.4.1.3	Main Engine (ME) Shop Tests	165
			4.4.1.4	Data Banks	166
		4.4.2	Chapter	Summary	166
5	Cas	e Stud	lies Resu	llts	168
	5.1	Chapt	er Overvi	ew	168
	5.2	Critic	al Equipn	nent Selection	169
		5.2.1	Reliabili	ty Importance Measures (IM)	169
		5.2.2	Clusteri	ng Analysis	171
	5.3	Data 1	Preparatio	on	177

	A.2	FTA E	Events .		•		312
	A.1	FTA (Gates .				306
A	FTA	A Gate	$es and E^{\gamma}$	vents			306
	6.7	Chapt	er Summ	ary	•	•	255
	6.6	Assum	ptions ar	nd Challenges	•		254
	6.5	Recom	nmendatio	on for future work	•	•	251
	6.4	Reflect	tions		•	•	249
	6.3	Genera	ated Nove	elty	•	•	246
	6.2	Fulfilm	nent of Ai	im and Objectives	•	•	239
	6.1	Chapt	er Overvi	ew	•	•	239
6	Disc	cussion	and Co	onclusions		4	239
	5.7	Chapt	er Summa	ary	•	•	200
	5.0	Chapt	or Summ		•	•	204
	56	Diceus	0.0.1.0	network Set-Op	•	•	220
			5.5.1.2	FD Results Aggregation Network Set Up	•	•	219
			0.0.1.1 5 5 1 0	Fault Mapping	•	•	213
		0.0.1		The Set-up	•	•	212
	0.0		D'	· · · · · · · · · · · · · · · · · · ·	•	•	212
		5.4.3 D'	Fault De	etection	•	•	205
			5.4.2.2	Testing	•	•	204
			5.4.2.1	Training and Validation	•	•	197
		5.4.2	Model E	Development	•	•	195
		5.4.1	Data Ch	necking	•	•	192
	5.4	Fault 1	Detection	$(FD) \dots \dots$	•	•	192
		5.3.3	Operatio	onal Analysis	•	•	191
		5.3.2	Imputat	ion Process	•	•	179
		5.3.1	Prelimin	nary Analysis	•	•	177

В	Ship Components Cost	314
	B.1 Costs	314
С	Critical Equipment Selection	320
	C.1 MTBFs	320
	C.2 IMs	324
D	Fault Detection	335
	D.1 Data Checking Results	335

List of Figures

1.1	Overview of the world fleet by ship type, adopted from (UNCTAD	
	2018b)	3
1.2	Overview of the remaining ship types of the world fleet, adopted	
	from (UNCTAD 2018b)	3
1.3	Bulk carriers age distribution, adopted from (UNCTAD 2018b)	4
1.4	Oil tankers age distribution, adopted from (UNCTAD 2018b)	4
1.5	Equipment failure, per ship type, leading into accidents percent-	
	age. Adopted from (EMSA 2018)	8
2.1	Basic maintenance concepts	21
2.2	Maintenance frameworks qualitative comparison	37
2.3	Representation of a basic fault tree structure	41
2.4	Representation of a basic Bayesian Network	43
2.5	RA tools qualitative comparison	48
2.6	Initial division of diagnostic models	66
2.7	Further division of diagnostic models based on the specific approach	66
3.1	Layout of the novel work, depicting the different methodologies,	
	the required inputs and the derived outputs of each part	83
3.2	Layout of the critical components selection methodology of the	
	proposed framework	85

3.3	Depiction of the FT structure, as created from the ship systems	
	analysis	87
3.4	Layout of the data preparation methodology of the proposed frame-	
	work, demonstrating the main steps and process. \ldots	97
3.5	Layout of the fault detection methodology of the proposed frame-	
	work presenting the main steps and process	109
3.6	Working process of the model development step, demonstrating the	
	different process, including the selection of the different predictor	
	variables.	114
3.7	Training and validation process using k -fold cross validation, with	
	k = 3 folds	117
3.8	Layout of the diagnostic methodology of the proposed framework	
	showing the different steps and demonstrating the input from the	
	FD methodology	123
3.9	General diagnostic network structure	126
3.10	sample of a basic BN	127
4 1		190
4.1	Identified systems for the studied vessel	138
4.2	F'I'A representation of the systems of vessel "A"	141
4.3	FTA representation of the essential machinery group	142
4.4	FTA representation of the remaining machinery groups	142
4.5	FTA representation of steering gear system	143
4.6	FTA representation of one of the four MGEs	144
4.7	FTA representation of the steam generation system $\ . \ . \ . \ .$	145
4.8	FTA representation of the IGG system	146
4.9	FTA representation of the cargo equipment	147
4.10	FTA representation of the bilge, fire and ballast systems $\ . \ . \ .$	148
4.11	FTA representation of the cooling system and the two sub-systems	148
4.12	FTA representation of the auxiliary cooling systems	149

4.13	FTA representation of the central MGE cooling system	150
4.14	FTA representation of the LO systems	151
4.15	FTA representation of the LO purification systems	152
4.16	FTA representation of the LO services system	153
4.17	FTA representation of the fuel systems	153
4.18	FTA representation of the gas fuel system	154
4.19	FTA representation of the MGE FO system	155
4.20	FTA representation of the FO purification systems	156
4.21	FTA representation of the FO transfer system	156
4.22	Diagram of a Main Engine (ME) system showing the physical in-	
	terconnections between the measured parameters (the compressor	
	is represented with C and the turbine with T) $\ . \ . \ . \ .$	158
4.23	Diagram of a Main Engine (ME) system showing the physical in-	
	terconnections between the measured parameters (the compressor	
	is represented with C and the turbine with T) $\ . \ . \ . \ .$	161
5.1	Three dimensional scatter plot showing the initial data-space (left)	
	and the results of the k -mean algorithm (right)	173
5.2	Three dimensional scatter plot showing the least critical compo-	
	nents (left) and the most critical components (right). \ldots .	174
5.3	Bar chart showing the I^B of the identified critical equipment	175
5.4	Bar chart showing the ${\cal I}^{CR}$ of the identified critical equipment	176
5.5	Distance of critical equipment to the origin of the axes \ldots .	176
5.6	Heat-map showing the Pearson correlation coefficient of the ME	
	and TC variables	178
5.7	TC LO inlet pressure imputation performance comparing the MICE,	
	$k{\rm NN}$ and hybrid methods	181
5.8	TC LO outlet temperature imputation performance comparing the	
	MICE, k NN and hybrid methods	182

5.9	ME Power imputation performance comparing the MICE, kNN	
	and hybrid methods.	183
5.10	ME Speed imputation performance comparing the MICE, kNN	
	and hybrid methods.	184
5.11	ME scavenging air pressure imputation performance comparing the	
	ME, k NN and hybrid methods	185
5.12	TC EG inlet temperature imputation performance comparing the	
	MICE, k NN and hybrid methods	186
5.13	TC EG outlet temperature imputation performance comparing the	
	MICE, k NN and hybrid method	187
5.14	TC Speed imputation performance comparing the MICE, $k{\rm NN}$ and	
	hybrid methods	188
5.15	Summary plot of imputation approaches performance	190
5.16	Data checking result for ME shaft power (top) and ME shaft speed	
	(bottom)	194
5.17	(bottom)	194
5.17	(bottom)	194 199
5.17 5.18	(bottom)	194 199
5.17 5.18	(bottom)	194 199 201
5.175.185.19	(bottom)	194 199 201
5.175.185.19	(bottom)	194 199 201
5.175.185.19	(bottom)	 194 199 201 203
5.175.185.195.20	(bottom)	 194 199 201 203 204
 5.17 5.18 5.19 5.20 5.21 	(bottom)	 194 199 201 203 204 206
 5.17 5.18 5.19 5.20 5.21 5.22 	(bottom)	 194 199 201 203 204 206 208
 5.17 5.18 5.19 5.20 5.21 5.22 5.23 	(bottom)	 194 199 201 203 204 206 208 209
 5.17 5.18 5.19 5.20 5.21 5.22 5.23 5.24 	(bottom)	 194 199 201 203 204 206 208 209 210

5.26	Diagnostic test for the detection of AC air-side fouling	214
5.27	Diagnostic test the detection of AC water-side fouling	215
5.28	Diagnostic test for the detection of ME AF fouling	215
5.29	Diagnostic test for the detection of fouling in the mechanical com-	
	ponents of the TC	216
5.30	Diagnostic test for the fouling in the turbine or compressor of the	
	TC	216
5.31	EWMA control chart for Cylinder 1 during the simulated fault	220
5.32	EWMA control chart for Cylinder 2 during the simulated fault	221
5.33	EWMA control chart for Cylinder 3 during the simulated fault	222
5.34	EWMA control chart for Cylinder 4 during the simulated fault	223
5.35	EWMA control chart for Cylinder 5 during the simulated fault	224
5.36	Aggregated results during the simulated fault for the ME cylinders	225
5.37	Initial diagnostic BN	227
5.38	Diagnostic BN with with observable nodes at Failed state	228
5.39	Fully investigated diagnostic BN	229
5.40	Fault profiles for the primary faults	231
5.41	Average rate of change of primary faults	232
5.42	Fault profiles for the secondary faults	232
5.43	Average rate of change of secondary faults	233
D 1	Data checking result for ME scavenging air temperature	336
D.2	Data checking result for ME scavenging air pressure	337
D.3	Data checking result for ME cylinder 1 EG temperature	338
D.4	Data checking result for ME cylinder 2 EG temperature	339
D.5	Data checking result for ME cylinder 3 EG temperature	340
D.6	Data checking result for ME cylinder 4 EG temperature	341
D.7	Data checking result for ME cylinder 5 EG temperature	342

List of Algorithms

1	Data filtering working process	99
2	Hybrid novel imputation method using a combination of $k NN$ and	
	MICE	105
3	Model development and hyperparameter optimisation \ldots \ldots \ldots	118

List of Tables

2.1	Corrective maintenance advantages and disadvantages	23
2.2	Preventive maintenance advantages and disadvantages	25
2.3	Predictive maintenance advantages and disadvantages	27
2.4	Status quo and added contribution	77
4.1	Main features of vessel "A"	134
4.2	Sample of the PMS of vessel "A"	135
4.3	Table of repair costs of components shown in Table 4.2 \ldots .	137
4.4	Main features of vessel "B"	158
4.5	Descriptive statistics of the collected dataset used in the data	
	preparation methodology	159
4.6	Ship system parameter limits used in the data preparation method-	
	ology	160
4.7	Main features of vessel "C"	161
4.8	Information for the variables used in the FD and diagnostics method-	
	ologies	163
4.9	Descriptive statistics of the historic ProMon data used in the FD	
	and diagnostic methodologies	164
4.10	Descriptive statistics of the incoming ProMon data used in FD and	
	diagnostic methodologies	165
4.11	Shop test sample for the ME of vessel "C"	165

5.1	Table of MTBFs used for the bilge, fire and ballast systems	169
5.2	Sample of the obtained I^B for vessel "A"	170
5.3	Sample of the obtained I^{CR} for vessel "A"	171
5.4	Resulting correlation based on the integration of the data-driven	
	Pearson coefficient and the first-principle domain knowledge	179
5.5	Summary table of imputation approaches performance	189
5.6	List of examined multiple linear ridge regression models	196
5.7	List of examined multiple polynomial ridge regression models	197
5.8	List of examined single regression models	197
5.9	Performance of linear and polynomial models in terms of \mathbb{R}^2	199
5.10	Performance of linear and polynomial models in terms of σ	200
5.11	Performance of models with the highest \mathbb{R}^2 and performance of	
	models with lowest σ	202
5.12	Verification cases description	207
5.13	Mapped faults for increases in EG temperature in all cylinders,	
	adapted from (MAN B&W 20017)	213
5.14	Mapped faults for increase in EG temperature of one cylinder,	
	adapted from (MAN B&W 20017)	217
5.15	Failure modes summary for the primary faults	230
B.1	Repair and replacement costs for the components of vessel "A"	314
C.1	Table of MTBFs used for the inert gas generation system	320
C.2	Table of MTBFs used for the cargo equipment system	320
C.3	Table of MTBFs used for the bilge, fire and ballast systems	321
C.4	Table of MTBFs used for the steering gear system	321
C.5	Table of MTBFs used for the steam generation system	321
C.6	Table of MTBFs used for the main generating engines system	322
C.7	Table of MTBFs used for the gas fuel system	322

C.8	Table of MTBFs used for the fuel oil feed system	322
C.9	Table of MTBFs used for the fuel oil purification system $\ . \ . \ .$	322
C.10	Table of MTBFs used for the fuel oil transfer system $\ . \ . \ . \ .$	323
C.11	Table of MTBFs used for the lubricating oil purification system $% \mathcal{T}_{\mathrm{s}}$.	323
C.12	Table of MTBFs used for the lubricating oil service system	323
C.13	Table of MTBFs used for the central MGE cooling system	323
C.14	Table of MTBFs used for the auxiliary cooling system	323
C.15	Obtained I^B for the components of vessel "A"	324
C.16	Obtained I^{CR} for the components of vessel "A"	329

Acronyms

AI	Artificial Intelligence.
ANN	Artificial Neural Networks.
APE	Absolute Percentage Error.
BCM	Business Centered Maintenance.
BN	Bayesian Network.
BWTS	Ballast Water Treatment System.
CBM	Condition Based Maintenance.
CL	Center Line.
CM	Corrective Maintenance.
CUMSUM	Cumulative Sum.
DA	Data Analytics.
DAQ	Data Acquisition.
DBSCAN	Density-BasedSpatial Clustering of Applica-
	tions with Noise.
DS	Data Science.
EB	Expected Behaviour.
EU	European Union.
EWMA	Exponentially Weighted Moving Average.
FD	Fault Detection.
FMEA	Failure Modes Event Analysis.

FMECA	Failure Modes Event and Criticality Analysis.
FO	Fuel Oil.
\mathbf{FR}	Failure Rate.
FT	Fault Tree.
FTA	Fault Tree Analysis.
GDP	Gross Domestic Product.
GHG	Green House Gas.
HE	Hard Evidence.
IGG	Inert Gas Generator.
IM	Importance Measures.
IMO	International Maritime Organisation.
IoT	Internet of Things.
LCL	Lower Control Limit.
LNG	Liquefied Natural Gas.
LO	Lubricating Oil.
MAPE	Mean Absolute Percentage Error.
MCS	Minimal Cut Sets.
ME	Main Engine.
MGE	Main Generating Engine.
MICE	Multiple Imputation by Chained Equations.
ML	Machine Learning.
MRV	Monitoring Reporting and Verification.
MTBF	Mean Time Between Failures.
NN	Nearest Neighbours.
OEE	Overall Equipment Effectiveness.
OLS	Ordinary Least Squares.
OREDA	Offshore and Onshore Reliability Data.
PdM	Predictive Maintenance.

PM	Preventive Maintenance.
RA	Reliability Assessment.
RBD	Reliability Block Diagrams.
RCM	Reliability Centered Maintenance.
SVM	Support Vector Machine.
TC	Turbocharger.
TPM	Total Productive Maintenance.
UCL	Upper Control Limit.
VE	Virtual Evidence.

Abstract

Shipping is a major driving force of the global economy, as seen by the 90% of the volume of the yearly trade transported by ships. As a result, shipping has a significant financial, environmental impact. Maritime maintenance can be used to safeguard shipping's impact by improving safety and avoiding accidents. This is especially true, when considering that nearly 22% of all the accidents between 2011 and 2017 were attributed to improper maintenance. Consequently, maritime maintenance can be used as a hazard mitigation tool, improve ship safety by reducing accidents. Modern maritime maintenance is best applied through predictive maintenance schemes, which take advantage of the developments of Shipping 4.0. Under this scope, the goal of this thesis is the development of a compound novel data-driven and reliability-based predictive maintenance framework for ship machinery system. The novel framework tackles the areas of maritime predictive maintenance holistically by addressing the topics of critical equipment selection, data preparation, fault detection and diagnostics. Each of the framework's topics are developed in individual methodologies and assessed in unique case studies demonstrating their effectiveness in the respective tasks. Initially, the methodology for the critical equipment selection includes the novel combination of Fault Tree Analysis with data clustering for the identification of critical equipment, as applied in the case of an LNG Carrier. As a result, the most critical components are identified by taking into account reliability indices and repair costs for the considered components. Identifying critical components improves safety, as it focuses the maintenance efforts in items whose failures can have economic consequences and safety implications. Next, the methodology for the data preparation is developed, which includes the novel integration of the kNN and MICE algorithms for the imputation of missing data. Combining these two algorithms allows for the novel integration a data-driven approach with domain knowledge in a single imputation model. The imputation methodology is applied in the case of a Chemical Tanker, showcasing the effectiveness of the novel method against a pure MICE and pure kNN approach. The treatment of missing values can improve ship safety, as it safeguards information contained within datasets and leads to more accurate condition assessing models. Following that, a novel Fault Detection methodology is established based on Expected Behaviour models, using Machine Learning, and Exponentially Weighted Moving Average control charts. This methodology aims at detecting developing faults in their early stages while avoiding the shortcomings of black-box approaches and having reasonable data requirements for training. Lastly, the diagnostics methodology is formed, which includes the novel integration of pre-processing and Machine Learning-based Fault detection with a diagnostic network using Bayesian Networks. The resulting methodology can identify the root cause of a detected fault, without using black-box Neural Network approaches, nor complicated and time-consuming physics-based models. Even though the Fault Detection and diagnostics methodologies are developed individually, they are both evaluated in the same case of a Bulk Carrier. The use of the same case study was dictated by restrictions in collecting additional data and by the use of the output of the Fault Detection methodology in the diagnostics. The detection of developing faults and the identification of their root-cause has a profound effect on ship safety, while also allowing for targeted maintenance actions.

Chapter 1

Introduction

1.1 Chapter Overview

This chapter presents the essential background information required for the establishment of the present thesis. Also, this chapter presents the direction of this work in terms of the research question, the aim and objectives. A brief outline of the different chapters alongside with an indication of their content is also given.

1.2 Maritime Industry Basics

Shipping has been crucial to the development and evolution of the global trade and economy. It helps to transport people and commodities across the globe and allows for the facilitation of commerce between nations. Shipping can form the backbone of a nation's economy, as it can create numerous streams of revenue and employment. The performance of seaborne trade is a proxy of the performance of the global economy, as it tends to mimic the cycles of the world's Gross Domestic Product (GDP) (Stopford 2018; IUMI 2018).

Seaborne transport is one of the most cost-effective ways of carrying commodities, due to the economies of scale. Notably, for the last half-century, shipping has been carrying approximately 90% of the volume and 70% of the value of the global trade (UNCTAD 2018b). These values are even higher in developing countries, where air and land transport options are as common.(UNCTAD 2018b). During the relatively same period, the amount of cargo carried from shipping has increased by approximately 8'000 million tons (UNCTAD 2017). Keeping in line with the positive overview of shipping, its future remains bright as seen by the 2.6% and 4% volume increases in 2016 and 2017 respectively (UNCTAD 2017).

The structure of the world fleet can influence its overall performance and trends. Notably, at the beginning of 2018, the world fleet consisted of 94'171 ships, as a result of the 3.7% dead-weight increase in 2017. In terms of ship types, the world fleet consists predominantly of dry bulk carriers and oil tankers. In detail, in 2018 the world fleet comprised of 42.5% of dry bulk carriers, 29.2% of oil tankers, 13% container ships, 4% cargo ships and 11% of other types, as seen in Figure 1.1 (UNCTAD 2018b). Moreover, from the 11% of other ship types, 37% of them were offshore vessels, 29% gas carries (e.g. Liquid Natural Gas carriers) and 20% chemical tankers, as seen in Figure 1.2 (UNCTAD 2018b).

In terms of the vessels' age, in 2018 the average age of the world's fleet was 20 years, with that number being higher for developing countries. In general, the average age of the world fleet has been increasing since 2016 as a result of a decrease in the number of new ship deliveries (UNCTAD 2017). The prevailing age group for bulk carriers includes vessels aged between 5 and 9 years old, representing 41% of the bulk carrier fleet, according to Figure 1.3. Similarly, the prevailing age group for oil tanker includes vessels aged over 20 years old, representing 38% of the oil tanker fleet, according to Figure 1.4 (UNCTAD 2018b).

Considering the increased age of the predominant ship types, maintenance is becoming an ever more important issue for tankers and bulk carriers, in order to avoid failures and accidents. This principle also applies to the remaining fleet,



Figure 1.1: Overview of the world fleet by ship type, adopted from (UNCTAD 2018b)





considering its rising age and the already high average age of its vessels. In other words, maintenance can be used to prevent failures and reduce the likelihood of maintenance related accidents, especially for order vessels, as further discussed in the following sections. Under that scope, maintenance is required to ensure that



Figure 1.3: Bulk carriers age distribution, adopted from (UNCTAD 2018b)

Oil Tanker Age Distribution



Figure 1.4: Oil tankers age distribution, adopted from (UNCTAD 2018b)

the world fleet continuous its prosperous trends by maintaining its functionality and reliability (Zhang and Wang 2014).

The development of shipping has transformed developing countries, especially in Asia, to crucial parts of the global supply chain. Even though this has had a positive impact on the economy of these countries, it has also increased the global demand for reliable shipping operations. The same requirements for increased reliability in shipping have also been introduced by the evolution of the just-in-time economies of the developed world (UNCTAD 2018a). Therefore, maintenance can safeguard reliable shipping operations by reducing the likelihoods of accidents, which can disturb supply chains and sensitive just-in-time economies.

1.3 Impact of Shipping

1.3.1 Financial Impact

As demonstrated in the previous sections, the impact of shipping in the global economy is unquestionable. Notably, in 2015, the European Union (EU) shipping industry employed 640'000 people, with salaries above the European average. This translates to \in 57 billion direct contributions to the EU's GDP. When considering the indirect benefits, the EU shipping industry in 2015 contributed \in 140 billion to the EU's GDP (Oxford Economics Ltd 2017).

Proper ship maintenance safeguards the profit-making capabilities of vessels and improves their profitability. This is achieved by reducing the downtime caused by accidents related to improper maintenance. Between 2017 and 2018, the freight rates of the majority of the ship types increased, resulting in an average daily earning of \$10'986 in all segments of the fleet (UNCTAD 2018b). This positive behavior continued has continued into 2019, with bulk carriers' earnings reaching \$31'000 per day and oil tankers \$55'000 per day (one year time-charter) (Fearnleys Research 2019b; Fearnleys Research 2019a). This positive behaviour has even led some tankers to achieve a reported record-breaking \$300'000 per day during October of 2019 (Fearnleys Research 2019c). Therefore, the potentially avoidable downtime of vessels due to maintenance-related accidents during highly profitable periods could have negative implications on the financial performance of the respective shipping companies.

1.3.2 Environmental Impact

Apart from its economic importance, shipping can have a significant positive environmental impact. Ships in degraded physical conditions have an increased risk of causing environmental damages. When ships and their machinery are not appropriately maintained, their state degrades faster than the expected wearand-tear. Premature equipment degradation can have adverse effects, including sub-optimal operations which can lead to increased fuel consumption and emissions. Regarding emissions, the global shipping industry produces only 2.2% of the global Greenhouse Gas (GHG) emissions, even though it carries 90% of the world's trade. The emissions produced by maritime transport are much lower compared to other modes of transportation, include rail, air and road transport. Despite the low emissions levels, the shipping industry has set goals to further reduce its emissions by the implementation of various schemes. From a regulatory point of view, most of these efforts are embedded in the Maritime Pollution (MARPOL) Convention, established by the International Maritime Organisation (IMO). Under the same scope, the EU's Monitoring, Reporting and Verification (MRV) scheme aims to control the emissions of GHGs (UNFCCC 2014). To meet the required emissions levels and reduce shipping's impact on the environment, the adoption of adequate maintenance schemes that ensure the proper upkeep of the assets is encouraged (Duran, Uriondo, and Moreno-Gutiérrez 2012; Lindstad et al. 2015).

1.3.3 Safety Impact

Poor maintenance practices can undermine the safety of ships, which in turn increases the risk of accidents. Ship accidents can cause environmental and economic damage, as well as lead to the loss of life. The lack of proper maintenance can be a significant accident contributor, especially if it is combined with design

Ship Maintenance

flaws and human errors. The absence of apt maintenance can manifest itself in the form of engine trouble, propulsion trouble, mechanical issues, generators malfunctions, electrical problems, and plumbing failures. In detail, during the 2011-2017 period, a total of 20'616 casualties and incidents were recorded, which resulted in 683 fatalities in total (EMSA 2017). During this period, the second most common contributing factor to the recorded accidents is equipment failure relating to maintenance actions. In detail, cargo ships recorded around 297 fatalities corresponding to approximately 32 lost vessels. Since 20% of these accidents were caused by equipment failures relating to maintenance actions (Figure 1.5), approximately 59 people and 6 ships were lost due to improper maintenance. Likewise, fishing vessels recorded around 209 fatalities corresponding to approximately 111 lost vessels. As 45% of these accidents were caused by equipment failures relating to maintenance actions (Figure 1.5), approximately 94 people and 50 ships were lost due to improper maintenance. Also, passenger ships recorded around 97 fatalities corresponding to approximately 13 lost vessels. Since 30%of these accidents were caused by equipment failures relating to maintenance actions (Figure 1.5), approximately 29 people and 4 ships were lost due to improper maintenance. Lastly, service ships recorded around 50 fatalities corresponding to approximately 16 lost vessels. Since 25% of these accidents were caused by equipment failures relating to maintenance actions (Figure 1.5), approximately 13 people and 4 ships were lost due to improper maintenance (EMSA 2018).

For example, a high profile case of a maintenance-caused machinery failure resulting in an incident is the case of M/V Carnival Triumph. The vessel owned by Carnival Cruise Line, suffered from an engine room fire while sailing in the Gulf of Mexico, carrying 3140 passengers and 1100 crew. The fire caused a total loss of power, which lasted for several days, created numerous health hazards to the crew and passengers, and damaged the company's image. The initial fire was caused by an under-maintained flexible fuel oil pipe, while the failure to



Figure 1.5: Equipment failure, per ship type, leading into accidents percentage. Adopted from (EMSA 2018)

contain it was attributed to a combination of failures in the main fire hydrant and emergency generator (The Bahamas Maritime Authority 2013). Similarly, in 2015 the container ship M/V Maersk Gunde, owned by Maersk lines, suffered a fire in its engine room. The fire caused damages to the vessel and substantial delays. The fire was caused by a leak in an under-maintained seal of the fuel supply of one the generating engines (NTSB 2015).

Taking the above into account, it is apparent that maintenance can be used as a hazard mitigation tool, as proper maintenance can reduce the likelihood of maintenance related accidents. Well-maintained vessels can exhibit fewer failures in their systems, thus promoting safety and reducing risk. Lastly, proper maintenance can safeguard the profit-making capabilities of the vessels by reducing downtime and increasing availability.

1.4 Research Direction

Having established the impact of maintenance and shipping, this section sets the main direction of this research. It presents the main research question, which translates to the main aim. Lastly, this section sets the objectives required to meet the main aim.

1.4.1 Research Question

The research question of this thesis serves as the starting point for this work. It represents the main research drive and it influences all the facets of this research. This question can be expressed as following:

How can ship operators make real-time maintenance decisions based on the actual condition of specially selected equipment, by taking advantage of the developments in the fields of data analytics and machine learning and with the aim of improving safety?

1.4.2 Aim and Objectives

The main aim of this research is to answer the question detailed above, thought the development of a compound novel and data-driven reliability-based predictive maintenance framework for machinery systems. As a result, the main aim is to provide a complete solution ranging from the criticality evaluation of ship systems to diagnostic tasks. In order to achieve the stated aim, the following distinct objectives are set:

The creation will allow the answering of the research question, by providing

1. The investigation of the relevant literature regarding maintenance strategies, reliability assessment and data-drive predictive modeling in a factual
and critical manner, in order to identify gaps and direct the novelty of the present research.

- 2. The proposal of a novel, data-driven and reliability-based predictive maintenance framework, tailor-made for the needs of the maritime industry.
- 3. The development of a novel methodology for the identification of the critical equipment of ship systems, aimed at prioritising maintenance efforts.
- 4. The establishment of a novel data preparation methodology, specifically concerned with handling missing values from data sets used on condition monitoring tasks.
- 5. The development of a novel fault detection methodology, lessening the amount of data-associated assumptions, and tailored to the needs of the maritime industry.
- 6. The establishment of a diagnostic methodology combining in a novel way, machine learning applications with domain knowledge for practical applications of ship systems.
- 7. The demonstration and validation of the effectiveness of the proposed predictive maintenance framework through different case studies, such as the main engine of a bulk carrier.
- 8. The discussion of the main outcomes of the developed framework together with suggestions for future work.

1.5 Shipping 4.0

The shipping industry is transforming under the effects of state-of-the-art technology. The applications of modern Data Analytics (DA), Artificial Intelligence (AI), Machine Learning (ML), cloud computing, big data analysis, cybersecurity, and the Internet of Things (IoT) are pushing the shipping industry into modernity, namely into the era of Shipping 4.0 (Lambrou 2017).

The application of these tools is transforming and affecting many of the industry's practices, as it increases the use of automation in practices traditionally relying on empirical knowledge and expert know-how. One of the most prominent examples is the push of the maritime industry for unmanned and autonomous vessels. Data-driven Fault Detection (FD), smart diagnostics, reliability and criticality evaluation, and real-time route optimisation are some of the prerequisites for unmanned and autonomous shipping (Kvamstad-lervold 2017).

Apart from autonomous shipping, another area that is pushing the maritime industry to the 4.0 era is the advancement towards digital twins. Digital twins can be used to create digital replicas of ships, fully modelling the physical and functional characteristics. Digital twins are used in every stage of a ship's lifecycle and can help to achieve optimal ship operations and ship maintenance. Digital twins rely on the real-time collection of performance and condition data and their subsequent analysis in various models (Bradley and Hehenberger 2016).

Despite the changes brought upon by the digital revolution, the shipping industry is adopting the new challenges with a rate higher than its average. There is a rapidly expanding literature providing data-driven and ML-based solutions on many different issues ranging from fault detection and diagnostic to marine data pre-processing, both from individual research groups and larger research consortia.

1.6 Maritime Maintenance Regulatory and Industrial Framework

Because of its importance, ship maintenance is a heavily regulated area. There exists a plethora of regulations, guidelines and regulatory bodies both at national and international levels. Ship maintenance is concerned with the performance and condition of shipboard machinery, as well as with the ship hull and its appendages. All the relevant regulations aim to set the minimum requirements and create a basic structure for the implementation of maintenance.

1.6.1 The International Maritime Organisation (IMO)

The main regulatory body of the maritime industry is the International Maritime Organisation (IMO). IMO was established in 1948 by the United Nations (UN) with the task of improving safety at sea by developing guidelines, regulations and treaties (IMO 2013). For that reason, IMO has developed regulations and codes that recognise maintenance as a hazard mitigation tool and provide information on how it should be addressed.

1.6.1.1 The International Safety Management (ISM) Code

To tackle the issue of safety in a complete way, IMO adopted the International Safety Management (ISM) Code in 1993. The ISM Code became mandatory in 1998 and provides guidelines to increase the safety profile of the maritime industry. The ISM Code is conceptualised to provide international standards for the safety of ships with respect to the environment and human life. It provides standards for the management and operation of vessels. The ISM code is developed by taking into account that no two shipping companies operate and manage their assets in the same way. Therefore, the code is expressed in broad terms, and it provides a scope for shipping companies to adopt their own Safety Management System (SMS). The SMS is an extensive document that each shipping company develops based on the ISM Code. As a result, each company has its SMS.

The ISM Code is divided into 13 sections, each of which covers a different aspect of ship management and operation. A particular section of interest is section 10, which is concerned with the maintenance of the ship and its equipment. In detail, it is specified that it is the responsibility of the shipping company to make sure that their assets are maintained according to relevant rules and regulations. The maintenance should be carried out in a structured and planned way by appropriate personnel while having safety as a priority (IMO 2015).

1.6.1.2 The Safety Management System (SMS)

As previously mentioned, each company must create its SMS in compliance with the ISM Code. The SMS is a document developed by a shipping company that deals with the safety of operations in a holistic view. Each chapter of the SMS corresponds to a chapter in the ISM Code. Consequently, chapter 10 of an SMS deals with the maintenance of the ship and its equipment. The company's objectives are embedded on its SMS, and it has to institute measures to ensure that its ships are maintained in conformity with classification and statutory requirements. It is the company's responsibility to implement the most cost-effective maintenance scheme while keeping high maintenance standards.

Chapter 10 contains the appropriate information that links the safety of operations with maintenance. More specifically, the SMS addresses the maintenance strategy, maintenance management, planned maintenance schedule and maintenance actions of ship systems including the lifesaving appliances, electrical and electronic devices and engine room equipment (IMO 2015).

1.6.2 Classification Societies

Apart from IMO's ISM code, there are industry bodies that regulate the issue maritime of maintenance. Classification societies, like the American Bureau of Shipping (ABS), Lloyd's Register (LR) and Bureau Veritas (BV) together with the International Association of Classification Societies (IACS) are bodies which produce regulations, guidelines and recommendations for maritime maintenance. The purpose of the classification societies, and consequently IACS, is to confirm the reliability of ship systems and to verify that systems and components are maintained appropriately. Therefore, such organisations develop their own rules and standards. However, classification societies and IACS are not guarantors of safety, as they do not have control over how a ship is crewed and operated. IMO recognises the role of the class societies and their contribution to maritime safety through the International Convention for the Safety of Life at Sea (SOLAS) in 1988 (IACS 2015).

Over the years, IACS members have agreed on a set of Unified Requirements (URs) that each class society has to adapt on its own rules. URs set minimum standards for important topics that the rules of each class must cover. More specifically, an area of interest is the UR-Z which deals with issues ranging from survey requirements to PMS requirements (IACS 2017). Class surveys are essential, as their findings can help to schedule maintenance tasks. Additionally, surveys can be used as a tool to assess the effectiveness of the companies PMS. As a result, UR Z sets minimum survey periods that shipping companies must comply with (IACS 2017).

Apart from the URs, the different classification societies provide guidelines and recommendations for specialised topics, including ConMon, ProMon, PeMon and PdM. Indicatively, LR has published recommended procedures for the integration of ConMon and PdM in the design and construction of vessels (Lloyd's Register 2007). Similarly, ABS has available guidelines and recommendations for the selection of the appropriate ConMon and PdM techniques for ship systems (ABS 2016).

1.6.3 The Tanker Management Self Assessment (TMSA)

A major contributor to maritime safety is the use of various self-assessment schemes for ship operators. The Tanker Management Self Assessment (TMSA) is an example of such programs. The TMSA is tailor-made for tanker operators, and it was introduced by the Oil Companies International Marine Forum (OCIMF) in 2004. It aims to improve safety through the promotion of self-regulation and constant improvement among tanker operators. One of the TMSA's aims is to improve the reliability of ships and increase their safety. The TMSA provides a basic framework for the assessment of a ship operator's safety management systems. It is structured in 13 distinct elements, each of which deals with a specific topic. Of particular interest is the TMSA Element 4, which deals with issues surrounding reliability and maintenance standards. In detail, Element 4 describes the best practices that ship operators should adopt to increase the reliability of their assets, adhere to high maintenance standards and identify critical equipment (OCIMF 2004).

1.7 Thesis Layout

The presented thesis consists of seven individual main chapters. Each chapter is included in the following synopsis, along with a brief description of its content. In the following synopsis, along with the brief description of each chapter, the generated novelty of each chapter is also shown.

Chapter 1.

This chapter includes the essential background information required to establish

this thesis and provides a necessary introduction to the maritime industry. It establishes how shipping can affect the economy and the environment and explores the use of maintenance as a hazard mitigation tool. Moreover, the regulatory framework surrounding maritime maintenance is presented before concluding with some of the future challenges the industry will face.

Chapter 2.

Chapter 2 sets the main direction for the research presented in this thesis. It includes the main research question that prompted the developed work and sets out the main aim mapped-out through a series of clear objectives.

Chapter 3.

Chapter 3 includes a review of the relevant literature. This chapter aims to uncover gaps in the current literature so that the novelty of this work can be directed. For that reason, the review is not only factual, but also critical. Initially, the current practices and state-of-the-art regarding the identification and selection of critical equipment are examined. This leads to the conclusion that the status quo is lacking in the incorporation of cost elements of components, together with the traditional reliability indices. Afterwards, the literature regarding the preparation of data (pre-processing) for fault detection and other maritime applications is examined, resulting in the identification of a gap for tailor-made tools for maritime applications. Lastly, the bibliography of fault detection and diagnostic tools for engineering applications is examined, resulting in a profound gap for maritime applications.

Chapter 4.

This chapter details the proposed framework that was developed during this work. The proposed framework generates novelty by addressing the gaps that are uncovered during the critical review of the examined literature. In this chapter each methodological component of the framework is examined in detail in order to establish its theoretical background and intended functionality.

Chapter 5.

Chapter 5 describes the case studies that are used to apply the developed framework. This chapter provides the introductory information regarding the different case studies and presents and generalised outline.

Chapter 6.

Chapter 6 includes the results of the developed framework, as applied in several cases studies. The case studies demonstrate the effectiveness of the framework and its methodological components, to assure that the desired functionality is achieved.

Chapter 7.

This final chapter provides an in-depth summary of the key learning outcomes and conclusions of the submitted work. In this chapter, the manner in which the proposed novelty covers the identified gaps in the pertinent literature is examined. Furthermore, the shortcomings of this work are presented together with possible future research efforts required to address them.

1.8 Chapter Summary

In this chapter, the importance of shipping in terms of the global economy was established. Also, the use of maintenance as a hazard mitigation tool was examined. In addition, the research question, the aim and objectives were presented. Lastly, some of the challenges facing the maritime industry were presented before presenting an overall layout for the rest of the thesis.

Chapter 2

Literature Review

2.1 Chapter Overview

This chapter includes the examination of the relevant literature in terms of maintenance concepts, frameworks, predictive maintenance tools and predictive maintenance processes. The main aim is to perform a factual and critical review to uncover several gaps and orient the novelty of this work. To this end, this chapter gives a detailed explanation of all the aspects required to tackle the main research question, presented in Section 2.2. Initially, the different maintenance concepts and different maintenance frameworks are presented and subsequently, compared. Then, the tools that are required for predictive maintenance are presented into two categories, including Reliability Assessment (RA) tools and Data Science (DS) tools. Following that, the predictive maintenance processes that can be attained by using DS and RA tools are described. These processes include the identification of critical equipment, preparation of data, fault detection, and diagnostics and are further examined under the scope of the maritime industry. Lastly, by comparing the advancements of the maritime predictive maintenance processes with those from other industries, the existing gaps are identified. These gaps are then used to orient the novelty of this work.

2.2 Maintenance Introduction

Modern society is a complex functioning system comprising of many different components, each of which carries out a specific task. The condition of these components is directly related to the state of the broader system they belong to. This description applies to almost any physical system, including economic, communication and transportation systems. Another common link that all these systems share is the unavoidable failures, which may take different forms depending on the system. All systems are inherently unreliable to some extent, as all components suffer from age-based degradation, usage-based fatigue and other design flaws. Eventually, when a component is no longer able to perform its intended task, it fails.

The failure of a component can affect its wider system in many different ways. On the context of this work and according to the British Standards (BS) failure is defined as *"the termination of the ability of an item to perform a required function"* (BSI Standards Publication 2010). A failure can have little or no effect on the system's operation and safety, or it can trigger disastrous chain reactions that can even include loss of life (Kobbacy 2008).

Following the failure of a component, its wider system can be restored to its original state of operation by replacing or repairing the failed part. This process is known as corrective maintenance and forms the starting point for the development of more complex and elaborate maintenance frameworks. Components and systems failures can never be eliminated, due to their inherent unreliability and the plethora of operational and performance variables. Nonetheless, their occurrences can be controlled through the implementation of maintenance frameworks. These frameworks can include actions ranging from visual inspections to the use of advanced performance and condition monitoring tools. It should be noted that the maintenance needs of different systems may vary. Different systems have different maintenance requirements depending on their operational environment and the relevant regulatory framework. Choosing an appropriate maintenance scheme can reduce the likelihood and consequences of failures and breakdowns (Kobbacy 2008). According to the British Standards(BS) maintenance is defined as "the combination of all technical and administrative actions, including supervision actions, intended to retain an item in, or restore it to, a state in which it can perform a required action" (BSI Standards Publication 2010). In more detail, when maintenance is performed effectively, it can extend the lifespan of systems, sub-systems and components.

Currently, the maintenance practices in many sectors, including the maritime, are rapidly advancing. Maintenance is currently adopting practices from the fields of modern data analytics and AI. Recognising the above, the submitted work examines maintenance as an ensemble of different activities. Therefore, in this work maintenance is not only limited to upkeeping tasks, but is also combined with the identification of critical components, the processing of relevant data and the early identification of developing faults.

During the review of the relevant literature, the maintenance concepts and frameworks are presented first, followed by the predictive maintenance tools and processes. Then, the focus is directed in the area of maritime predictive maintenance which yields the identified gaps regarding maritime predictive maintenance.

2.3 Maintenance Concepts

Maintenance concepts represent the most basic approaches used in the maritime and other sectors and define how maintenance tasks are perceived and addressed. The discussed maintenance concepts were developed in chronological order and were driven by technological advancements and increased requirements for safety (Kobbacy 2008; Mohanty 2015). The basic maintenance concepts are shown in Figure 2.1, and they are Corrective Maintenance (CM), Preventive Maintenance (PM) and Predictive Maintenance (PdM).



Figure 2.1: Basic maintenance concepts

2.3.1 Corrective Maintenance

Corrective maintenance, also known as reactive or unplanned maintenance, is the oldest and simplest maintenance concept. The predominant CM approach is known as run-to-failure, were machinery operate until they fail, exploiting the entirety of their useful life (Shreve 2003). According to BS corrective maintenance can be defined as *"maintenance carried out after fault recognition and intended to put an item into a state in which it can perform a required function"* (BSI Standards Publication 2010). In the run-to-failure approach, any repairing or replacing actions take place by the available personnel following a machinery's failure (Wang 2002). Due to its simplicity, CM poses a logical starting point when dealing with maintenance in many industries (Arunraj and Maiti 2007).

The fact that some failures are unavoidable, due to the aforementioned inherent unreliability, makes CM a popular choice. Nonetheless, current proactive approaches, such as predictive maintenance, are usually more effective (Wang et al. 2014). In that sense, proactive approaches can avoid failures by maintaining systems based on their actual condition, prior to a failure. Modern maintenance approaches can significantly reduce the possibility of critical failures with disastrous effects (loss of life, pollution etc.), which is the biggest shortcoming of reactive maintenance (Prajapati, Bechtel, and Ganesan 2012). For that reason, CM can be used on non-critical auxiliary systems, where the risk of failure is low. Viable candidates for CM can include ship entertainment systems, lighting systems, secondary air conditioning systems, pipe filters, sealing gaskets and various other minor components (Mobley, Higgins, and Wikoff 2008).

One of the biggest challenges CM approaches face is the identification of the components to replace, given a system's failure (Fedele 2011). Complex systems have many different failure modes that can lead to a breakdown (Zhou, Xi, and Lee 2007). The challenge of corrective maintenance lies with its ability to identify the causes of each failure and consequently, relies heavily on acquired empirical knowledge. Moreover, CM relies heavily on the availability of spare parts, as any rectifying actions depend on the available spare parts inventory (Kobbacy 2008; Dikis 2017).

Another major disadvantage of CM, compared with other approaches is its associated cost. CM can result in increased operational expenses, primarily if it is used in all the systems of an asset (e.g. ship) (Fang and Zhaodong 2015). The increased cost can include downtime cost, cost of the repair and cost of cascaded damages. When using CM, there is an increased probability that a failure can cascade and cause damages to subsequent systems (Stenström et al. 2016). The probability of this can be reduced by identifying components that link different systems and failures can propagate through them. This process can be part of the critical equipment selection and under the scope of this work, it is considered as part of more advanced maintenance concepts(Polotski, Kenne, and Gharbi 2019).

Despite the above, CM is still applied in the maritime industry due to its simplicity and minimal maintenance planning cost. When CM is used, it is subjected to regulations and standards. Any use of CM must ensure that the shipping company complies with requirements set by the International Management Code for the Safe Operation of Ships and for Pollution Prevention (ISM Code) (IACS 2017). Lastly, the advantages and disadvantages of CM are summarised in Table 2.1.

Table 2.1: Corrective maintenance advantages and disadvantages

Corrective Maintenance		
Advantages	Disadvantages	
Simple for operators	Failures can cascade in other systems	
Simple maintenance plan	No information for fault-finding	
Ideal for non-critical items	More failures than proactive concepts	
Exploits the entire life of the equipment	Heavily relies on available spare parts	
Minimal maintenance planning cost	Can be very costly for the operator	

2.3.2 Preventive Maintenance

Preventive Maintenance (PM) represents a more advanced maintenance concept compared to CM. PM was developed shortly after WWII, at the beginning of the 1950s, and is a concept that aims in minimising failures, moving beyond the reactive character of CM (Garg and Deshmukh 2006). According to BS preventive maintenance can be defined as "maintenance carried out at predetermined intervals or according to prescribed criteria and intended to reduce the probability of failure or the degradation of the functioning of an item" (BSI Standards Publication 2010). In its core, PM is a task-oriented approach with tasks that are embedded in the everyday duties of the relevant personnel. PM is a structured maintenance concept and allows preemptive actions and planning of the maintenance. These tasks aim to prevent the equipment's degradation and can include visual inspections, testing, servicing and replacements (Smith and Mobley 2008).

Stepping away from CM can have a positive financial impact on companies and organisation. Notably, replacing CM with non-reactive approaches that schedule maintenance actions based on the actual needs of the asset (i.e. dynamic approaches) can have up to a 50% reduction in the overall maintenance cost (Stenström et al. 2016). This reduction arises from the increased reliability of the assets, which reduces the number of unexpected failures. This, in turn, reduces emergency expenses and allows companies to plan maintenance and take preemptive measures (Prajapati, Bechtel, and Ganesan 2012). The minimisation of unexpected failures also requires fewer spare parts to be stocked by companies. Having a reduced spare parts inventory allows for more efficient operations and reduces unnecessary parts procurement costs (Mobley, Higgins, and Wikoff 2008).

In PM, maintenance efforts occur when the system under examination is still operational. Several PM approaches are divided on usage-based and managementbased (Wang 2002). In general, the usage-based approaches specify criteria which determine when the repair or replacement of a system's component will take place. A popular usage-based approach is an age-dependent PM. In this approach, a system's component is replaced at a predefined age, T, assuming that a failure has not occurred (Kuboki and Takata 2019). Another popular usage-based approach is calendar-dependent PM. In the calendar-dependent PM, components are maintained at fixed time intervals (i.e. every Z hours of operation) or after a set usage (i.e. every X kWh) (Goossens and Basten 2015).

Despite the above, the successful implementation of PM can be heavily dependent on experience, practical and empirical knowledge. PM may be more dynamic than CM, as it is non-reactive and aims to schedule maintenance actions based on the actual needs of the asset. However, PM is not the most accurate approach for predicting the actual maintenance needs of the asset (Gandhare and Akarte 2012). In other words, the operational wear and various environmental factors are not taken into account. Therefore, systems can be unnecessarily maintained too often (over-maintained) or maintained too seldom (under-maintained); both of these situations can lead to unnecessary expenses or unexpected failures. Overmaintaining can increase operational expenses, as the full operational life of the

Preventive Maintenance		
Advantages	Disadvantages	
Structured and organised approach	Can over-maintain or under-maintain	
Aims at the minimisation of failures	Failures can still occur	
Allows for preemptive actions	Relies heavily on empirical knowledge	
More advanced than CM	Not as dynamic as possible	
More efficient use of spare parts	Can increase operational costs	

 Table 2.2: Preventive maintenance advantages and disadvantages

equipment is not exploited (Lin et al. 2019). Similarly, under-maintaining can result in failures and downtime. This can directly affect the profit-making capability of the asset in question, as the operational costs can increase (Yang et al. 2019). More dynamic maintenance concepts (based on the condition of the asset) can be used to maintain systems based on evidence of developing faults, prior to a failure taking place, as seen by the developments of this work. Lastly, the advantages and disadvantages of PM can be summarised in Table 2.2.

2.3.3 Predictive Maintenance

Predictive Maintenance (PdM) represents one of the most recent maintenance concepts. PdM was established around 1970 and represents a more dynamic and complex approach compared with CM and PM. PdM tries to minimise failures by scheduling the maintenance actions when the machinery and components are still operational (Sullivan et al. 2010a). As defined by the U.S Office of Energy Efficiency and Renewable Energy "Predictive maintenance attempts to detect the onset of a degradation mechanism with the goal of correcting that degradation prior to significant deterioration in the component or equipment" (Sullivan et al. 2010b).

PdM approaches aim at determining an equipment's state and condition based on data collection and non-destructive testing. The underlying philosophy of PdM is that equipment failures are the result of a gradual deterioration. Therefore, the use of appropriate, and real-time, condition-describing data could detect signs of degradation and identify developing faults before breakdowns occur (De Faria, Costa, and Olivas 2015). The successful implementation of PdM is based on the integration of condition-describing data with data-processing modules to achieve Fault Detection (FD) and possibly even proceed with diagnostic and prognostic tasks. It is becoming apparent that PdM can be used as a starting for additional innovation and further advancements (Niu 2016).

What sets PdM apart from PM, is the use of the equipment's actual condition for maintenance and inspection, instead of predefined and arbitrary agebased and use-based thresholds. Moreover, PdM can overcome the PM issues of over-maintaining and under-maintaining equipment while also moving beyond the reactive character of CM (Tian et al. 2011). In addition, PdM allows for better planning of maintenance, as the condition of the equipment can be assessed in real-time (Sharma, Yadava, and Deshmukh 2011). Regarding the maintenance cost, PdM can be up to 35% more cost-efficient than CM and up to 10% than PM. This is attributed to the improved efficiency of the spare parts use, viewed as part of the optimisation of the operational costs of the asset (Dikis 2017; Dikis and Lazakis 2019).

Despite the above merits, the implementation of PdM requires a shift in the culture of the company. Maintenance needs to be viewed as a hazard mitigation tool and not as a simple technical task. PdM requires an extra capital expense for its initiation, as it depends upon further training of personnel and the procurement of the appropriate equipment (sensors, communication relays, etc.). Moreover, PdM depends on additional expertise and know-how to accurately assess the various collected parameter (Lughofer and Sayed-Mouchaweh 2019). In summary, the advantages and disadvantages of PdM can be found in Table 2.3.

Predictive Maintenance		
Advantages	Disadvantages	
Real-time condition assessment	Requires cultural change	
Dynamic nature	Associated capital costs	
Reduced long-term maintenance cost	Additional expertise	
Optimisation of maintenance	Additional crew training	
Starting point for innovation	Added complexity	

Table 2.3: Predictive maintenance advantages and disadvantages

2.4 Maintenance Frameworks

The different maintenance concepts have given rise to several different maintenance frameworks. Maintenance frameworks address the concept of maintenance under a broader scope and often in conjunction with the management and profitability of a company. Maintenance frameworks do not only specify criteria for a component's maintenance but layout broader strategies that involve the entire organisation (Smith and Mobley 2008; Cheliotis and Lazakis 2015; Azadeh and Abdolhossein Zadeh 2016). Such frameworks usually involve more than one department of a company and can include both technical and managerial levels (Lazakis and Ölçer 2015). There are many different maintenance frameworks applicable to the maritime and other industries, for which there are a plethora of available literature (Eruguz, Tan, and Houtum 2017; Borjalilu and Ghambari 2018). However, this work focuses on Total Productive Maintenance (TPM), Business Centered Maintenance (BCM), Reliability Centered Maintenance (RCM) and Condition Based Maintenance (CBM) as these frameworks are found to be the most popular in terms of application.

2.4.1 Total Productive Maintenance

TPM is a maintenance framework that originated in Japan's automotive sector in the 1970s and has since then expanded in other sectors, including the manufacturing and nuclear. Nonetheless, TPM has very limited applicability in the maritime sector. TPM aims at maximising the equipment's effectiveness by maintaining components in their optimal state (Chan et al. 2005; Venkatesh 2003). This is achieved by minimising unnecessary breakdowns, downtime, defects and accidents and in general, by increasing the Overall Equipment Effectiveness (OEE) (Singh et al. 2013).

TPM is considered by many as a more advanced version of PM (Shafiee et al. 2019). In TPM, the maintenance actions are based on productivity-based criteria, as opposed to age-based and use-based thresholds (Deepak Prabhakar and Jagathy Raj 2014). TPM's advanced character stems from the belief that equipment degradation is accelerated when upkeeping efforts are not efficient (Sherwin 2000).

An essential characteristic of TPM is that it tries to involve all the departments of a company for its implementation. TPM is a highly structured framework and requires detailed preparation and planning for its fruition. It is essential for the successful use of TPM that an effective communication and collaboration system between different departments of a company is present (Ahuja and Khamba 2008). The requirements of TPM are summarised by the presence of five main pillars, as discussed in Braglia, Castellano, and Gallo (2019). The first pillar is the increase of the OEE. This is achieved by the minimisation of breakdowns, set-up and adjustment times, small stops, reduced speed, quality defects and start-up losses. The second pillar is the establishment of a comprehensive PM system, implemented for the entirety of the assets life cycle. The third pillar includes the involvement of different departments within a company for the implementation of TPM. The fourth pillar is centred around the participation of both white-collar and blue-collar employs. Finally, the fifth pillar is the promotion of PM through autonomous and self-motivated small groups.

Eruguz, Tan, and Houtum (2017) distilled from the above that TPM's pri-

mary advantage is the promotion of synergy within a company, which can have a positive effect on the quality of the maintenance. Along the same lines, Azadeh and Abdolhossein Zadeh (2016) discussed that TPM promotes the efficient use of the machinery, which can help to optimise the production process and improve profitability. Despite the above merits, TPM has several shortcomings. First of all, it is a resource-expensive framework that can be cumbersome to perform and implement, as discussed by Arca and Prado (2008). Moreover, any benefits from TPM can take a long time to manifest, creating an illusion of ineffectiveness. Lastly, according to Alsyouf (2009), TPM lacks on the technical aspect, as it does not specify detailed maintenance measures, and it does not set engineering-related goals.

2.4.2 Business Centered Maintenance

BCM is a framework that incorporates the optimal maintenance of assets in the overall strategy of the company, and hence it can be considered as a maintenance optimisation framework. BCM is considered to be an evolved form of TPM with an added emphasis on the company's productivity (Fedele 2011). Currently, BCM has applications in production-oriented organisation, including the manufacturing and automotive sectors, in various industrial processes and healthcare systems (Kelly 2006; Salim, Mazlan, and Salim 2019). Regarding the maritime industry, there are limited instances of BCM especially compared to the other discussed frameworks. Taher, Lazakis, and Turan (2014) established a BCM influenced framework for marine control systems based on probabilistic networks. Even thought this framework represents one of the few examples of maritime BCM, it does not address several issues. For instance, the effects of the data requirements of the maritime industry are not discussed and the topic of data preparation is also omitted. Similarly, the areas such as fault detection and diagnostics are not thoroughly elaborated.

The main aim of BCM is the continuous improvement of the maintenance quality and the various maintenance-related tasks (Kelly 2006). In BCM, the objectives of a company are considered as a function of the company's profitability. Moreover, BCM examines the specific inputs of the objectives of the business in its wider framework (Braglia, Castellano, and Gallo 2019).

Salim, Mazlan, and Salim (2019) stated that when BCM is applied correctly, it can have a positive influence on the profitability of the company. In the same publication, the authors stated that BCM is best suited for complex organisations with various complicated internal structures. As presented by Waeyenbergh and Pintelon (2002), a point of criticism for BCM is that it can quickly become very complicated and cumbersome, if it is not appropriately managed. Similarly, Gandhare and Akarte (2012) described that BCM is resource expensive both in terms of resources and human capital and that the positive results require a long time to manifest.

2.4.3 Reliability Centered Maintenance

RCM is a maintenance framework that originated from the aviation industry and more specifically, during the development of the Boeing 747 aircraft (Ten-Wolde and Ghobbar 2013). RCM has also spread in many other industries, including the offshore renewable and the maritime. More specifically, there are numerous RCM implementations for the maritime industry. For instance, Mokashi, Wang, and Vermar (2002) developed and RCM framework applied in the fuel purification system of a ship. However, the developed approach was based on qualitative methods and lacked quantitative aspects. Also, Vorkapic, Kralj, and Martinovic (2017) proposed an RCM framework for petroleum gas carriers, however, the work was qualitative and did not discuss technical aspects in-depth. RCM was conceptualised to improve the safety of systems by examining maintenance from a reliability improving viewpoint (Ten-Wolde and Ghobbar 2013). According to Moubray (1991) RCM can be defined as "A process used to determine what must be done to ensure that any physical asset continues to fulfil its intended functions in its present operating context". Unlike TPM and BCM, RCM is a more technical oriented framework. Instead of involving all the facets of a company, RCM is contained in the technical and maintenance departments. As a result, RCM requires more technical expertise and a deeper understanding of the functions of a system (Manzini et al. 2010).

A unique characteristic of RCM is its focus on the reliability of the components, as a function of their condition. Simply put, failures that do not reduce the reliability of a system are given a lower priority, under an RCM framework. For that reason, RCM can also be viewed as a maintenance prioritisation framework (Selvik and Aven 2011). As a result, the main aim of RCM is two-fold. This framework tries to increase safety and reliability while minimising maintenance costs (Kobbacy 2008).

For RCM to fulfil its aim, a sequence of processes must be followed. Initially, the examined system must be functionally decomposed and studied. Then, the mapping of the reliability of the components must take place, which allows for the selection of critical and non-critical components (Selvik and Aven 2011). The selection of critical components can take place either through the use of Failure Modes, Effects and Criticality Analysis (FMECA) or by using alternative datadriven methods (Takata et al. 2004). Finally, appropriate maintenance tasks are assigned to the different components, and a comprehensive maintenance plan is created (Cheliotis and Lazakis 2018).

As it can be inferred, RCM combines different maintenance concepts. Several non-critical components can be maintained under run-to-failure approaches, following the CM concept. Also, PM approaches can be used in components with low, to intermediate, effects on the system's reliability and safety. Lastly, critical components can be maintained using PdM concepts to optimise their reliability and minimise the probabilities of failures. However, this requires additional steps in terms of fault mapping and subsequent analysis. The use of multiple maintenance concepts in a single maintenance framework requires the combination of different areas of expertise. This can be a challenging task during the maintenance planning and requires a degree of coordination within the company (Fedele 2011; Ben-Daya et al. 2009).

An important advantage of RCM is its technical orientation. As discussed by Eruguz, Tan, and Houtum (2017) and Borjalilu and Ghambari (2018), the performed maintenance actions are a function of the technical requirements of the equipment. Moreover, RCM can be used to prioritise upkeeping tasks, which can help to streamline operations and minimise costs, as discussed by Smith and Mobley (2008). RCM is a framework that proactively addresses maintenance, and as such, it can be used as a stepping stone for more advanced approaches. Despite the above merits, RCM is a maintenance framework that is sometimes viewed with scepticism in the industry. Afzali, Keynia, and Rashidinejad (2019) explained that this is due to the resource-demanding nature of RCM, both in terms of its planning and implementation. In addition to that, RCM requires significant involvement from managerial levels, which can lead to complex systems (Lazakis 2011). Moreover, RCM requires the use of additional expertise and knowledge, which can also increase costs, as discussed by Mokashi, Wang, and Vermar (2002) and Cheliotis and Lazakis (2015).

2.4.4 Condition Based Maintenance

CBM is a maintenance framework that was developed at the same time as PdM and traces its origins in the railway sector. The concept behind CBM is the minimisation of failures through the assessment of the condition of the examined equipment (Prajapati, Bechtel, and Ganesan 2012). According to the BS, CBM can be defined as *"the maintenance policy carried out in response to a significant* deterioration in a machine as indicated by a change in a monitored parameter of the machine condition" (BSI Standards Publication 2010).

As it can be seen, CBM follows a proactive approach in maintenance, utilising relevant real-time condition-describing parameters (Prajapati, Bechtel, and Ganesan 2012). In CBM, maintenance actions are a function of the equipment's actual condition and are not based on arbitrary age and usage thresholds. A successful CBM framework should not only be able to identify the failure of equipment, but also to capture its degradation and even make predictions on its development and underlying cause (Bengtsson 2004). Therefore, CBM requires the integration of data collection techniques with fault detection, diagnostic and prognostic modules (Sullivan and Andridge 2015).

There are numerous publications discussing in detail all the different approaches of CBM. For instance, Jardine, Lin, and Banjevic (2006) performed a comprehensive review of the advancements in machinery CBM with emphasis on the diagnostic and prognostic tasks. Similarly, Liu et al. (2018) examined the use of AI for CBM applications in rotating machinery (e.g. pumps and turbines). Moreover, Hong et al. (2007) discussed and reviewed the developments in CBM, with an emphasis in fault detection, under the scope of offshore wind turbines. Lastly, Bernal, Spiryagin, and Cole (2019) elaborated on the advancements of real-time CBM in the freight railway sector. In this publication, the advancements in the enabling sensor technology and the developments in the available fault detection algorithms are discussed. It can be inferred from the above, that the area of CBM is under intensive development from a plethora of different sectors. CBM is taking advantage of modern technological advancements (e.g. ML, AI and advanced data analytics) with a high rate. It is also seen that CBM frameworks can easily integrate different models and algorithms for the completion of a single task. For example, different ML algorithms can be deployed in series in a single FD model, as seen in Liu et al. (2018).

Based on the selection of the monitored parameters, CBM, and be classified into Condition Monitoring (ConMon), Performance Monitoring (PeMon) and Process Monitoring (ProMon) (García Márquez et al. 2012). ConMon includes the use of condition-describing parameters. Such parameters can include vibration levels, Lubricating Oil (LO) characteristics, temperature distributions (thermography), acoustic emissions and electrical signature analysis (Wiggelinkhuizen et al. 2008; Mohanty 2015). For instance, ConMon can be used in various pumps, motors and engines (Liu and Zhang 2019). On the other hand, PeMon uses parameters that are directly related to the performance and output of the equipment. An example of such parameters can include the power output, torque output and rotational speed (Papagiannakis and Hountalas 2004; Rakopoulos et al. 2006). For example, PeMon can be used for the generating and Main Engine (ME) of a ship (Tsitsilonis and Theotokatos 2018; Baldi, Theotokatos, and Andersson 2015). Lastly, ProMon uses parameters that are related to the internal processes of the equipment. For example, the Exhaust Gas (EG) temperature, injection pressure, cooling temperature and oil viscosity can be used in ProMon (Zaher and McArthur 2007). ME, Turbochargers (TCs) and various other systems are typical candidates for ProMon (García Márquez et al. 2012). Apart from the different types of parameters used, ConMon, PeMon and ProMon have the same aim of describing, in real-time, the condition of the equipment (Kobbacy 2008).

An important decision in any CBM framework is the selection of the equipment that will be subjected to this approach. In addition to that, the examination of the failure modes of the equipment and a fault mapping must be conducted. Both of these tasks are based on domain knowledge and require additional expertise (Mobley, Higgins, and Wikoff 2008).

The successful implementation of CBM can have many benefits within a company. As discussed by Logan (2015), the proactive character of CBM can reduce maintenance costs for up to 30% and reduce unexpected failures by up to 70%. Also, CBM can allow for better planning of the maintenance and allocation of the available resources.

The advantages of CBM are numerous, as seen through the examination of the relevant literature. As deducted by examining Lazakis, Raptodimos, and Varelas (2018b), an advantage of CBM is that takes into account the real-time condition of the equipment in the maintenance decision making process. Consequently, operational efficiency and cost can be optimised dynamically, as confirmed by Dikis and Lazakis (2019). Even though CBM can offer significant benefits, there are still some limitations and issues that need to be addressed. Firstly, there is an associated capital cost regarding the procurement and installation of the data gathering equipment (Azadeh and Abdolhossein Zadeh 2016). Moreover, if the data gathering is continuous, there needs to be infrastructure for the storage of the data. In addition, the analysis of the data for the detection, diagnostic and prognostic tasks requires additional expertise, as presented by Bae et al. (2019). Lastly, the transition to CBM frameworks may require a cultural change within companies which can prove challenging (Jardine, Lin, and Banjevic 2006).

CBM has many applications in the maritime industry, and as seen through the evaluation of the literature, it is amongst the frameworks with the most maritime applications. For instance, Wang, Hussin, and Jefferis (2012) developed a CBM framework for marine main engines based on metal contamination analysis of the lubricating oil. A shortcoming of this framework is that it is based on periodic sampling and as such, it may not be the best option for continuous realtime monitoring. Also, Anantharaman (2013) applied a CBM framework for the propulsion plants of ships. This framework is focused on the use of a specific tool without emphasis on data preparation, fault detection or diagnostics. Similarly, Giorgio, Guida, and Pulcini (2015) deployed a CBM framework applicable to cylinder liners of marine main engines. However, the way in which this framework can be expanded to include other components or systems is not clear. Moreover, Lorencin et al. (2019) also used an ML based CBM framework with applications to marine propulsion plants. This developed framework displayed promising results, even thought data preparation was not addressed and the developed model was compared with the status quo.

2.4.5 Comparison of Frameworks

The previous sections presented the basic maintenance framework, detailing qualitatively their respective benefits and shortcomings. This section aims to perform a side-by-side qualitative comparison of the identified frameworks, as seen in Figure 2.2. Through the examination of the relevant literature, six criteria are distilled for the comparison of the different frameworks. These are simplicity, ease of application, fast results, required expertise, required training and sustainability. As seen, the criteria are selected based on their ability to assess the performance of the different frameworks broadly and inclusively, covering both technical and managerial criteria. Lastly, the selection of the discussed criteria is validated as in Deepak Prabhakar and Jagathy Raj (2014) and Lazakis and Ölçer (2015) report similar findings.

In detail, the simplicity criterion examines the overall complexity of the different frameworks. As seen in Figure 2.2, CBM outperforms the remaining frameworks. More specifically, RCM requires the additional mapping of the components' reliability, which increases its complexity, as also observed by Deepak Prabhakar and Jagathy Raj (2014). Similarly, both TPM and BCM include the incorporation of business-based and productivity-based considerations which results in more complex frameworks.

The ease of application criterion examines the required time to set-up and implement the various frameworks. Since CBM does not include additional reliability analysis, or business-oriented objectives it outperforms the other frameworks.



Figure 2.2: Maintenance frameworks qualitative comparison

Similarly, RCM is found to be easier to apply compared to TPM and BCM due to their additional business-oriented objectives.

The fast results criterion considers the required time to improve the effectiveness of maintenance by improving safety. As seen in Figure 2.2, CBM has the highest score since it can assess in real-time the condition of assets, mitigating the risks of failures.

The required expertise criterion gauges the requirements for additional knowhow, or expert knowledge, necessary to implement the frameworks. As demonstrated in Figure 2.2, RCM and CBM are the best performing frameworks, since these frameworks are isolated to the technical and maintenance departments. As a result, there is no need for managerial expertise that is required to coordinate and manage the complex functions of TPM and BCM frameworks, as also discussed by Shafiee et al. (2019) and Borjalilu and Ghambari (2018).

The required training criterion considers the need for supplementary training of the maintenance staff. As presented in in Figure 2.2, CBM and RCM are the lowest performing frameworks. This is caused by the additional skills that are required for the reliability analysis of RCM and for the handling and interpretation of the condition describing parameters of CBM. On the other hand, TPM and BCM are more focused on the communication and synergy of the different departments, rather than the development of new technical skills.

Finally, the sustainability criterion examines the potential for effective longterm application of the different frameworks. As discussed by Shafiee et al. (2019) and Azadeh and Abdolhossein Zadeh (2016), the sustainability of CBM is preferable, followed by RCM, TPM and BCM.

2.5 Predictive Maintenance Tools

When adopting a PdM concept, implemented through any PdM-based framework, numerous tools can be used. These tools are broadly divided into Reliability Assessment (RA) tools and Data Science (DS) tools. Each of these categories can be used for different tasks, and their selection depends on the specific application at hand. Moreover, the RA and DS tools represent the building blocks based on which different tasks and processes performed.

2.5.1 Reliability Assessment (RA) Tools

RA tools are extensively used in PdM concepts and especially in RCM. These tools are used to analyse the risk, safety, reliability and criticality of systems. Commonly, RA tools can be classified as qualitative or quantitative and as topdown or bottom-up approaches. Qualitative RA tools address the issues of risk, safety, reliability and criticality descriptively, whereas quantitative RA tools try to quantify these issues numerically. Similarly, top-down tools, focus on the broader context of risk, safety, reliability and criticality by analysing the causes of specific events. On the other hand, bottom-up approaches examine the behaviour of a system subjected to disturbances (Raptodimos 2018; Lazakis 2011).

2.5.1.1 Fault Tree Analysis (FTA)

Fault Tree Analysis (FTA) is one of the most common and widely recognised RA tools. FTA was conceptualised in 1962 by the US Air Force and was quickly adopted by the wider aviation industry (Haasl et al. 1981). Nowadays, FTA is applied in many industries, including the manufacturing, nuclear, automotive, maritime and offshore sectors (Kabir 2017).

FTA utilises logic-gates and events to represent an engineering system (e.g. fuel supply system) and to create a visual model with interconnected pathways that can lead to an undesirable failure within the system (Kang, Sun, and Guedes Soares 2019). The logic-gates simulate the functional dependencies within the examined system, and they are usually employed to represents sub-systems and sub-assemblies. On the other hand, events are used to model components, and they are located at the lower level of the system's model architecture. The events are also used to quantify the Fault Tree, as they require the input of failure statistics (e.g. failure rate, probability of failure) for each component (e.g. fuel injector) (Henriques de Gusmão et al. 2018a). Once a Fault Tree is quantified, the failure statistics of each event are used for reliability calculations in the modelled gates. As a result, the failure statistics propagate through the Fault Tree's structure, leading to the investigation of the examined top-event. FTA is a topdown approach and initiates by stating an undesirable event, which is referred to as a top-event. Common undesirable top-events can include software failure, machinery failures and structural failures. The examination of the top event is a function of the fault tree's structure and a set of logical rules represented through Boolean Logic and more recently, Fuzzy Logic, as seen in Yazdi, Nikfar, and Nasrabadi (2017) and Kabir et al. (2016). From an analytical perspective, FTA can be considered both as a qualitative and a quantitative method (Ruijters and Stoelinga 2015). Qualitatively, FTA can be used to represent functional dependencies within a system. For quantitative tasks, FTA can be used to compute the reliability of systems and other metrics, including failure rates and reliability Importance Measures (IM).

As previously mentioned, FTA can be used to identify paths to failures. To achieve this goal, the scope of the analysis must be set, through the system definition step. Following this, the top-level faults are selected to define the failure of interest, leading to the investigation of their possible causes. Then, the next level of events is investigated, as the causes of the top-events also have precipitating events. Following this top-down approach, the root-causes are identified, and they represent the lowest level of the Fault Tree. The root-causes are the initiating points of the different sequences that lead to the different failures (Haasl et al. 1981; Vesely 2002). Figure 2.3 demonstrates a rudimentary Fault Tree structure that is derived using the above approach, comprising of two root-causes and a single top-level fault. In this case, the presented Fault Tree examines how two basic events (Input A, Input B) influence the occurrence of a top-event (Output A), through a Boolean AND-gate. With the AND-gate, the Output A occurs when Input A and Input B occur at the same time (Relex Software Corporation 2003). For example, the presented figure could be used to examine the loss of function in a redundant power generation systems with two generating engines operating in parallel (Kabir 2017). Alternatively, OR-gates can be used, where the examined output occurs if any of the inputs occur. For instance, an OR-gate can be used to model the loss of function of a propulsion system with only one ME (Vesely 2002). There are more gates available, apart from OR-gates and

AND-gates, which are used for different applications. For a detailed accounting of the available gates, the reader is referred to Relex Software Corporation (2003) and Rausan and Hoyland (2004).



Figure 2.3: Representation of a basic fault tree structure

As previously mentioned, FTA has widespread applicability. Whiteley, Dunnett, and Jackson (2016) employed FTA for the qualitative and quantitative reliability assessment of a Polymer Electrolyte Membrane Fuel Cell. Yazdi, Nikfar, and Nasrabadi (2017) developed a methodology for the assessment of failure probabilities in components of chemical plants, aiming at reducing the number of accidents. Also, Henriques de Gusmão et al. (2018b) used FTA in conjunction with fuzzy theory under the context of cybersecurity. More specifically, the developed methodology was used to manage cybersecurity risks and investigate the vulnerability of systems. Lastly, Khare, Nema, and Baredar (2019) used FTA for the reliability assessment of renewable energy systems. The developed methodology used weather data and power-consumption information to control the occurrence of critical faults.

FTA is a well-established and widely recognised RA tool. Also, as concluded from this research, FTA is easy to learn and offers quick and interoperable results. Moreover, FTA can be used both as a qualitative and a quantitative tool, significantly improving its flexibility. Despite the above merits, FTA cannot model effectively multiple functional dependencies and can become cumbersome when modelling large systems (Fussell 1975; Vesely 2002). Moreover, obtaining the required failure statistics for quantitative analysis can sometimes be difficult (Jiang et al. 2018).

2.5.1.2 Bayesian Networks (BN)

Bayesian Networks (BN) is a popular RA tool that traces their origin in computer science, where they were developed in 1985 by Judea Pearl (Pearl 1985; Pearl 1988). Like FTA, BNs have a widespread recognition with applications in many industries including the nuclear, manufacturing and recently maritime sector (Cai et al. 2019).

BNs are probabilistic Directed Acyclic Graphical (DAG) models that depict functional and causal dependencies between random variables. In other words, BN represent a joint probability distribution of a set of random variables. Like FTA, BN consist of a qualitative part and a quantitative part. The qualitative part is defined by a DAG model where each variable is depicted as a node. Depending on how the nodes are connected, they can be subdivided into parent, child, leaf and root nodes (Ruggeri, Faltin, and Kenett 2007). The qualitative part also includes directed links between the nodes to define causal relationships and functional dependencies. Similarly, the quantitative part is defined by the conditional probability distribution in the Conditional Probability Table (CPT) of each node (variable).

Figure 2.4 represents an indicative BN. In the presented network, the four random variables X1, X2, X3 and X4 are represented as nodes and complete the qualitative part of the network. The directed links demonstrate the causal relationships between the nodes. In more detail, the X1 node is the parent node for nodes X2 and X3. Also, the X1 node is a root node, as it only has child nodes. Similarly, the X4 node is the child node of nodes X1 and X2, and it is also a leaf node as it only has parent nodes. BNs are based on Baye's theorem, with the goal of calculating the posterior conditional probability distribution of a fault, state, or condition given some observable evidence.



Figure 2.4: Representation of a basic Bayesian Network

Due to the causal dependencies, BNs are widely used for a variety of diagnostic tasks ranging from medical to engine diagnostics (Langseth and Portinale 2007; Borunda et al. 2016). Yuan et al. (2015) developed a methodology for the risk analysis of dust explosions based on BNs. The methodology also examined the relationship between causing factors and consequences. Shin et al. (2015) employed BNs to evaluate cybersecurity risks in the nuclear industry by incorporating procedural and technical aspects. Also, Hosseini and Barker (2016) used BNs to examine resilience-building strategies in infrastructure systems. Moreover, Cai, Liu, and Xie (2016a) developed a methodology for probabilistic reasoning and real-time root-cause analysis in industrial processes based on BNs. Lastly, Cai, Huang, and Xie (2017a) provided an extensive literature review of the application of BNs for FD and diagnostics in engineering systems.

In general, BN can accurately model very complex systems, which makes them ideal candidates for diagnostic tasks. However, as noted from this research, diagnostic BN require a plethora of failure statistics which are not always readily available. Similarly, with FTA, BNs can be used both as qualitative and quantitative tools. The structure of the network can give a qualitative description of a system and the functional dependencies within. Also, BNs can perform powerful reliability calculations by integration information from different sources (i.e. sensor fusion) (Zhang and Thai 2016; Mittal and Kassim 2007). Despite the above merits, BNs can become too complicated. As also noted from this research, when modelling complex networks, attention needs to be paid to avoid unnecessary connections between nodes. Since the number of connections between each node increases to complexity of the CPTs, the network must not have unneeded connections, so that the CPTs are manageable Similar results regarding the complexity of the BNs as a function of their size are noted by Horný (2014) and Cai, Huang, and Xie (2017b).

2.5.1.3 Failure Modes Event Analysis (FMEA) and Failure Modes Event and Criticality Analysis (FMECA)

FMEA and FMECA are two very similar RA tools that are widely used in many different sectors (Ben-Daya et al. 2009). FMEA and FMECA can be used to foresee possible failures during the design of a system, by identifying all of the potential failure modes, through the examination of engineering hazards (Mobley, Higgins, and Wikoff 2008). The main philosophy of these two RA tools is the anticipation and prevention of failures in a system by examining the different ways a system can fail (CDNSWC 2010).

FMEA and FMECA were developed around 1950 and represent some of the earliest structured reliability improvement methods (Mohanty 2015). Both tools require extensive engineering knowledge of the studied system, and their results are presented in a tabulated format (Lughofer and Sayed-Mouchaweh 2019). FMEA can be considered as a qualitative tool, which is developed by using chained "what-if?" questions (Kobbacy 2008; Isermann 2006). On the other hand, FMECA is quantitative as it tries to quantify the criticality of each failure, caused by the different hazards. In other words, FMEA can be performed first, and after, a criticality analysis through FMECA can follow (Bertsche 2007).

As previously mentioned, FMEA and FMECA have applications in many industries. Lazakis, Turan, and Aksu (2010a) used FMECA in a ship reliability improvement methodology, aimed at minimising downtime and improving operability. Dinmohammadi and Shafiee (2013) developed an integrated methodology combining FMEA with fuzzy logic for the reliability assessment of wind turbines. More recently, Peeters, Basten, and Tinga (2018) combined FTA with FMEA for failures analysis for the manufacturing sector. Lastly, Balaraju, Govinda Raj, and Murthy (2019) used FMECA for reliability modelling in the mining sector to improve productivity by examining failure behaviours.

Both FMEA and FMECA are widely applicable tools, as they provide a structured approach to reliability improvement (Mobley, Higgins, and Wikoff 2008). These tools address in detail the technical issues of the examined system, provide a starting point for mitigating risk (Spreafico, Russo, and Rizzi 2017). FMEA and FMECA can also be used for the prioritisation of maintenance, which enhances their functionality (Peeters, Basten, and Tinga 2018). Despite the above merits, these approaches face several shortcomings. Initially, they require a substantial amount of resource for their completion (Alsyouf 2009). Moreover, they can be very complex and cumbersome, primarily when used in extended systems (Tixier et al. 2002; Verma, Ajit, and Karanki 2010). Lastly, FMEA and FMECA may need to be re-initiated after extensive repairs or alterations of the specific system (Joshi and Joshi 2014). This may be required, since significant alterations to a system can either introduce new hazards (i.e. through the addition of new components) or alter the significance of existing ones (i.e. through the change of the system's architecture). In either case, the criticality and importance of components and hazards need to be reconsidered, and FMEA and FMECA re-initiated.
2.5.1.4 Event Tree Analysis (ETA)

Event Tree Analysis (ETA) is a popular RA tool that examines the possible outcomes in a system resulting from an initiating event, usually a failure (Ben-Daya et al. 2009). ETA is often used to identify the potential chains of events and resulting outcomes by examining the response of a system to a disturbance (Tixier et al. 2002). Moreover, it can be used to identify procedural and technical weaknesses and to manage risks by calculating the probabilities of different scenarios, (Dikis, Lazakis, and Theotokatos 2015). ETA was conceptualised around 1970 in the nuclear sector and currently is used in many industries and for different applications (Jankovsky, Denman, and Aldemir 2018; Mobley, Higgins, and Wikoff 2008).

ETA is a quantitative approach and requires the input of failure statistics, to assess, through binary logic, the probabilities of different possible outcomes (Mohanty 2015). In more detail, each of the different events are represented in individual branches that also include the probability of the different events. ETA is performed sequentially and initiates by stating an examined event (e.g. failure). The consequences of the initiating event are examined through a series of different outcomes, with each different outcome represented in its branch (Smith and Mobley 2008; Rausan and Hoyland 2004). As the ETA progresses, the scenarios represented by the structure of the Event Tree are considered and the different chains of events are investigated. Completing an ETA results in quantified chains of events with computed probabilities for the different branches.

ETA is used in many different application from different sectors. Ramzali, Lavasani, and Ghodousi (2015) assessed the effectiveness of safety barriers in oil and gas drilling systems by using ETA, to minimise the impact and occurrence of accidents. Fu and Zhang (2016), combine ETA with fuzzy logic for the investigation of accident scenarios and related consequences of ships stuck in ice in Arctic waters. Also, Raiyan, Das, and Islam (2017) examined the root-cause of maritime accidents by using ETA. Aziz et al. (2019) developed a methodology for the quantification of risks related to ship systems, under different failure scenarios. Lastly, Mares, Nagy, and Radu (2020) focused on the investigation of work accidents in the construction sector by applying ETA to identify procedural and technical weaknesses.

ETA is a widely recognised RA tool that is accepted in many industries and for a variety of purposes. It is instrumental in modelling successive events and analysing the propagation of hazards. Moreover, ETA can identify both procedural and technical problems (Ben-Daya et al. 2009; Mokashi, Wang, and Vermar 2002). However, ETA is limited in the analysis of only one initiating event at a time. As a result, it can be pervasive when trying to study multiple initiating events (Mobley, Higgins, and Wikoff 2008). Lastly, ETA operates under the premise that the modelled events are independent, which can lead to inaccurate results when this assumption is violated (Fu and Zhang 2016; Raiyan, Das, and Islam 2017).

2.5.1.5 Comparison of RA tools

The previous sections presented some of the necessary RA tools, describing their characteristics and discussing their respective shortcoming and benefits qualitatively. The main goal of this section is to complete a qualitative assessment of the discussed RA tools.

The RA tools are assessed against five criteria, which were identified in the previous sections. These criteria summarise the performance of the RA tools transparently and range from user-based to more technical ones. More specifically, the used criteria include how well established and flexible each tool is, the different applications of each tool and the ability to model functional dependencies and sophisticated systems. In detail, the flexibility criterion examines the ability of each tool to be used in both qualitative and quantitative manners. Also, the applications criterion assess the variety of applications each tool can be used. The functional dependencies criterion explores the ability of each tool to model systems with intricately interconnected components. Lastly, the complex systems criterion considers the ability of each tool to model complex systems by also incorporating information from different sources (i.e. sensor fusion).

Figure 2.5 shows the performance of the examined RA tools against the distilled criteria from the previous sections. FTA is a well-established tool with high



Figure 2.5: RA tools qualitative comparison

flexibility, as it can be used for both qualitative and quantitative tasks. It can be used in several sectors; however, it is usually restricted with reliability-related tasks (e.g. reliability analysis, criticality analysis). FTA is also limited in its

Ship Maintenance

ability to model functional dependencies in examined systems. Lastly, FTA can model complex systems and could use information from different sensors. Similarly, BNs are well established; however, they are not suitable for quantitative analysis. BNs can be used in many applications, ranging from diagnostics and FD to decision support, and they can also model functional dependencies effectively. Lastly, BNs are suitable for modelling complex systems and are very good at integrating information from different sources. FMEA and FMECA are also well-established tools which can be used both for qualitative (FMEA) and quantitative (FMECA) purposes. These tools are limited to reliability analysis and cannot model functional dependencies. Lastly, FMEA and FMECA can model complex systems; however, the process of doing that can be time consuming, due to its exhaustive tabulated format. ETA is also a very well established RA tool. Its flexibility is limited to only quantitative analysis; however, the structure of the Event Tree may give a rudimentary understanding of the chain of causality. ETA is limited in its applications, with most examples from the areas of accidents investigations and reliability analysis. Lastly, ETA cannot model complex systems and situations, as the resulting Event Tree can become too complex and cumbersome.

It is seen that BNs demonstrate well-rounded behaviour. They are useful in modelling functional dependencies and have a vast range of applications, with many examples from the area of FD and diagnostics. As a result, they are good in modelling complex systems, provided that the required data are available. Similarly, FTA is very flexible and can be used both for qualitative and quantitative tasks, which makes it an ideal tool for the initial steps of an analysis. FTA is a good candidate for reliability analysis and can provide more precise results than FMEA or FMECA.

2.5.2 Data Science (DS) Tools

DS tools represent another set of tools and methods that are frequently used in PdM-based schemes. As described by Han, Kamber, and Pei (2012), the aim DS tools are to provide the means with which to analyse data and extract useful information about a studied system. DS tools are used when the PdM-based framework is heavily based on the use of data. That could include cases where the reliability of equipment is calculated based on collected data (e.g. RCM), or when the condition of the equipment is inferred by process, performance or condition describing parameters (e.g. CBM).

DS tools represent a different approach to tackling problems in PdM schemes. Apart from that, there are several practical differences between DS and RA tools. Firstly, DS tools have a larger degree of utilisation of data than RA tools. In more detail, DS tools may require a substantial amount of data to train different algorithms, as further discussed in the following section (Bishop 2006). Also, RA tools are usually employed for reliability and risk assessment tasks. In contrast, DS tools tend to have wider applicability, with applications ranging from condition assessment to fault detection and diagnostics. In other words, DS tools offer a higher degree of modularity between the different approaches. Lastly, DS tools do not have qualitative aspects, as they are purely quantitative, which places a greater emphasis on the requirements for data.

PdM-based frameworks can be developed by using traditional analytic tools and methodologies from DS. For example, statistical models have been employed in the nuclear sector as seen in Hoseyni, Di Maio, and Zio (2019), in the offshore sector as seen in Taylor and Jeon (2018), in the coastal engineering sector as seen in Chen and Mehrabani (2019) and in the maritime sector as seen inDikis and Lazakis (2019). Additional maintenance frameworks based on other DS tools, and more specifically on control charts, have also been proposed. For instance, Salmasnia, Kaveie, and Namdar (2018) examined the use of the Hotelling control chart in an application for production plants. Also, Boullosa-Falces et al. (2019) used the Cumulative Sum (CUMSUM) control chart for the optimal maintenance of heat exchangers. Moreover, Holmes and Mergen (2000) examined the application of the Exponentially Weighted Moving Average (EWMA) control chart for fault detection, while Badodkar and Dwarakanath (2017) focused its use for detecting developing cracks in mechanical gears.

There are also numerous PdM-based frameworks that are developed using ML. ML is a sub-field of DS with increasing applicability not only in PdM frameworks but in most of the aspects of modern society. The primary purpose of ML is to extract knowledge from data in the same way as humans do. ML is unique in its ability to learn from data and even improve certain processes, without being explicitly programmed to do so (Kirk 2017). In the core of ML are the various algorithms which can be broadly divided into three categories: a) supervised learning and b) unsupervised learning Müller and Guido (2015).

2.5.2.1 Supervised Learning

Supervised learning is one of the most common categories of ML algorithms. The main objective of supervised learning is to fit a function to the available data, to automate a specific process by generalising from provided samples (Kirk 2017). In supervised learning, the algorithms are given sets of inputs and known outputs, (Bishop 2006). Based on that information, the algorithm tries to uncover patterns and relationships between the input/output pairs (mapping) so that it can later produce outputs based on previously unseen inputs. This process is called training, as the algorithm is trained (i.e. learns) based on the given data. Nonetheless, the training of an algorithm can be a laborious and challenging task. As observed from this work, training algorithms has many intricacies that can lead to inaccurate models if overlooked. Similarly, assessing the accuracy of the training is a crucial task that can be prone to misinterpretation with negative

consequences. Similar findings are also reported by Müller and Guido (2015).

Supervised learning algorithms can be further subdivided in regression and classification algorithms, both of which aim at predicting the value of a variable. Regression algorithms aim at predicting the value of a continuous variable, whereas classification algorithms are concerned with the value of discrete variables (Ayodele 2010).

There are a plethora of supervised learning algorithms both for classification and regression tasks. Notably, the most common classification algorithms include Support Vector Machines (SVM) and Logistic Regression (Mohammed and Wagner 2014). Similarly, the most widely recognised regression algorithms include Ordinary Least Squares (OLS) Regression, Ridge Regression and Lasso Regression. Apart from these, Artificial Neural Networks (ANN), Decision Trees and k-Nearest Neighbours (kNN) can be adapted and used for both regressive and classifying purposes (Bishop 2006).

There are vast applications of supervised learning algorithms ranging from facial recognition and credit card fraud detection to fault detection and diagnostics (Bishop 2006). King, Feng, and Sutherland (1995) provided one of the earliest benchmark comparisons between the performance of different classification algorithms. Caruana and Niculescu-Mizil (2006) performed a large scale comparison of various supervised learning algorithms, including SVMs and kNN. The comparison was facilitated by analysing the classification performance of the different algorithms in a normalised dataset obtained from the University of California Irvine Machine Learning Repository (Irvine 2020). Similarly, Ahmed et al. (2010) performed a large scale comparison of regression models under the context of time series forecasting. In the study, the most prominent ML algorithms, including ANNs and kNN, were compared on a dataset obtain from the International Institute of Forecasters (International Institute of Forecasters 2020). Also, Raczko and Zagajewski (2017) compared classification algorithms, including SVMs and ANNs, with emphasis on time-series forecasting and image recognition. Lastly, Louzada, Ara, and Fernandes (2016) presented a systematic review and evaluation of the most prominent classification algorithms under the scope of the credit score sector.

2.5.2.2 Unsupervised learning

Unsupervised learning is another broad category of popular ML algorithms. The main objective of unsupervised algorithms is to examine and scrutinise the data without any feedback. Unlike supervised algorithms, unsupervised algorithms do not have a training phase. In other words, there is no mapping between inputs and know outputs (Bishop 2006). Instead, the various algorithms attempt to make inference of the data without being given explicit examples. Despite the absence of a training phase, unsupervised algorithms are useful in many applications (e.g. market segmentation), albeit they are sometimes harder to interpret (Müller and Guido 2015; Kirk 2017).

The primary purpose of unsupervised learning is to understand, group, and quantify how similar, or dissimilar, the given data are. The most common unsupervised learning methods is known as clustering, or segmentation analysis (Han, Kamber, and Pei 2012; Kirk 2017). Clustering methods aim at creating groups (i.e. clusters) of data that are used to find hidden patterns within the dataset or to categorise the data. Data in the same cluster are considered similar, and dissimilar with those in another cluster. In addition, distance metrics can be used to quantify the degree of similarity (Mohammed and Wagner 2014).

There are many ways in which clustering algorithms can be categorised. Selim and Ismail (1984) suggested that clustering algorithms can be divided between soft and hard clusters, depending on if single data points can have partial membership between two clusters. Alternatively, as suggested by Han, Kamber, and Pei (2012) clustering algorithms can be divided between hierarchical and partitional, depending on the underlying method for the specification of the clusters. The most common and widely applied clustering algorithms are the k-means, cmeans, Density-Based Spatial Clustering of Applications with Noise (DBSCAN), and agglomerative hierarchical clustering (Schubert et al. 2017; Müller and Guido 2015; Rahmah and Sitanggang 2016).

The applicability of clustering ranges from market segmentation to anomaly detection (Bishop 2006). Saxena et al. (2017) and Renjith, Sreekumar, and Jathavedan (2020) presented an overall review of the different clustering algorithms in terms of their characteristics, performance and applicability. Also, Hegde and Rokseth (2020) provided an extensive review of ML tools, including unsupervised models, applied for the quantification and control of risk in engineering systems. Similarly, Fan et al. (2018) and Miller, Nagy, and Schlueter (2018) reviewed the applications of unsupervised models for the enhancement of the energy efficiency of large buildings. Lastly, Mukherjee et al. (2020) presented a clustering approach streamlined for the large volume information requirements of IoT applications.

2.6 Predictive Maintenance Processes

Based on the PdM tools discussed in Section 3.5, the main processes and applications of PdM-based frameworks are identified. These processes include the identification of critical equipment within the examined system, the preparation of the required data, the creation of FD and diagnostic modules.

2.6.1 Critical Equipment

The manner in which a failed component affects its system depends on the component's criticality. The criticality of a component denotes its overall importance to the mission of the system it belongs. It can be a quantitative risk index that examines different failure modes of components and how these affect the different systems (ABS 2016). Therefore, identifying the critical components of the system under consideration becomes an essential task. It is a crucial first step towards more effective management of the systems and assets under examination and also a prevalent starting point for RCM and CBM frameworks.

In general, the critical components identification process is divided into four broad categories, depending on the underlying identification method. These categories, as reviewed by Erozan (2019) and Sarih et al. (2018), include methods based on Network Analysis, Expert Judgement, DA and Reliability Related Techniques.

Network Analysis is a very popular critical components identification method that is widely used in the field of electrical power distribution systems (Chen et al. 2018). Network Analysis is often used to identify critical branches, buses and generators of large power grids. This method is effective when examining large interconnected, and grid-like systems. As seen in Chen et al. (2018), these methods can be used for examining the flow of power transmission in complex network models. Moreover, as examined by Xu et al. (2012) and expanded by Pordanjani, Wang, and Xu (2013), electric circuit analysis based on Thevenin circuits can be incorporated with channel components transform to detect critical components on power grids.

Apart from Network Analysis, methods based on Expert Judgement are often used for identifying low-reliability and high-risk items. These methods are developed as they are easier to impart domain and first-principle knowledge. In contrast with Network Analysis, Expert Judgement methods have a wider applicability with examples in industrial machinery and offshore wind turbines. Dehghanian et al. (2012) developed a method for the identification of critical components for power grids by combining Fuzzy Logic with Analytical Hierarchy Process (AHP). Similarly, Musman and Ahmad (2018) also created an AHP methodology combined with Fuzzy Logic with an emphasis on autonomous maintenance actions. This methodology was showcased in an application for lathe Computer Numerical Control (CNC) machines. Also, Gupta and Mishra (2018) developed a methodology based on Analytical Network Process (ANP) for the identification of critical parts in CNC lathes. Lastly, Özcan, Ünlüsoy, and Eren (2017) developed an integrated framework combining the Technique for Order of Preference by Similarity to Ideal Solution (TOPSIS) and the AHP for the identification of critical components in large-scale hydroelectric power plants.

As mentioned earlier, critical component identification can also be based on DA. This group of methods has received the least attention and is currently under development. For instance, Sarih et al. (2018) created a methodology for critical components for the media and broadcasts sector. The developed methodology was based on recorded data from various components and used logistic regression and Pareto analysis. Also, Erozan (2019) developed a critical component identification methodology for the manufacturing sector. The proposed method was based on the exponential distribution and in the safety shock and redundancy effects.

Another popular group of methods for the identification and selection of critical components are based on reliability related techniques. This group of methods employ various RA tools and Reliability Indices (RI) with applications in many industries, including the offshore and maritime sector. For instance, Hilber et al. (2007) created a methodology for critical components in power grids combining the component reliability importance index calculated through Monte Carlo simulations. In the same sector, Dehghanian and Fotuhi-Firuzabad (2012) developed a methodology based on the common load RI, which was used as part of an RCM framework. Also, Afzali, Keynia, and Rashidinejad (2019) developed a new weighted importance RI for the identification of critical components, used for the maintenance prioritisation of mechanical components in distribution centres. Two of the most common tools for critical component identification are FMEA and FMECA (Erozan 2019). In that respect, Narayanagounder and Gurusami (2009) developed an improved FMEA framework, incorporating ANOVA analysis for applications in industrial motors. Also, Choudhary and Sidharthan (2016) used FMECA for critical components identification of electronic components. Similarly, Cevasco, Collu, and Lin (2018) developed an FMECA framework for offshore Wind Turbines (WT). The framework was developed as part of an RCM framework aiming at minimising maintenance costs.

2.6.2 Data Preparation

Apart from the identification of critical components, another vital process of PdM frameworks includes the preparation of data, also known as pre-processing. Data pre-processing improves the effectiveness of PdM frameworks by improving the quality of the used data, as extensively reviewed by Han, Kamber, and Pei (2012), Tan, Steinbach, and Vipin Kumar (2006), Mohammed and Wagner (2014), and Kotsiantis, Kanellopoulos, and Pintelas (2006). Under the scope of this work, the data preparation efforts are divided into two categories, including Outlier Detection and Imputation, presented next.

2.6.2.1 Outlier Detection

Outliers are considered as sparse data points with significantly different values from the rest of the instances of the same variables. Outliers are usually caused by sensor errors and other instrumental faults and are not part of a fault indicative pattern. For example, negative EG temperatures and power outputs above an engine's rated power are considered as outliers. Consequently, outliers can be perceived as data "anomalies" and unreliable readings, and if they are not removed, they can skew the results and decrease the accuracy of the developed models (Javed and Wolf 2012). The methods based on which outliers are detected can by divided into algorithmic and domain knowledge processes (Karkouch et al. 2016; Hodge and Jim 2004).

Domain knowledge processes are simplistic and involve the manual check of the data to identify points that do not conform to an expected behaviour. These processes can rely on information from the equipment's manufacturer, by filtering the data against alarm levels, operational limits and benchmark tests. Moreover, visualisation tools, including scatter plots, can be used to visually check the data. For example, Cheliotis et al. (2019) detected outliers by comparing recordings with limits obtain from engineering principles. Similarly, Dikis and Lazakis (2019) detected outliers from ship-system signals by comparing the available data with the alarm limits set by the manufacturers.

Algorithmic outlier detection is suitable for faster and automated processes and does not require domain knowledge. A common detection method is based on the assumption that the data follow a specific distribution (e.g. Gaussian distribution) (Maimon and Rokach n.d.). For example, Lazakis, Gkerekos, and Theotokatos (2018) detected outliers based on the distance of each data point from the data mean, under the normality assumption. Even though the normality assumption is a popular choice, it is frequently unrealistic for given datasets. To counter this shortcoming, clustering algorithms can be used to detect outliers in an unsupervised manner. Celik, Dadaser-Celik, and Dokuz (2011) used the Density-Based Spatial Clustering of Applications with Noise (DBSCAN) algorithm to detect outliers in a dataset containing daily average ambient temperatures. Similarly, Chen and Li (2011) used the DBSCAN algorithm to detect outliers under the scope of cyber-security and more specifically in detecting unauthorised accesses in private systems. Thang and Kim (2011) also examined the use of DBASCAN for the detection of outliers related with cyber-security issues. Lastly, Ijaz et al. (2018) used DBSCAN for the identification of outliers for medical applications, including data for hypertension and diabetes.

2.6.2.2 Imputation

Missing values befall in most data-driven research efforts and applications and involve the loss of relevant information (e.g. outliers). If they are not dealt with in a case-appropriate manner, they can reduce the power of models and skew results. As discussed by Domingos (2012) and Cheliotis et al. (2019), reducing the number of missing values has a positive effect on the accuracy of any subsequent models, as it preserves the predictive power of the used datasets. Imputation is a very popular approach in managing missing values and includes techniques for replacing missing values with substitutes.

Understanding the underlying causes of missing data is an important step in data imputation, as it dictates the way the missing data can be handled. In general, three mechanisms affect how data are missing as established by Rubin (1976), Taylor and Rubin (1996), and Little and Rubin (2002). The Missing Completely at Random (MCAR) mechanism refers to cases where the missingness is independent of the data. In that case, there is no correlation between the missing data and the variables in the dataset, as the missingness is entirely unsystematic. For example, in the maritime domain, a random failure of the fuel flow meter will lead to data that are MCAR. Missing at Random (MAR) is when missing data are related to other observations. In other words, the missingness is conditional on another variable. Even though MCAR and MAR seem similar, they should not be used interchangeably as the key difference lies in the condition of the missingness. For instance, if a ME is not operational, data from dependant systems may not be recorded, for example, the TC speed or the EG temperature. Missing Not at Random (MNAR) refers to cases where the missingness of an observation depends only on the variable with the missing data; that is, the missingness is conditional to itself. This is inherently a complicated mechanism to identify. For example, MNAR could result when missing data originate from a data collecting sensor has an established failure during the time of the recording.

A very common imputation technique is called vertical imputation, which uses information from the same column as the one with the missing value. In this case, it is very common to carry forward the last value, or use mean, or mode of the observed values (Pigott 2003). As reviewed by Donders et al. (2006) and McKnight et al. (2007), vertical imputation is easy to implement; however, it is not suitable for successive missing points as it artificially increases the observations' frequency.

An alternative to vertical imputation is to impute the missing values with ones from similar parameters, or by using other logical rules. This is commonly known as horizontal imputation as information from the same record is used. Indicatively, this approach could be used to treat missing values between two identical pieces of machinery (e.g. pumps). The main drawback of this approach is that there may not be two similar parameters in the available dataset; therefore this approach is not always viable (Longford 2005; Gibert 2014)

Hot-deck imputation is an approach that is based on the similarity of a missing instance with a complete one, as initially suggested by Ford (1983), Rizvi (1983), and Roth (1994). This approach matches donors (i.e., instances with observed values) with recipients (i.e., instances with missing values). A pool of possible donors is formed based on the similarity between the recipient and the complete instances. The similarity is quantified using a variety of different metrics, including the Euclidean distance, Manhattan distance, Mahalanobis distance and maximum deviation. The main benefit of hot-deck imputation is that it does not rely on parameter specific models, hence the imputation is not influenced by any parameter selection. Also, as the imputation is based on actual values, the dataset is not completed by artificial ones. However, there is no explicit mathematical model behind the hot-deck methodology. Taking into account the merits of hot-deck imputation, it is worth examining in greater detail its most useful and widespread implementation, kNN. kNN can be an effective imputation tool on its own, and can easily be used in hybrid models. Moreover, kNN is a "lazy" algorithm, and as such, it does not require an explicit training phase; this dramatically reduces the required computational cost and increases the efficiency (Batista and Monard 2003; Armina et al. 2017). Also, the effectiveness of kNN imputation has been demonstrated in various examples, including Zhang (2012), Huang et al. (2017), Cheliotis et al. (2019), and Zhang et al. (2018).

Fitting a regression model to appropriate instances with recorded values is another widely used imputation method. Regression-based imputation is more complicated compared with vertical and horizontal imputation (Lang and Little 2016). A regression model is fitted between the target variable (i.e. variable with missing data) and the selected independent variables. The regression model can be linear, polynomial, or of another type, depending on the dataset. The resulting regression equation is then used to impute instances with missing points in the target variable (Enders 2001). As shown by Longford (2005), the general form of regression-based imputation, between the associated variables Y (with missing instances) and W (complete data set), is provided by:

$$Y = f(W) + \epsilon \tag{2.1}$$

In Equation 2.1, f(W) is some appropriate function (e.g. linear, polynomial, etc.) and ϵ is the error term which is used to account for the uncertainty.

Multiple Imputation (MI) represents a modern and more sophisticated approach to imputation. MI can increase the accuracy of the imputation while reducing bias. As discussed by Azur et al. (2011), MI allows for better accountability of the statistical uncertainty, as opposed to single imputations. MI is based on the improved use of predominant imputation techniques. Assuming an incomplete dataset Y, MI follows the subsequent steps:

1. Impute the missing values of Y m times.

- 2. Analyse separately the m different datasets.
- 3. Merge the m different results into one dataset.

MI depends on certain user-specified selection steps. Initially, the imputation method has to be specified; selection can include deterministic or stochastic methods. Next, the number of imputation cycles must be identified. Increasing the number of cycles can improve the model's accuracy; however, this comes at a computational cost. The final selection involves the determination of the concluding missing values from m different datasets. The selection can be facilitated in either a deterministic (mean, median) or a stochastic (random or probabilistic selection) way. All of the above choices are case- and application-dependant. Through the recent literature, it is seen that the Multiple Imputation by Chained Equations (MICE) approach is one of the most promising and accurate implementations of MI (Royston 2004). MICE is an effective imputation tool that can be used both as a stand-alone solution and in hybrid models. MICE aims to cyclically fit regression models to the different variables with missing data. This process is repeated for a predefined number of iterations, as the predictions for the missing data are gradually improved (Royston and White 2015). The effectiveness of MICE has been demonstrated in different research efforts including Cheliotis et al. (2019), Buuren and Oudshoorn (1999), and White, Royston, and Wood (2011).

2.6.3 Fault Detection (FD)

Following the identification of critical components and the data preparation stages, FD is often the subsequent process of a PdM framework. The area of FD has been rapidly expanding in the past decades and currently is commonly facilitated using various condition describing signals, as shown by Martinez-Guerra and Luis Mata-Machuca (2013), Sari (2013), and Sayed-Mouchaweh (2018). As reviewed by Jardine, Lin, and Banjevic (2006), FD includes a variety of different methods ranging from statistical to ML, all aimed at identifying the presence of a fault in the examined system.

As discussed in Isermann (2006), the most basic statistical approach for FD is through limits checking (e.g. maximum and minimum values). In contrast, more advanced approaches are built around the identification of specific trends (e.g. cyclic patterns, rates of change, etc.). The use of the EWMA for FD is an approach gaining popularity due to its versatility and accuracy and based on the identification trends in the examined signal (Garoudja et al. 2017). FD models based on the EWMA use selected signals and plot the signals' EWMA in a control chart, which contains upper and lower control limits for the detection of faults (Nounou et al. 2018). This type of FD creates easy to visualise models that can be used to detect various faults. For instance, Harrou et al. (2015) combined partial least squares with EWMA for the detection of faults in industrial processes. The effectiveness of this model was showcased through its ability to detect developing faults in distillation columns. Similarly, Badodkar and Dwarakanath (2017) developed a methodology based on EMWA for the detection of broken teeth in mechanical gearboxes. The EWMA analysed timeseries acceleration signals, which showed excellent performance in detecting faults in their early stages. Awad, AlHamaydeh, and Faris (2018) developed a method for the detection of structural damage in buildings, based on ANNs and control charts. Nounou et al. (2018) proposed a condition monitoring scheme for grid-connected photovoltaic panels. The scheme was based on the monitoring of environmental and performance parameters (voltage, current, and frequency) in an EWMA control chart. Finally, Adegoke et al. (2019) proposed the use of an EWMA-based FD methodology for the manufacturing sector. The effectiveness of the methodology was showcased in an example of a continuous stirred tank reactor.

As discussed by Ma and Jiang (2011), FD based on data-driven ML approaches are gaining popularity in a variety of applications, including the manufacturing, nuclear and offshore sectors. ML-based approaches for FD are traditionally based on classification algorithms, like SVM and Logistic Regression, as reviewed by Liu et al. (2018). However, these algorithms are restricted and not easily used in FD models trained on fault-free data only. Also, classification-based FD models are not easy to integrate with prognostic and diagnostic task (Hong et al. 2007).

FD based on Expected Behaviour (EB) models is an alternative approach to classification models (Hong et al. 2007). EB models are often used for FD tasks as part of a PdM framework in a variety of applications, including the offshore, automotive, nuclear and manufacturing sectors. The use of such models offers several advantages, as they replicate the normal behaviour of various signals, leveraging ML, and assesses any deviations from the normality to detect faults. Zaher et al. (2009) examined the use of an ANN for the development of an EB model for FD in wind turbines, based on operational data. The ANN was trained with more than three months of operational data and was used to monitor the condition of the turbine's gearbox. Similarly, Schlechtingen and Ferreira Santos (2011) examined the application of ANNs and polynomial regression models for the development of EB-based FD for wind turbines. The examined models showed good performance in detecting faults in the stator and gearbox of a wind turbine, by modelling the power, speed, and various temperatures. The same authors developed an ANN-based EB model for the detection of a variety of faults in wind turbines based on operational data. The networks were trained by using more than 30 months of operational data (Schlechtingen, Santos, and Achiche 2013; Schlechtingen and Santos 2014). Lastly, Bangalore and Patriksson (2018) studied the topic of optimal maintenance planning for wind turbines by using an ANN-based EB model for the detection of faults in critical components. As with the previous cases, the developed models were trained on readily available operational data.

2.6.3.1 FD Discussion

ML classification models are the standard choice for FD and offer accurate results when they are trained in the presence of faulty data. However, since in some cases labelled faulty data are not available for training, EB models are developed. EB models have a wide application for FD in many different sectors, as previously discussed. They detect faults by assessing a signal's deviation from an expected normal behaviour. The majority of these models employ ANN to create models which require large amounts of training data. Even though ANN exhibit good predictive behaviour, models that are less reliant on large training datasets, and are equally effective, ought to be explored. In addition, EWMA control charts have an increasing application for FD. They create visual models that can detect developing faults in the early stages. Moreover, EWMA-based FD can also define an envelope of normal operation, for a selected signal, using different control limits. Finally, EWMA control charts can be easily combined with other approaches to offer enhanced detection capabilities.

2.6.4 Diagnostics

The performance of diagnostic tasks is often one of the final processes in a PdM framework and is the following step after FD. Diagnostics aims to identify the root cause of a detected fault, anomaly or error (i.e. localisation). In general, diagnostic models can be divided based on either the type of data, or the types of models used, as seen in Figure 2.6.

Diagnostic efforts, when examined in terms of the data they use for the fault localisation, can be subdivided into a) ConMon, b) PeMon and c) ProMon, as also mentioned in Section 3.4.4. In brief, ConMon uses condition-describing parameters, including vibrations and acoustic emissions. PeMon is based on the use of



Figure 2.6: Initial division of diagnostic models

information relating to the output of the examined system, including power output. Lastly, ProMon uses information which reflect the internal processes of the examined system such as injection pressure and cooling temperature. Apart from providing insight into the type of data used, this division of diagnostic models is not very useful in explaining how faults are localised.

Diagnostic efforts when examined in terms of the types of models used can be further subdivided into a) physics-based models, b) data-driven models and c) knowledge-based models (Jardine, Lin, and Banjevic 2006; McKee et al. 2014), as seen in Figure 2.7. This type of division can be more useful in providing an



Figure 2.7: Further division of diagnostic models based on the specific approach

understanding of the working process of fault localisation.

Physics-based models use physical principles to create mathematical equations that describe the examined system and the respective failures (Nordmann and Aenis 2004). For instance, physics-based models can include thermodynamic, inertial and fluid models; however, the majority of machinery-based diagnostic models are based on thermodynamic principles (Jardine, Lin, and Banjevic 2006). Cocquempot and Izadi-zamanabadi (2006) created an inertial physics-based model for the identification and isolation of faults in centrifugal pumps driven by induction motors. Similarly, Murphy et al. (2015) developed a thermodynamic-based model for the isolation of faults of marine diesel engines under the scope of reducing the engine's environmental impact. Lastly, Theotokatos et al. (2018) developed a thermodynamic model based on extended mean values for the PdM of marine two-stroke engines.

Data-driven models use data-oriented methods to uncover patterns and behaviours that relate to specific faults in the examined systems. Data-driven models can be further split into ML or statistical-based. Statistical approaches can be based on hypothesis testing and other statistical tests and processes. For instance, Trachi et al. (2017) developed a diagnostic methodology for induction machines based on hypothesis testing and by using the likelihood ratio test. Also, Davarifar et al. (2013) developed a fault localisation methodology for photovoltaic systems based on statistical signal processing. Likewise, Weimer et al. (2013) employed an invariant hypothesis testing approach for the FD and diagnostics in Heating Ventilation and Air Conditioning (HVAC) systems. On the other hand, MLbased approaches use a plethora of both supervised and unsupervised methods (Galar Pascual 2015; Liu et al. 2018). Lu et al. (2001) examined the application of back-propagating auto-associative neural networks for diagnostics in combustion engines. Kim, Ball, and Nwadiogbu (2009) patented a methodology based on selforganising maps clustering for diagnostics in steam and gas turbines. Radionov et al. (2015) used subtractive clustering techniques for the diagnosis of faults in power transformers. Also, Giorgi, Campilongo, and Ficarella (2018) developed a PdM and diagnostic model gas turbines by using ANNs and SVMs based on the use of synthetic machinery data.

Lastly, knowledge-based models are examined which aim at mimicking specialists' reasoning, while effectively handling uncertainties and having increased modularity (Lin, Chen, and Zhou 2013; Diakaki et al. 2015; Atoui, Verron, and Kobi 2015; Cai, Liu, and Xie 2016b). In general, knowledge-based models have many implementations, but the more prominent approaches are based on Fuzzy Logic or BN (Nourian, Mousavi, and Raissi 2019; Chojnacki, Plumecocq, and Audouin 2019). Diagnostic models based on Fuzzy Logic have many applications, but they can be less modular and fast to set-up, compared with BNs (Nourian and Mousavi 2019; Wang et al. 2019; McKee et al. 2014). Riascos, Simoes, and Miyagi (2007) developed a diagnostic network, based on BN, for the diagnosis of different faults in a proton exchange membrane fuel cell. Also, Diakaki et al. (2015) developed a decision support system for merchant ships. This system addressed the issues of route optimisation and fault localisation leveraging BN. Atoui, Verron, and Kobi (2015) examined the use of a BN-based classifier for the detection and diagnosis of three different faults present in chemical process plants. Moreover, Zhao, Wen, and Wang (2015) and Zhao et al. (2017) created a multimode BN for the diagnosis of more than 27 faults in industrial air handling units. The developed networks demonstrate the versatility of BN, as they make use of data fusion. Similarly, the versatility and accuracy of BNs are demonstrated in the work of Wang et al. (2017), through the development of a diagnostic network for chiller units which can also handle classification problems. Amin, Khan, and Imtiaz (2018) looked at the development of dynamic BN for FD and root-cause analysis for chemical process plants, generating evidence for collected data under the assumption of a Gaussian distribution.

2.6.4.1 Diagnostics Discussion

Diagnostic methods can be divided either based on the type of data they use or based on the models they employ, with the latter categorisation providing more insight into the working process of the fault localisation tasks. Regarding the different types of diagnostic models, the majority of physics-based models for machinery diagnostics are based on thermodynamic modelling. Even though these models are accurate, they are time expensive, both in terms of the set-up time and the required maintenance of the models. Data-driven models, both statistical and ML, have good performance; however, they can employ blackbox tools (e.g. ANNs). Moreover, statistical diagnostic models have reduced modularity and interoperability. Lastly, knowledge-based models are developed to counter the difficulties of building mathematical models while providing accurate results. Knowledge-based BN is a versatile diagnostic tool that can effectively handle uncertainty. They are extremely popular in diagnostic tasks due to their compact nature, consistency, and modularity.

2.7 Maritime Predictive Maintenance

Maritime maintenance is a continually evolving dynamic area. Technological advancements, complex managerial schemes and strict safety and reliability criteria, are all factors that contribute to the development of marine maintenance (Borjalilu and Ghambari 2018). In addition, advancements in maintenance in other industries (e.g. manufacturing, nuclear, automotive and renewables) are having a positive influence on the development of maritime maintenance (Lazakis and Ölçer 2015; Cheliotis et al. 2019). The following sections will present the maritime predictive maintenance processes, summarising the state-of-the-art and presenting in detail the relevant developments.

This section will discuss the status quo in the maritime industry in terms of critical equipment selection, data preparation, FD and diagnostics. Regarding the identification or selection of critical equipment, the available literature is limited, especially when comparing to other industries. In general, the identi-

Ship Maintenance

fication of critical equipment is most often performed using FTA and FMECA, and in some cases based on BN. However, there are some cases that are based on accident's report analysis. In more detail, Ersdal and Kvitrud (2000) identified critical equipment with respect to green-water ingress in product carriers. The critical equipment was identified as a result of a post-hoc analysis of green water ingress reported in the studied ships. Next, Lazakis, Turan, and Aksu (2010a) combined the use of FMECA with FTA in a methodology for critical components identification, applied in the electric generation system of a cruise ship. Moreover, Anantharaman et al. (2014) developed a methodology for the identification of critical components in the main propulsion system of a merchant ship. The methodology was applied in the lubrication sub-system and was based on the use of FTA. Also, Dikis, Lazakis, and Turan (2014a) developed a methodology for the identification of critical equipment in ship systems as a part of a broader risk assessment framework for ships. The developed methodology was based on reliability analysis using FTA and BN. Lastly, and most recently, Lazakis, Raptodimos, and Varelas (2018a) identified critical components in ship systems by also using FMECA and FTA.

Regarding the imputation of missing data, applications in the maritime domain also have limited literature. In general, imputation in the maritime industry is predominantly examined under the scope of maritime operations, including traffic management, logistics and situational awareness. As a result, most applications really on operational, non-dense datasets. In detail, Iphar, Napoli, and Ray (2015) used an imputation method for maritime data originating from Automated Identification Systems (AIS). The treatment of the missing data is part of a broad data-driven methodology aimed at increasing situational awareness in sea-passages. Similarly, Claramunt et al. (2017) used the imputation of missing data as part of a traffic management methodology aiming to increase ship safety and security. Also, Fruth and Teuteberg (2017) used data imputation under the scope of optimising maritime logistics to improve safety. Lastly, Dobrkovic, Iacob, and Hillegersberg (2018) also used imputation on AIS data in a methodology to improve barge logistics.

In contrast with critical components identification and data imputation, the literature on maritime FD is much richer. The state-of-the-art of maritime FD can be divided into EB models based on ML, ML models based on classification, methods leveraging RA tools (e.g. BN, FTA) and physics-based models. For example, the benefits of predictive maintenance have been addressed under a decision support framework using fuzzy-sets enhanced with AHP (Lazakis and Olcer 2015). Also, Ahn et al. (2017), examined the use of a fuzzy-based FMEA approach to study the risk profile of the gas turbine system of specialised tankers. Similarly, Cem Kuzu, Akyuz, and Arslan (2019) proposed the use of a fuzzy-based FTA to analyse the inherent risks of ship mooring operations. Dikis, Lazakis, and Turan (2014b) examined the use of data-driven dynamic BN for the maintenance prioritisation of multiple ship systems. Expanding on this, the coupling of BN with data mining and Markov Chains (MC) has been studied for the development of a predictive maintenance scheme of marine ME and their supporting systems (Dikis and Lazakis 2019). The application of a regularised feed-forward ANN classifier for the monitoring of the EG valve of a marine two-stroke engine, using acoustic emissions signals, has also been examined (Fog et al. 1999). Also, Li et al. (2011) examined the use of a back-propagating ANN classifier for the condition monitoring a marine gearbox, based on the spectrum analysis of a vibration signal. Similarly, the use of a three-layer feed-forward ANN for the condition monitoring of the air intake and fuel injection system of a medium speed marine engine has been examined (Basurko and Uriondo 2015). Raptodimos and Lazakis (2018) and Lazakis, Raptodimos, and Varelas (2018b) examined the application of ANNs and their combination with Self Organising Maps (SOM) and interclustering for the monitoring, prediction and healthiness assessment of a marine ME. Lazakis, Gkerekos, and Theotokatos (2018) demonstrated the use of SVMs for the classification of faults and the development of a data-driven normality index for a marine generating engine. Similarly, Zhan et al. (2007) and Zhan, Shi, and Liu (2007) examined the use of a multi-class SVM for the fault diagnosis of marine ME cylinder covers, based on vibration analysis and Principal Components Analysis (PCA). Lastly, the combination of data simulation, through physical modelling, with both supervised and unsupervised ML algorithms has been examined with application to system decay in naval vessels (Cipollini et al. 2018; Coraddu et al. 2016).

Moving away from FD, the area of maritime diagnostics for shipboard systems is minimal and underdeveloped. The state-of-the-art of maritime diagnostics can be divided into approaches based on thermodynamic modelling and simulations and on ML-based approaches. For instance, Silva et al. (2018) developed a methodology for the diagnosis of faults in the electric drive system of electric-powered ships by employing two-dimensional wavelet transforms from sensor data. Moreover, Campora, Cravero, and Zaccone (2018) combined an ANN with thermodynamic modelling, for data simulation, in order to diagnose faults of a naval gas turbine. Korczewski (2016) investigated the use of the ME EG Temperature (EGT) in thermal engine models for the diagnosis of internal engine faults. Also, Homik (2010) developed a methodology for FD and diagnostics of torsional vibration dampers and marine ME crankshafts. The methodology is based on the combination of vibration analysis with statistical testing. Lastly, Ranachowski and Bejger (2005) used a wavelet analysis in acoustic signals to diagnose the most common faults of the fuel injection sub-system of a marine diesel engine.

2.8 Identified Gaps

From the previously cited literature, and mainly by comparing the maritime predictive maintenance status quo with other industries, the following conclusions regarding maritime critical equipment selection, data preparation, FD and diagnostics can be made. Based on these gaps, the contribution of this work is developed and the novelty is directed. The contribution of this work, concerning maritime predictive maintenance processes, is summarised in Table 2.4. To highlight the discussed contribution, Table 2.4 also summarises the state-of-the-art (status quo) of the maritime predictive maintenance processes, also detailed in the previous section.

Firstly, predictive maintenance, in the maritime sector, is at its infancy and is severely lagging compared to other industries. There is a gap in a complete predictive maintenance framework taking into account the particular needs of the maritime industry and providing data-driven and knowledge-based solutions. However, data-driven solutions are heavily influenced by the characteristics of the available data (i.e. size, density and quality). In more detail, the age of the examined vessel plays a vital role in the data characteristics. Older vessels tend to have sparser data, due to the absence of modern DAQ systems, in contrast to newer ships which are fitted with DAQ systems during their newbuilding stage. Also, the operating environment of ships can reduce the quality of the gathered data, due to loss of sensor calibration, the loss of sensor connectivity, and the inherent difficulty to replace failed sensors. Lastly, the lack of proper data storage facilities, coupled with the possible confidentiality of the data, create issues unique to the maritime sector.

Critical equipment selection methods in the maritime industry are very simplistic, as seen from the presented literature. They typically rely on some combination of RA tools, including FTA, FMEA, or FMECA. Consequently, there is a gap in the incorporation of data-driven tools for the identification of critical components. Similarly, there is also a gap in combining cost aspects with reliability characteristics for identifying critical components. As a result, the contribution of this work includes the combination of ML approaches with FTA and the incorporation of costs with reliability IMs for the identification of critical equipment (Table 2.4). This ensures that critical equipment are selected based on a subjective and multi-criteria approach.

Regarding data preparation, there is a substantial lack of a formalised approach in the maritime industry. Despite the growing popularity of modern data analytics, attempts for outliers detection and imputation based on modern ML are scarce. The lack of a formalised and modern imputation approach for the maritime industry is very concerning, as datasets used for maritime predictive maintenance contain from 4.4% to 26% missing values, depending on the application (Lazakis, Gkerekos, and Theotokatos 2018; Tsitsilonis and Theotokatos 2018). The same need is demonstrated when considering the increased use of ML algorithms which are sensitive or restrictive to missing values, as discussed in Cheliotis et al. (2019). Lastly, the urgency for the above can be seen from similar attempts in other sectors, including the offshore wind industry, as discussed by Martinez-Luengo, Shafiee, and Kolios (2019). A major gap in the maritime industry is the lack of a formalised and accurate approach for the imputation of missing data that includes all necessary imputation preparatory steps, and any further post-imputation processes have not yet been suggested. Consequently, this work contributes with the development of an imputation approach for maritime ProMon data, combining data-driven and knowledge-based approaches (Table 2.4). This safeguards ProMon datasets, as otherwise missing information is preserved.

In terms of FD, there is a gap in the application and use of FD models addressing the particular needs and requirements of maritime predictive maintenance. Moreover, it is seen that the majority of the EB models for FD are based on ANNs. Even though ANNs offer good prediction results, they require large datasets for training, which usually are not always available within the maritime industry, and have a more complex model development phase. In addition to that, ANNs are black-box approaches, which makes it difficult to impart domain knowledge as there is no insight into how outputs are produced. Also, despite the numerous applications of ANNs, there is an upcoming trend among data science practitioners to increase the applications of white-box approaches, as discussed by Loyola-Gonzalez (2019). Consequently, the selection of the underlying approach for the EB models should serve the application, the data available and take into account the above issues. In cases of limited data and when accurate and fast results are required, regression-based EB models should be examined. Regressionbased models, such as the ones presented in this work, offer a white-box approach to EB modelling. This allows for the creation of models that are easy to explain, interpret and impart domain and previous knowledge. This is achieved as the form of models allows the examination of the influence each predictor has on the output. The selection of the EB model approach for the detection of faults is also beneficial when compared to the alternative classification approaches. Firstly, with EB models, there is more flexibility in the selection of the underlying algorithms used. With classification approaches, in the absence of observed faulty data, one-class SVM is the standard choice, with limited alternatives. In contrast with classification, EB models have greater flexibility in the selection of the algorithm (e.g. ANNs, polynomial regression). Moreover, the output of the EB models (i.e. a time-series) is more interpretable and useful for future tasks (e.g. diagnostics), when compared to the output of classification approaches (i.e. decision space). Therefore, the contribution of this work includes the investigation for the optimal regression model, used for EB modelling combined with EWMA control charts for FD (Table 2.4). This creates FD models that can be developed without labelled faulty data, improving their flexibility and bypassing the discussed issues of black-box ANN modelling.

Concerning diagnostics, it is observed that most such attempts are physicsbased. Even though these models are well-performing, they are time-consuming to develop and apply. Likewise, data-driven diagnostic models also exhibit good behaviour, but they depend on extensive training datasets, which are scarce in the maritime industry. On the contrary, knowledge-based diagnostics, including BN, offer accurate performance, as seen in this work, without requiring lengthy set-up times. Also, knowledge-based diagnostics do not require dedicated training phase, which simplifies their development. Moreover, knowledge-based diagnostics are modular, which improves their compatibility with FD modules and makes it easier to expand in other engineering systems. As a result, the contribution of this work is the integration of ML-based FD model with a BN for diagnostics (Table 2.4). This allows for the investigation of the root-case of a detected fault.

Critical Equipment Selection	
	Based on RA tools (FTA_BN_EMECA)
Status quo	No combination of costs with reliability matrice
	Contraction of costs with reliability metrics
Added Contribution	Combination of ML clustering with FTA
	Examination of costs with reliability IMs
Data Preparation	
Status quo	Developed for ship operations application
	Majority of purely data-driven models
Added Contribution	Imputation of ProMon data
	Combination of data-driven and knowledge-based
	approaches
Fault Detection	
Status quo	Based on ML classification and ML EB models
	Thermodynamic models and RA tools
	are also used
Added Contribution	Combination of ML-based EB models with EWMA
	control charts
	Investigation of optimal regression model
Diagnostics	
Status quo	Based on ML approaches
	Thermodynamic models and simulations
	are also used
Added Contribution	Integration of ML-based FD with BN for diagnostics
	Combination of data-driven with knowledge-based
	approches
	approxime.

Table 2.4: Status quo and added contribution

2.9 Chapter Summary

This chapter presented the factual and critical review of the relevant literature. Initially, the fundamental maintenance concepts and maintenance frameworks were presented. Afterwards, the reliability assessment and data science tools used in predictive maintenance together with the resulting processes were examined. Then, the predictive maintenance processes of critical equipment selection, data preparation, FD and diagnostics in the maritime field were compared with those in other domains. Consequently, this comparison allowed to uncover several gaps for maritime applications which then oriented the novelty of the present thesis.

Chapter 3

Methodology

3.1 Chapter Overview

After the gaps of the literature are identified, through the critical examination of the relevant literature, this chapter aims to present in detail the overall framework, while highlighting the generated novelty. The framework aims to address maritime predictive maintenance holistically, by including topics ranging from critical equipment selection to system diagnostics. Section 4.2 gives a detailed outline of the generated novelty, and Section 4.3 provides a high-level description of the overall framework. The methodology addressing the critical equipment selection is presented first. Then, the data preparation methodology is shown. Finally, the methodologies for the FD and diagnostics are presented in turn.

3.2 Novelty

By considering the gaps in Section 3.9, this thesis aims to propose a novel framework, holistically addressing the requirements of maritime predictive maintenance. Thus, the proposed framework combines in a novel manner the identification of critical equipment, data preparation, FD and diagnostics. In more detail, the novelty of this work is listed below:

- This work is based on the novel combination of the developed methodologies to address the issues of maritime PdM. This work combines FTA and data mining for the identification of critical components, with the development of a *k*NN and MICE-based imputation approach, coupled with the use of regression and EWMA based FD which leads into a BN-based diagnostic tool. The developed novel methodologies, addressing the different needs of maritime PdM, can be used in series and in a single vessel, conditional on the availability of the required data.
- To address the limitations regarding the selection of critical equipment, a novel methodology is developed. The methodology combines proven RA tools with data-driven efforts. In more detail, FTA analysis is performed to obtain specific reliability Importance Measures (IM). Then, the IMs are clustered together with repair and replacement costs, using the *k*-means algorithm, to obtain the resulting critical components.
- To tackle the gaps regarding data preparation, the novelty of this study lies within the proposal of a new hybrid imputation method that combines data-driven solutions with valuable First Principles (FP) domain knowledge. This approach is shown to yield more accurate results compared to traditional, application-agnostic, imputation methods, as it will be shown in the following sections. Moreover, alongside the hybrid imputation method, all the needed pre-imputation and post-imputation steps are presented.
- In terms of FD, the novelty of this thesis lies in the combination of the pre-processing steps, with the regression-based EB modelling and EWMA control charts for FD. Moreover, the novelty includes a systematic and structured examination for the selection of the optimum regression model. This encompasses the selection of the ideal predictor variable and types

of training datasets (recorded data vs shop test data). The developed FD methodology also includes the combination of data-driven models with engineering analysis, resulting in interpretable and effective models using real data.

• Lastly, with respect to diagnostics, the novelty of this work lies in the development of a novel framework that combines an ML-based FD module with a BN-based diagnostic module. The FD module is combined in a novel way with the diagnostic module which includes the mapping of faults and the construction of a BN. Evidence of detected faults is propagated in a BN network. The output of the BN diagnostic is the quantified probabilities of the mapped faults, together with the fault profiles of different failure modes.

3.3 Overview of the Novel Framework

This section gives a detailed account of the proposed novel framework. As mentioned above, the main aim of this framework is to address the topic of maritime predictive maintenance in a comprehensive and robust manner, by individually addressing the topics of critical equipment selection, data preparation, fault detection and system diagnostics. Each of these methodologies is developed and assessed individually, as described in detail in Chapter 5.0. Despite this sequential development, all of these methodologies are integral parts of the proposed PdM framework.

Figure 3.1, illustrates the overall novel framework by specifying the sequence of the developed methodologies together with the required inputs and expected output in each case. The first step of the framework is the identification of the critical equipment of ship systems. This is a crucial first step, and forms the starting point of the proposed novel framework, as it allows to focus the
maintenance efforts on components which have the maximum impact on safety, reliability and availability. As seen in Figure 3.1, this methodology requires the use of the maintenance schedule and repair costs to identify the critical equipment by using the novel combination of FTA with k-means clustering. Once the critical equipment selection methodology is developed, the topic of data preparation is addressed. The data preparation methodology is developed to ensure that the knowledge extracting potential of the used data is maximised. As depicted in Figure 3.1, this methodology is based on the use of ship performance data to detect outliers and impute missing values by employing the novel combination of the MICE and kNN ML algorithms. Then, the FD part of the framework is developed, which aims at the early detection of developing faults. The FD is facilitated by using shop tests and ProMon data in a novel combination of regression-based EB models with residuals-based EWMA control charts. The last part of the framework includes the development of a diagnostics methodology. This methodology is developed to identify the root-cause of detect faults, as shown Section 5.5. As seen in Figure 3.1, the diagnostics are based on aggregated results from the FD part, operating manuals and data banks. This novel methodology also requires a fault mapping process to create a diagnostic BN, which is used to calculate the probabilities of mapped faults, based on evidence of faults from the FD part.

3.4 Critical Equipment Selection

The critical equipment selection is the first part of the proposed framework. This first part of the proposed novel framework includes a novel methodology for the identification and selection of critical ship equipment. This is a process that takes into account both criticality indices and cost-related information. The aim is to create a methodology that allows for the systematic identification of critical

Ship Maintenance



Figure 3.1: Layout of the novel work, depicting the different methodologies, the required inputs and the derived outputs of each part.

components of ship systems. The developed methodology, not only examines components with respect to their importance on the mission of the system they belong to (criticality), but also with respect to the financial implications of their failure on the ship operation.

3.4.1 Overview

The critical equipment methodology initiates with a data collection effort, which aims at fulfilling the methodology's data requirements, as seen in Figure 3.2. The collected data include information regarding the maintenance schedule (i.e. maintenance interval) of the examined vessel and the repair costs of the machinery. These data were selected for collection after examining the data requirements of similar research efforts, such as in Lazakis, Turan, and Aksu (2010a) and Lazakis, Raptodimos, and Varelas (2018b). After the required data are collected, the ship system analysis step initiates. This step allows the examination of the ship in terms of its systems, sub-systems and components, a process which is based on first-principle knowledge. This is also the stepping-stone for the FTA, as the structure of the Fault Tree (FT) is based on the analysis of the ship systems.

The FT is quantified by using the Mean Time Between Failures (MTBF) of each of the components. The MTBFs are extracted from the maintenance schedule of the vessel, and they represent one of the most common ways for quantifying FTs. Once the FT is quantified the analysis initiates by selecting a calculation method, as further explained in Section 4.4.4.3. As a result, the Birnbaum IM (I^B), Criticality IM (I^{CR}) and Fussel-Vesely IM(I^{FV}) for each component are obtained. The IMs for the depicted components are analysed, and the most appropriate are selected for the next step of the methodology.

Then, the cut-sets of the FT are obtained and the events in the lower order cut-sets (i.e. the smallest sets of events that can lead into failures) are identified for the next step of the analysis. The selected IMs for these components, together with the associated repair costs are clustered using the k-means algorithm. The use of the k-means clustering algorithm allows for the creation of a visually descriptive model that categorises the data based on their criticality and cost. Therefore, to avoid added complexity and to maintain the model's practical application, the clustering algorithm is limited to three dimensions. This allows the identification of the critical components, which belong to the cluster of components that have the highest values for the selected IMs and the highest repair costs. The critical components selection methodology of the proposed framework is demonstrated in Figure 3.2 and are presented in detail in the following sections. 1. Critical Equipment Selection



Figure 3.2: Layout of the critical components selection methodology of the proposed framework

3.4.2 Data Collection

Data collection is the initiating step of the critical equipment methodology. This step ensures that the required data to quantify the FT (i.e. MTBF) are available. To this end, the entire maintenance schedule and maintenance plan of the studied vessel is obtained. Moreover, the repair costs for the various ship components are collected, as required for the identification of the cluster of critical components. Such data are usually available from ship owners and ship operators.

3.4.3 Ship Systems Analysis

Following the data collection, the ship systems analysis step takes place. This step allows the examination of the ship in terms of its systems, sub-systems and components. Also, the identified systems, sub-systems and components are organised in a table format, which helps with the determination of the structure of the Fault Tree (FT). This is a very common starting step when performing FTA in engineering systems, as discussed in Bertsche (2007). As also mentioned above, the ship systems analysis is the stepping-stone for the creation of the structure of the FT.

3.4.4 Fault Tree Analysis (FTA)

FTA is a well-established and flexible RA tool that can effectively model complex systems, as described in Section 3.5.1.5. The use of FTA in the developed methodology serves two functionalities. First, the qualitative part of the FTA (i.e. the FT structure) gives a clear depiction of the reliability dependencies between different systems and sub-systems of the ship. Then, the calculation of the different IMs (i.e. the quantitative part) allows for the numerical assessment of the criticality of the various components. In summary, the FTA aims to perform a pictorial and quantified representation of how sub-systems (intermediate-gates) and components (basic-events) can lead into the loss of the reliability of the broader systems they influence (top-gate) (Verma, Ajit, and Karanki 2010).

3.4.4.1 Examined Gates and Events

As mentioned above, the structure of the FT was derived as a result of the ship system analysis, as demonstrated in Figure 3.3. The identified systems and sub-systems are represented in the FTA using gates, and the components are described as basic events. The gates used to connect the systems with their respective sub-systems and components were selected based on the different functional dependencies (Verma, Ajit, and Karanki 2010).

One of the most common gates is the "OR" gate, which requires a minimum of two inputs. When used, the outcome of the gate occurs if any of the inputs occur. In other words, the "OR" gate is represented as a Boolean union. In



Figure 3.3: Depiction of the FT structure, as created from the ship systems analysis

engineering systems, "OR" gates are used to represents systems and sub-systems which are prone to failures and have limited fail-safe capabilities. Assuming an examined output A with a probability of occurrence P(A), connect through an "OR" gate with n inputs $A_1, A_2...A_n$, then P(A) is obtained by Equation 3.2 (Relex Software Corporation 2003).

$$P(A) = P(A_1) \cdots P(A_n) \tag{3.1}$$

Another very common gate is the "AND" gate, which also requires a minimum of two inputs. If only one input is given to an "AND" gate, the gate's logic is negated as there are not enough inputs for comparison and the output of the gate is conditional only to the single input. In other words, single-input "AND" gates behave like "OR" gates. The outcome of an "AND" gate occurs only if all the inputs occur at the same time. In terms of Boolean logic, "AND" gates are represented as sets intersection. In engineering systems, "AND" gates are used to model systems and sub-systems which are less prone to single-point failures, commonly by having redundant components. Assuming a different examined output A with a probability of occurrence P(A), connect through an "AND" gate with different n inputs $A_1, A_2...A_n$, then P(A) is obtained by Equation 3.1 (Relex Software Corporation 2003).

$$P(A) = 1 - [1 - P(A_1)][1 - P(A_2)] \cdots [1 - P(A_n)]$$
(3.2)

The "VOTING" gate represents another widely used way of modelling systems and sub-systems with partial fail-safe capabilities. In engineering systems, "VOTING" gates are used to model systems which are tolerant to single-point failures but are not as robust as the systems modelled with "AND" gates. The output of a "VOTING" gate occurs when a predefined number, J, of the Yinputs occur. When J = 1 the "VOTING" gate behaves like an "OR" gate, whereas when J = Y the "VOTING" gates acts like an "AND gate". Assuming an examined output A with a probability of occurrence P(A), connect through a "VOTING" gate requiring two of the three inputs A_1, A_2, A_3 , then P(A) is obtained by Equation 3.3 (Relex Software Corporation 2003).

$$P(A) = P(A_1 \cap A_2) \cup P(A_1 \cap A_3) \cup P(A_2 \cap A_3)$$
(3.3)

The identified components are represented as "BASIC" events. These events are located at the lowest level of the FT structure, and they represent software, hardware and component failures. "BASIC" events are the most common way of representing component-based failures and have the widest applicability (PTC Windchill 2019). The "BASIC" events are quantified by using failure statistics for the respective components. This information is supplied prior to the initiation of the analysis, and they are in the form of MTBFs, failure rates or probabilities of failures (Relex Software Corporation 2003). During this work, the events are quantified using the MTBFs as they are obtained from the collected maintenance schedule. More information on the available gates and events is available in Appendix A.

3.4.4.2 Qualitative Analysis

As soon as the structure of the FT is specified and the "BASIC" events are populated with their MTBFs, the main analysis initiates. The analysis can be performed both qualitatively and quantitatively, depending on the application and the availability of data.

Quantitative analysis is based on Minimal Cut-Sets (MCS), which are obtained by leveraging gate logic (Ruilin and Lowndes 2010). A cut-set is a set of "BASIC" events, the occurrence of which can cause the top-event to occur. The use of cut-sets allows for the identification of weak points in large and complex systems and is based solely on gate logic. In other words, cut-sets can order and prioritise the examination of different events. A basic event that belongs in a cut-set can provide information regarding single point of failures. An MCS is the smallest set of events which must co-occur for the top-event to occur (Lazakis, Raptodimos, and Varelas 2018b). In general, MCS can be used as a starting point for the analysis of the examined system. Assuming a rudimentary FT with a single "OR" gate and inputs the events A_1, A_2, A_3 , then three MCS are obtain. The first is C_1 which includes event A_1 , the second one is C_2 which includes event A_2 and the third one is C_3 which includes event A_3 . Lastly, cut-sets can be organised in terms of their order, indicating the number of events in each cut-set. In general, cut-sets of lower order are more important and which also reflects to the individual events in them (Shafiee, Enjema, and Kolios 2019).

3.4.4.3 Quantitative Analysis

Quantitative analysis can be performed when all the "BASIC" events are populated with their respective failure statistics. In quantitative analysis, there are different calculation methods available. Each calculation method influences how the probabilities of the different gates are calculated.

The first option is the use of simulations by using Monte Carlo models. Simulation models are intuitively easy to understand, but they can be time-consuming (Relex Software Corporation 2003). Simulations generate random numbers associated with each event and then determine whether that event has occurred or not. Based on that, the status of the top-event is calculated. Simulation models are robust in their performance; however, they are time-consuming, and their performance may be comparable with less expensive models (PTC Windchill 2019).

Another approach for performing the reliability calculations of different gates is the cut-set summation method. The reliability calculations for each gate are obtained by adding the probabilities of the cut-sets of that gate (Relex Software Corporation 2003). In more detail, the probabilities of each cut-set are calculated as the product of the respective events. This approach is fast and accurate; however, its use is only recommended in FTs with low failure rates quantifying the "BASIC" events (Lazakis 2011).

Another calculation approach based on cut-sets information is the Esary-Proschan (EP) method. This method performs the reliability calculations for the top-event by defining an upper and lower reliability limit (Relex Software Corporation 2003). Since the EP method is based on the cut-sets, it has reduced computational times. Nonetheless, the application of this method is based on the assumption that each failure is the result of a gradual degradation (PTC Windchill 2019). As a result, failures affected by external influences cannot be modelled. Also, this method requires that every "BASIC" event appears at least in one MCS (Lazakis 2011).

Another calculation method that is based on cut-sets is the cross-product approach, which is very similar to the cut-set summation method (Relex Software Corporation 2003). This approach performs the reliability calculations by using the summation and product terms of the cut-set probabilities of the FT(Lazakis 2011). Since the cross-product is based on cut-sets, it has reduced computational time. However, it is based on the assumption that the simultaneous occurrence of multiple cut-sets is not probable (PTC Windchill 2019).

Finally, the exact calculation method is examined, which is purely based on gate-logic and does not use cut-sets information (Relex Software Corporation 2003). This approach performs the reliability calculations in all the gates on the FT. The main advantage of this approach is its accuracy due to the limited assumptions; however, this comes at the expense of the required computational time (PTC Windchill 2019).

From the presented quantitative calculation methods, the latter is applied. The main benefit of the exact calculation method is the minimal assumptions it uses and its accurate performance. Also, since this work did not have strict time constraints, the increased computational time of this method was not a concern.

3.4.4.4 Importance Measures (IMs)

The reliability IMs represent another useful functionality of FTA. Unlike the quantitative calculation methods, which examine the reliability of the top and intermediate gates, and the MCS which group different events based on the influence to the top-gate, the IMs examine each of the event individually. The main goal of the IMs is to identify the events whose improvement will have the most positive influence on the top-event and the intermediate gates. In other words, IMs can identify the critical components by ranking the different "BASIC" events in the FTA. In the present thesis, the I^B , I^{CR} and I^{FV} are examined.

The I^B was first introduced in 1969 as a method of numerically ranking the importance of a system's components and remains as one of the most widely used IMs. The I^B measures the probability of a component (event) being reliable from the failure of its system. It must be clarified that the Birnbaum IM of a component is not a function of the reliability of the components. The I^B measures the rate of change of the probability of the top-gate as a function of the availability of the examined "BASIC" event. The result of the Birnbaum IM is a ranking of events which can be used to decide which end-events need to be improved (Lazakis, Turan, and Aksu 2010b). Assuming a top-gate X and a "BASIC" event A, the I^B for A can be calculated as the difference of the probability of X given that the event A did occur minus the probability of X given that the event A did not occur (Equation 3.4).

$$I^{B}(A) = P(X|A) - P(X|A')$$
(3.4)

The criticality IM is another method for ranking the events of an FT. The goal of the ranking process is to assess how much the failure of an event influences the gate they participate in. The I^{CR} examines the global probability of the top-gate occurring due to the occurrence of the considered event. In other words, the I^{CR} calculates the probability that an examined event is critical for the entire system and will occur if the top-gate occurs. Assuming a top-gate X and a "BASIC" event A, the I^{CR} for A can be calculated according to Equation 3.5.

$$I^{CR}(A) = P(X|A) - P(X|A')\frac{P(A)}{P(X)}$$
(3.5)

As can be seen from Equation 3.4 and Equation 3.5, the criticality IM is a modified version of the Birnbaum IM, as it is adjusted for the relative probability of the basic event A. This is done to reflect the possibility of occurrence of event A and at the same time how possible it is to improve it. Consequently, in contradiction with the Birnbaum IM, the criticality IM focuses only on the link between the event A and its gate. It is not necessarily tied with the failure of the top event, even though the occurrence of A may affect it (Lazakis, Turan, and Aksu 2010b).

The last importance measure that is considered is the I^{FV} , which measures the overall percent contribution of cut sets containing an event of interest to the top-gate failure. The event under consideration is not the most critical one; however, it can cause the entire system to fail. The Fussell-Vesely IM is expressed as the ratio of the probability of occurrence of any cut sets containing the event A and the probability of the top-gate. The I^{FV} is calculated based on Equation 3.6

$$I_i^{FV}(A) = \frac{1 - \bigcap_{j=1}^m [1 - P(M_j(t))]}{1 - R_s[r(t)]}$$
(3.6)

where:

 m_i : the number of MCS containing i

 $\cap_{i=1}^m$: a MCS

 $M_j(t)$: the j^{th} MCS of those containing i, at a time t

 R_s : the system reliability

r(t): an end event occurring at time t

As briefly mentioned in Section 4.4.1, the selected clustering algorithm is restricted to three dimensions, to preserve the model's practical application, visual interpretability, while controlling the model's complexity. Therefore, from the calculated IMs, two of them are selected and combined with the repair costs for the clustering analysis. The use of two IMs provides a more detailed understanding of the criticality of each component, compared with the use of only one IM. The selection of the IMs is based on the theoretical background of each IM. A disadvantage of the I^{FV} , compared with the other ones, is its usage of MCS. Even though MCS offer good qualitative results, they do not take into account the failure statistics of each component. On the contrary, the I^B and I^{CR} are not based on MCS and instead take into account the failure statistics of each event. Even though this process may increase the required computational time, it can have a positive influence on the accuracy of the analysis. As a result, from the presented IMs, the I^B and I^{CR} propagate to the clustering analysis.

3.4.5 Clustering Analysis

The clustering analysis is the final step of the critical equipment selection. As mentioned above, the critical equipment are identified based on cut-sets, the selected IMs and the repair costs for the components modelled in the FTA. To control the complexity of the clustering analysis, while maintaining the visual interpretability, the analysis is restricted to three dimensions. In that way, the selected components are items with both high criticality and high repair costs. The failure of these items would cost the most, both in terms of their replacement and the resulting unavailability of the system.

The clustering analysis uses three inputs, two of the calculated IMs and the repair costs for the components. The resulting three-dimensional plot is used to identify the cluster whose centroid has the biggest distance from the origin of the axes. The components in that cluster are identified as critical. This visual representation of the critical components creates a model that is easy to interpret.

For that purpose, the k-means clustering algorithm is used, which is one of the most popular unsupervised ML algorithms. The k-means algorithm has the widest applicability of all the clustering algorithms due to its performance and simplicity (Yu et al. 2018). This clustering algorithm is optimum when the data are divided into distinct groups. This assumption is present in this methodology since the components can be divided based on their ranging criticality (Dikis and Lazakis 2019). Lastly, the k-means algorithm was selected as it is the most common option when dealing with non-oddly shaped data (Müller and Guido 2015).

K-means groups similar points together by looking for a fixed number of k clusters in the data. This algorithm iterative assigns the data into the best suited of the k clusters (Wu 2012). The aim of k-means is to minimise the intra-cluster variance of the data, as seen in Equation 3.7, where k is the number of clusters,

 μ_i is the mean values of the i^{th} cluster and x_j is the j^{th} data-point.

$$J = \sum_{i=1}^{k} \sum_{j=1}^{n} \|x_j - \mu_i\|^2$$
(3.7)

To initiate this algorithm, the number of clusters, k, is selected. Then, k points are randomly selected to act as μ_i for the first iteration of the algorithm. All the data-points are then assigned to a cluster, based on their proximity to the clusters' centroids (Equations 3.7). Lastly, the new centroids of the k clusters are calculated according to Equation 3.8, where S_i is the set of all the points assigned to the i^{ith} cluster.

$$\mu_i = \frac{1}{|S_i|} \sum_{x_i \in S_i} x_i \tag{3.8}$$

This process is repeated, until there is no change in the data-points assigned in each clusters in consecutive iterations of the algorithm.

3.5 Data Preparation

Once the critical equipment selection methodology is developed, the data preparation methodology of the proposed framework follows. This methodology is developed independently and as explained in Section 5.3, is tested on a different case study with different data. This part of the proposed novel framework, includes the development of a novel hybrid imputation method based on data-driven efforts and FP knowledge, as discussed in Section 3.8.1. The data preparation methodology ensures that the data used in the subsequent FD and diagnostic tasks are pre-treated and have reached their full knowledge-extracting potential. This mainly includes the imputation of any missing data, together with the necessary pre-and post-imputation steps. In the maritime industry, and especially regarding PdM, missing data can cause many problems. Notably, missing data can lead to inaccurate maintenance scheduling, which can cause machinery failures and possibly accidents. Data collection sensors, in shipboard systems, are prone to generating missing values due to a plethora of factors. Such factors can include the loss of calibration of the sensors due to external influences, sensor anomalies, the loss of sensors' connectivity, and the difficulty to replace failed sensors due to environmental complications. Consequently, the preparation of data and the imputation of the missing values is of paramount importance in PdM frameworks.

3.5.1 Overview

The data preparation methodology is initiated with the data collection step, which differs from the previous methodology. As with the critical equipment methodology, the data collection step ensures that the required data for the development of the methodology are available. The next step is the preliminary analysis of the gathered data. This step includes four processes which are designed to prepare the collected data for imputation. These four processes include the form handling, synchronisation, filtering and correlation examination of the data. Once the preliminary analysis is completed, the imputation process, which is the main focus of the data preparation methodology, takes place. The imputation process includes the imputation of missing values through the use of the novel hybrid method based on kNN and MICE. Moreover, the imputation performance of the novel hybrid method is assessed through the use of the Mean Absolute Percentage Error (MAPE), Absolute Percentage Error (APE) and standard deviation (σ) and compared with the kNN and MICE imputation methods. Once the novel hybrid imputation method is established, the operational analysis step concludes the data preparation methodology. The data preparation methodology of the proposed framework is demonstrated in Figure 3.4.



Figure 3.4: Layout of the data preparation methodology of the proposed framework, demonstrating the main steps and process.

3.5.2 Data Collection

The data collection step encompasses the necessary actions to collect the data used in the methodology. For that purpose, a commercial Data Acquisition (DAQ) system installed on board a merchant navy vessel was used. The data collected during the data preparation methodology differ from the similar step of the critical equipment selection methodology. In detail and as seen in Figure 3.1, instead of collecting repair costs and MTBFs, ship system performance data were gathered. The collected data are used to test the performance of the developed novel hybrid imputation tool. Lastly, performance data are collected, as similar research efforts are tested on performance data from their respective systems (Martinez-Luengo, Shafiee, and Kolios 2019).

3.5.3 Preliminary Analysis

The preliminary analysis is the step that follows the data collection step. The preliminary analysis serves the essential task of preparing the collected data for the imputation process. The four processes of this step are the form handling, synchronisation, filtering and correlation examination of the collected data and variables.

As seen in Figure 3.4, the first process is the form handling of the data. Form handling is a simple starting point for the preliminary analysis, yet its importance is paramount. This process ensures that the collected data are in tabulated form, ready for the next step of the methodology. Form handling is a process that is often the starting point of data-driven research efforts that contain unstructured data, collected from various sources.

As soon as the collected data are in a clear and tabulated format, the data synchronisation process takes place. Data synchronisation is needed as the different variables from the collected data may not be recorded uniformly. This may occur for two main reasons: a) the data collection sensors may not have the same sampling rate, and b) not all data collection sensors begin recording at the same time. Consequently, the synchronisation of the data warrants the consistency and the harmonisation of the data over time. Having all the variables synchronised over time can be very useful, especially when using similarity-based imputation approaches, or when trying to correlate variables for FD and diagnostic purposes. For the synchronisation of a variable, a timestamp is selected, and then, the timestamps before and after the selected one are also used. The synchronisation of the data is based on linear interpolation and is performed according to Equation 3.9

$$y_2 = \frac{(x_2 - x_1)(y_3 - y_1)}{(x_3 - x_1)} + y_1 \tag{3.9}$$

where:

- x_1 : the time before the timestamp of the synchronisation
- x_2 : the timestamp of the synchronisation
- x_3 : the time after the timestamp of the synchronisation
- y_1 : the variable before the timestamp of the synchronisation
- y_2 : the variable at the timestamp of the synchronisation
- y_3 : the variable after the timestamp of the synchronisation

The next process of the preliminary analysis step is the data filtering. This process is used to identify the points that need to be imputed, by determining if a sensor reading is missing, or it has an illogical value. The assessment of whether a recorded value is logical, or not, depends on the engineering knowledge of the variable being measured. As discussed in Section 3.6.2, domain knowledge is used to identify points that do not conform to an expected behaviour. The working process of the data filtering phase is shown in Algorithm 1.

Algorithm 1 Data filtering working process

Require: the upper limit (UL) and lower $limit(LL)$ for each variable (v) in the
collected data
for v in variables do
for i in v do
if i is NaN then
Flag i as a missing reading
else if $i > UL$ or $i < LL$ then
Flag i as illogical
end if
end for
end for

Lastly, the final process of the preliminary analysis is the examination of the correlation of the variables. This is a common step in most data-driven research efforts, as it helps with the examination of the data. The identification of the correlation between the variables from the collected data helps with imparting FP knowledge during the imputation process. The correlation between the collected variables was examined by using the Pearson correlation coefficient, as seen in Han, Kamber, and Pei (2012), and cross-referencing the results with FP domain knowledge. The Pearson correlation coefficient ranges between -1 (perfect negative linear correlation) and 1 (perfect positive linear correlation) while 0 denotes no linear correlation.

3.5.4 Imputation Process

Following the completion of the four processes of the preliminary analysis step, the imputation of the missing data takes place. This step includes the implementation of the novel hybrid imputation method. Specifically, the novelty includes the comparison of the state-of-the-art MICE against the widely used kNN imputation algorithms and the combination of these two methods in a single, new, imputation method. The comparison of the hybrid, kNN and MICE imputation algorithms include their application to the points identified during the data filtering process of the preliminary analysis. Finally, their imputation performance is assessed by using the APE, MAPE and σ error metrics.

The kNN algorithm is a very popular ML tool with widespread applicability in imputation applications. The popularity of this tool is due to its easy learning curve and ability to produce accurate results (Zhang 2012; Huang et al. 2017; Zhang et al. 2018). As also presented in Section 3.6.2.2, kNN does not require an explicit training phase, which increases its overall efficiency. Lastly, kNN is very easy to integrate into hybrid models. Similarly, the MICE imputation algorithm is also a very useful tool in predicting values in multivariate datasets, albeit it is relatively new compared to the more traditional imputation approaches. Nonetheless, the MICE algorithm has the advantage of incorporating the benefits of MI with regression-based imputation. As also discussed in Section 3.6.2.2, MICE can easily be incorporated into hybrid models. The novel hybrid imputation method that is proposed combines the benefits of the kNN and MICE imputation algorithms while avoiding their respective shortcomings. In more detail, the benefits of the hybrid imputation method are listed below:

- 1. Provides realistic imputations due to the use of the kNN algorithm
- 2. Provides easy incorporation of FP knowledge due to the use of the kNN algorithm
- 3. Is based on the widely accepted kNN algorithm
- 4. Takes advantage of the non-artificial replication of values offered by MICE
- 5. Takes advantage of the flexible implementation of MICE

MICE is a flexible and state-of-the-art imputation algorithm that works by fitting a series of regression models in the data (Shah et al. 2014). As previously mentioned, MICE is used to assess the effectiveness of the hybrid imputation method. This approach is considered as an implementation of MI which uses linear regressions to help in the estimate of the missing values. For the rest of the section, assume two random variables Y and K with observed (Y_{obs} , K_{obs}) and missing points (Y_{miss} , K_{miss}). Also, assume a set of complete variables Z with Z_{obs} and Z_{miss} corresponding to the observed and missing points of Y and K. MICE uses a Bayesian approach to calculate the missing points by updating the prior distributions of the random variables. The steps for the application of MICE are the following:

1. Impute all the missing points of Y and K with the averages of the Y_{obs} and K_{obs} (Equation 3.11 and 3.10). In addition, n_{Yobs} and n_{Kobs} represent the

total number of observations for the Y and K variables respectively.

$$\hat{Y}_{miss,i} = \frac{\sum Y_{obs,i}}{n_{Yobs}} \tag{3.10}$$

$$\hat{K}_{miss,i} = \frac{\sum K_{obs,i}}{n_{Kobs}} \tag{3.11}$$

The initial estimates $(\hat{Y}_{miss,i}, \hat{K}_{miss,i})$ are placeholders and are only used to initiate the process.

- 2. Set the placeholders of one of the variables (e.g. $\hat{Y}_{miss,i}$) back to missing.
- 3. Fit a linear regression model (Equation 3.12) between the observed points of the target variable (e.g. Y_{obs}) and the appropriate independent variables (either all, or a subset of Z).

$$\hat{Y}_{miss,i} = \theta^T Z \tag{3.12}$$

In equation 3.12, Z is a column vector of the independent variables and θ is row vector of the regression parameters. The $\hat{Y}_{miss,i}$ parameter represents the imputation estimates produced by the regression model, to replace the original placeholders.

4. Find the row vector θ by minimising the mean squared error (Equation 3.13).

$$MSE = \frac{1}{n_{Yobs}} \sum_{i=1}^{n_{Yobs}} (Y_{obs,i} - \hat{Y}_{obs,i})^2$$
(3.13)

The row vector θ can be calculated based on two different approaches. If the dataset is large, then an optimasation approach can be used (e.g. gradient decent) to fit the regression model and find the row vector θ . However, due to the size of the dataset used, an algebraic method was employed to fit the regression model and find the row vector θ . Generally, if the dataset

is relatively small algebraic methods can be used, as they offer greater simplicity. As the size of the dataset increases, optimisation approaches are used as they offer a significant reduction in the required computational time.

- 5. Having found θ , use Equation 3.12 to impute the missing points of the Y variable (Y_{miss}) .
- 6. Repeat steps 2, 3, 4, and 5 for every variable with missing points in the dataset
- Reaching step 6 is one cycle. The entire process is repeated for a predetermined number of cycles (usually 10 repetitions is an empirically accepted number).

To summarise, MICE uses linear regression in an iterative manner. The process initiates by using mean (vertical) imputation. Every variable with missing values is used in a regression model to update the initial mean imputation.

Apart from MICE, kNN is also used individually to examine the effectiveness of the proposed novel hybrid imputation method. In kNN, k represents a userdefined number of instances (i.e. nearest neighbours) that are considered for the hot-deck imputation. This is a non-parametric and lazy algorithm as it does not take into account the distribution of the data in the examined vectors, and it has no explicit training phase, as presented in Zhang and Zhou (2007). As with any hot-deck approach, kNN is based on the similarity between features, which is assessed by the Minkowski distance (Equation 3.14).

$$D = \left(\sum_{i=1}^{n} |x_i - y_i|^p\right)^{1/p}$$
(3.14)

In equation 3.14 the p hyperparameter is set to 2, which transforms D to the Euclidean distance. The Euclidean distance is the most common and widely used

distance metric, with implementations in many different applications (Groenen and Jajuga 2001; Dikis and Lazakis 2019). Similarly, x_i and y_i represent the examined instances. In addition to the distance metric, a weight is also assigned to each possible donor based on its distance. By doing so, closer neighbouring points (i.e. similar and most recent operating conditions) have a more considerable influence over the instance to be predicted. This is a vital feature as it allows taking into account the actual operation of the system under examination. The number for the k hyperparameter is not standardised. It depends on the field of application, and its selection lies with the researcher. In general, a small k will restrict the algorithm to a small region of the data, and as a result, it will produce results with low bias and high variance. A very small value for k (e.g. k = 1) can create models sensitive to outliers, noise and anomalous data, as the model is overfitted and not generalised enough for use in out-of-sample data. On the other hand, a high k (e.g. k = 30) can create overgeneralised models, as it averages more possible donors, generating results with low variance and an increased bias.

Lastly, the hybrid approach integrating kNN and MICE is applied and assessed against the previously discussed algorithms. The kNN component of the hybrid approach is based on FP knowledge. This approach initiates with the correlation analysis, where the systemic correlations between the collected variables are determined. In the hybrid approach, each vector in the dataset is examined in turn. When an instance with a missing value is identified, the kNN imputation algorithm is used. However, kNN looks for possible donors only in correlated variables; as determined during the correlation analysis. This has the following two benefits:

- The imputation process is expedited as only certain vectors of the dataset are examined.
- The predictive power of the model is improved as only correlated variables

are used to predict missing points.

This process is repeated until no further changes occur to the data set. In this case, the suspension of the kNN algorithm signifies its inability to impute any more missing points, as points from the correlated vectors may be missing simultaneously. Therefore, the remaining missing points are predicted using the aforementioned MICE approach. The structure of the proposed novel hybrid methodology is demonstrated in Algorithm 2 (Cheliotis et al. 2019).

Algorithm 2 Hybrid novel imputation method using a combination of kNN and MICE

Require: filtered dataset x of dimension $m \times n$
$modify_flag \leftarrow 1$
while $modify_{-}flag == 1$ do
$modify_flag \leftarrow 0 \{ Check x was updated in prev. loop \}$
for $i = 1, 2 \dots n$ do
$temp_column \leftarrow i^{th}columnofx$
$corr_columns \leftarrow columns correlated with temp_column$
for $j = 1, 2 \dots n$ do
if j^{th} element of $temp_column$ does not exist then
if j^{th} element of $corr_columns$ exists then
j^{th} element of $temp_column \leftarrow kNN$ imputation
$modify_flag \leftarrow 1$
end if
i^{th} column of $x \leftarrow temp_column$
end if
end for
end for
end while
$x \leftarrow \text{MICE imputation}$
return x

For the comparison of the mentioned imputation tools the APE (Equation 3.15), MAPE (Equation 3.16) and σ (Equation 3.17) are used. The aim is to choose the imputation approach with minimum MAPE and σ . The MAPE is selected as it is a popular and easy to understand metric for the evaluation of the model's imputation performance (Byrne 2012). Moreover, MAPE has

widespread applicability for predictive models and is commonly used in imputation (Martinez-Luengo, Shafiee, and Kolios 2019; Cheliotis et al. 2019). The MAPE is expressed as the percentage difference between an actual and forecast value. The standard deviation of the error is used to evaluate the sparsity of the errors and to determine the possibility of introducing outliers within the predictions. In the following equations, x_i and \hat{x}_i represent the actual value of the variable and the predicted value, respectively.

$$APE = \mid \frac{x_i - \hat{x}_i}{x_i} \mid \tag{3.15}$$

$$MAPE = \frac{100}{n} \sum_{i=1}^{n} APE \tag{3.16}$$

$$\sigma = \sqrt{\frac{\sum (APE_i - MAPE)^2}{n}} \tag{3.17}$$

In summary, three different approaches are implemented, compared and assessed. The first one is MICE, which is applied to the entire dataset until no missing points were left. The second approach that is tried is the kNN. This approach is applied to the entire dataset, without taking into consideration any systemic correlations, until all the missing points in the dataset are imputed. Lastly, the hybrid approach is tested, which combines kNN with FP analysis and MICE. The kNN algorithm is deployed by taking into account systemic interdependencies between variables. Then, the MICE algorithm is used to impute any missing points that the kNN cannot predict. These three approaches are verified by testing them in the collected data. A sample of the data are removed and the approaches are evaluated in their ability to predict the removed values, simulating the imputation of missing data, as detailed in Section 5.3.2.

3.5.5 Operational Analysis

The data preparation methodology concludes with the operational analysis. This step includes the correction of the variables to account for ambient conditions and ensures that the data are prepared for the following FD and diagnostic tasks. To account for the ambient conditions, various sources are taken into consideration, including international standards and the manufacturers' recommendations. Accounting for ambient conditions is a common step in many applications, as it ensures that the affected variables are adjusted accordingly.

3.6 Fault Detection (FD)

After the preparation of the data, the FD methodology of the framework follows. The FD methodology includes the novel combination of data checking steps, with regression-based EB modelling and the use of EWMA control charts for FD in ship systems. Moreover, the types of the different predictor variables are investigated, as mentioned in Section 3.8.1. The FD methodology ensures that developing faults are captured in a timely manner. In more detail, this methodology examines the detection of faults that result from gradual degradation of components. Such faults can include fouling, corrosion and wear-and-tear of components. Faults that are caused by a sudden shock or breakage are not considered. By developing the proposed methodology, the ability to capture previously unseen anomalies based on an EB model is enhanced. Besides, there is also the advantage of examining how signals evolve in real-time, based on contributing factors and uncoupled from operating conditions, due to the use of an EB approach. Also, the use of the EWMA control charts allows for the accurate detection of developing faults. Similarly, the superior performance of EWMA control charts in filtering-out noise compared to traditional control charts (e.g. Hotelling's T2 statistic), as seen in Cheliotis, Lazakis, and Theotokatos (2020),

is also beneficial. For clarification, noise is considered as systematic unwanted disturbances to the signal occurring from the system's interference with the environment. This is contrary to faults resulting from degradation which have measurable and observable trends (Guo et al. 2018).

3.6.1 Overview

As with the previous methodologies of the framework, the FD methodology initiates with its data collection step, as seen in Figure 3.5. As discussed in Section 5.4, the FD methodology is developed individually, and it uses different data from the previous methodologies. This step gathers data from multiple sources, including operating companies and data banks, to enable the development of this methodology. Once the data are collected, the data checking step follows, a step which prepares the collected data for subsequent FD steps. The data checking steps includes the application of the DBSCAN algorithm for outlier detection. and data filtering for the isolation of non-operational data points. Then, the model development step follows, which is based on historic ProMon data and shop tests and yields the developed EB model. During this step, the feasibility of different regression techniques and predictor variables is assessed in terms of their suitability for an EB model. Once this model is established, it is used for a comparison between the recorded and the expected values of a target variable, generating the residuals. In detail, once the predictor variables from the incoming data are checked, they are used as input to the EB model to produce the expected values of the target variables. These values are compared with the target variable from the incoming data, as demonstrated by the dotted lines of Figure 3.5. Thereupon, the obtained residuals are assessed in an EWMA control chart for FD.



Figure 3.5: Layout of the fault detection methodology of the proposed framework presenting the main steps and process.

3.6.2 Data Collection

The data collection is the first of the FD methodology. The output of the data collection step is the creation of a database with historical information, which is used for both model development and methodology verification purposes. As also seen in Figure 3.5, data collection includes the gathering of three different types of information. Initially, the shop tests of the ship's main engine are collected. The shop tests are machinery tests and represent a form of benchmark

and commissioning test. Even though shop tests are widely used in the maritime industry, their role in condition monitoring and PdM is limited. This is caused, as the actual operating conditions of the ship change throughout its life and differ from the conditions during the shop tests. Nonetheless, shop tests can be used to obtain initial estimates on limits of operation. Then, a DAQ system installed on board a merchant vessel is employed to gather historic ProMon data, The same system is also used to collect incoming ProMon data which are used to simulate different faults to evaluate the capabilities of the FD and diagnostic methodologies. Since the historic and incoming ProMon data are obtained from the same DAQ system, both of these datasets have the same sampling characteristics. In addition, any required pre-processing is the same between the two datasets. Typically, DAQ signals for FD tasks of engineering systems include power output, rotational speed, injection and scavenging pressure and EG, cooling medium, and LO temperature. Due to the nature of the FD methodology and unlike the data preparation methodology, the selected data must meet specific criteria regarding their diagnostic powers. In detail, developing faults must be able to manifest through the behaviour of the collected variables. Lastly, the data frequency usually ranges from one sample per second to one sample per five minutes, depending on the application.

3.6.3 Data Checking

Data checking is the next step following data collection and is concerned with ensuring that the collected datasets reach their full knowledge-extracting potential. Data checking, including the removal of outliers, is a standard step in most data-driven research efforts (Martinez-Guerra and Luis Mata-Machuca 2013; Sari 2013; Sayed-Mouchaweh 2018). The output of the data checking step is the creation of a checked dataset, ready for model development. There are four processes that are included in this step. The first process is a data formatting step, where the units of the collected data are checked, and their form is altered if needed. By doing so, it is ensured that the data are tabulated and placed in a suitable format for the next steps of the methodology. This process is very similar to the preliminary analysis, discussed in Section 4.4.3.

In the next process, the DBSCAN algorithm is used to identify and remove outliers and transient states of operation, which are out of scope in the present work. As mentioned in Section 3.6.2, outliers are considered as sparse data points with significantly different values from the rest of the instances of the same variables. They are often caused by sensor errors and other instrumental faults and are not part of a fault indicative pattern. For instance, negative EG temperatures and power output above an engine's rated power are considered as outliers. Thus, outliers can be considered as data "anomalies", and if they are not removed, they can have a negative impact on the developed models.

DBSCAN algorithm is very effective in detecting outliers and does not rely on domain knowledge, which offers several advantages. It is a density-based spatial algorithm that works by examining each point in the dataset and identifying dense areas of points (clusters). The DBSCAN algorithm requires the use of the user-defined minP hyperparameter. The minP defines the minimum number of points that are required to form a cluster. The minP is simple to specify as it is a function of the dimensionality of the dataset. Larger values are preferable, with an exclusive global lower bound of 3 (Schubert et al. 2017). Lastly, the value minP should be close to the number of dimensions of the dataset (Chen and Li 2011; Ester et al. 1996). As suggested in the literature, the minP hyperparameter is selected by combining the above restrictions with domain knowledge (Thang and Kim 2011; Schubert et al. 2017). Also, the ϵ hyperparameter is required, which defines the maximum distance between points for them to be considered to be in the same cluster. If ϵ is too small, the majority of the data points will be clustered as noise, whereas if it is too big, all the data points will be in the same cluster. In general, smaller values are preferred. An approach for calculating the ϵ hyperparameter is by considering the rate of change of the distance of each point to the nearest neighbour (k-nearest neighbour graph), as shown in the relevant literature (Rahmah and Sitanggang 2016; Gaonkar and Sawant 2013). However, this approach is not usable when using space data (i.e. time-series with constant sampling rate) evenly. As a result, the value of ϵ is obtained after iterative attempts.

Given these hyperparameters, the data are categorised in three groups. Core points are considered as data points with more than minP points within a radius of ϵ . Border points are defined as data points with fewer than minP points within a radius of ϵ . The remaining points are considered as outliers or noise. Moreover, a point q is directly density-reachable from a point p, if p is a core point and q is within a radius ϵ from p. Assuming another point q_1 which is directly density-reachable from point q only, it is said that points p and q_1 are indirectly density-reachable (Thang and Kim 2011; Chen and Li 2011; Çelik, Dadaşer-Çelik, and Dokuz 2011). The working process of the DBSCAN algorithm, as used in this methodology is shown below:

- 1. Find all core points
- 2. Assign all points that are directly density-reachable and indirectly densityreachable in the same cluster
- 3. Mark any unassigned points as outliers.

Following the removal of the transients and outliers, the data filtering process takes place. Specifically, the data are filtered to retain the points that represent operational periods. Since the data collection took place over an extended period, some points could have been recorded when the ship and its main engine were not operational. The data filtering is performed by using a value-based approach. Therefore, this process removes non-operational points while retaining the rest. The last process of the data checking step is the correction of the data to reference conditions, according to the ISO 3046-1:2002 international standards and the original engine's manufacturer recommendations, as also discussed in Section 4.4.5.

3.6.4 Model Development

The model development step follows the data checking step and uses as input the checked data, as seen in Figure 3.6. The aim is the development of an EB model that can predict the ideal (expected) behaviour of a selected variable of a system based on appropriately selected inputs. EB models are often used in for FD tasks, as they can model the expected behaviour of a variable subject to changing operating conditions. As discussed in Section 3.6.3.1, EB models are also ideal in the absence of faulty labelled data, as they can detect developing faults by defining a range of normal operation. The output of this step is the developed EB model which is in turn used for the FD step. It should be noted that the model development step includes iterations for the identification of the optimal predictor variables. Throughout this step, the used data are divided for training, validation and testing, based on empirical rules and common practices. The training sample of the data is used to fit the different models. It is said amongst practitioners, that the models "see" and "learns" for the training data. The validation sample of the data is then used to tune the models' hyperparameters. The validation set is withheld from the models during the training phase, but it can still affect the models' performance, albeit in a more limited manner than training. Ultimately, the appropriate model is selected based on its performance on the validation set. Lastly, the test sample of the data is used to evaluate the overall performance of the selected model, after they are trained, and their hyperparameters are tuned. The test sample is also withheld from the models during both training and validation.



Figure 3.6: Working process of the model development step, demonstrating the different process, including the selection of the different predictor variables.

3.6.4.1 Training and Validation

The training and validation process is used to fit and fine-tune the different models and is structured around the use of historic ProMon data collected during the ship's operation. The aim is to use a training set to fit the different models, and a validation set to fine-tune and ultimately select the best performing model, prior to the evaluation of its generalisation capabilities in a test set. This process uses training and validation datasets, which are a portion of the historic ProMon data. Finally, it must be stressed, that the recorded ProMon data represent "healthy" ship operation, as established by the ship's operators.

During the EB model development, four regression models are generated including Ordinary Least Squares (OLS) single linear regression, multiple linear ridge regression, OLS single polynomial regression and Multiple polynomial ridge regression. As discussed in Section 3.6.3.1, regression-based EB models do not depend on extremely large training datasets and offer greater flexibility in imparting domain knowledge, as they are not black-box approaches. This is achieved by the selection of the inputs and by observing and influencing the weight each input has (i.e. the coefficients) on the output. These models are used to produce an estimated output for a selected target variable by relying on the use of appropriately selected inputs (predictor variables). The examined EB models are trained and validated, using the R^2 metric, for both the shop tests and the historical data. Linear and polynomial regression models are developed to examine the best type of fit, given the acquired data. There are several advantages for each type of model; however, the selection is application-specific and a function of the available data, as discussed by Müller and Guido (2015). Single-input OLS regression models, both linear and polynomial, are produced as a basis for comparison with the more accurate ridge regression models, as seen through the work of Lepore et al. (2017), Erto et al. (2015), Naik, Bisoi, and Dash (2018), and Assaf, Tsionas, and Tasiopoulos (2019). Lastly, the specific inputs (i.e. the used predictor variables) for the EB models are investigated separately.

In general, the developed linear regression models have a form as shown in Equation 3.18, where \hat{y} represents an estimate for the target variable, w_0 to w_p are the regression coefficients, b is the axis intercept and x_0 to x_p represent the p different predictors (inputs).

$$\hat{y} = w_0 x_0 + \dots w_p x_p + b = \sum_{i=0}^p w_i x_i + b$$
 (3.18)

On the other hand, the developed polynomial regression models have a form as

shown below. This equation represents the general form of k^{th} order polynomial based on two predictors (x_1, x_2) and including the interaction terms between the predictors (Bowerman, O'Connell, and Murphree 2015; Olive 2005).

$$\hat{y} = w_0 x_1 + w_1 x_2 + w_2 x_1 x_2 + \dots + w_p x_1^k + w_p x_2^k + b$$
(3.19)

During the training phase, sets of known predictors $(x_0 \text{ to } x_p)$ and target variables (y) are used as input in either the linear or polynomial regression models to obtain \hat{y} . Then, the objective functions in either Equation 3.20 (OLS regression) or Equation 3.21 (ridge regression) are minimised. As a result, the estimates for the coefficients (w) and intercept (b) are obtained.

When OLS regression is used, the coefficients and intercept are estimated by minimising the sum of the squared difference between the predicted and the actual values of the target variable (residuals). The minimisation of this objective function is enabled since both y and \hat{y} are available during the training phase.

$$OLS: \|\hat{y} - y\|_2^2 \tag{3.20}$$

When ridge regression is used, the coefficients and intercept are estimated by minimising an objective function similar to the OLS. In addition to the sum of the squared residuals, an additional term is included. The additional term is called L2 regularisation and limits the magnitude of the coefficients. L2 regularisation explicitly restricts the model to avoid overfitting. The limiting capability of the regularisation term is attributed by the user-specified hyperparameter α . This hyperparameter limits the influence of the predictors to the target, given that α is appropriately selected. When α is equal to zero, the objective function becomes OLS, and on the other hand, if α is very large, the model will underfit the data. During this research effort, k-fold cross-validation was used to estimate the optimal α value (Olive 2005; Bowerman, O'Connell, and Murphree 2015; Bishop 2006).

$$Ridge: \|\hat{y} - y\|_{2}^{2} + \alpha \|w\|_{2}^{2} \quad with \quad \alpha \in [0, \infty)$$
(3.21)

K-fold cross-validation is a process which iteratively trains and validates the examined models by using all the possible combinations of training and validating sets. The working principle of this process is demonstrated in Figure 3.7, which is a common example with k = 3 folds. In essence, the k-fold cross validation is



Figure 3.7: Training and validation process using k-fold cross validation, with k = 3 folds.

used to evaluate the performance of the trained models and select the best performing approach for testing. This process trains and validates as many models as there are different combinations of model hyperparameters. The different hyperparameters included in this work are parameters of the learning methods (e.g. α regularization term), and the different inputs used (e.g. predictor variables). The k-fold cross-validation partitions the data in k different folds. Each fold is set aside once, and the examined models are trained on the remaining k-1 folds. Then, the fold withheld from training is used, to obtain the validation score the

Chapter 3
trained models. This sequence is repeated until every fold is used for validation once. For each model, the mean validation score is calculated, and the model with the highest mean score is selected. Finally, the identified model is trained with all the training and validation data before its generalisation capabilities are assessed in the test set.

Moreover, the general working process that is followed as part of the development of the EB model is shown in Algorithm 3. Algorithm 3 requires as input the predictor (X) and target (Y) variables. Also, it requires the number of folds (k) for the k-fold cross-validation and the size of the test set. Lastly, the set of the considered values for the model's hyperparameters is given. Algorithm 3 represents the generalised process for the development of the supervised model, including the optimisation of the α hyperparameter.

Algorithm 3 Model development and hyperparameter optimisation

Require: X, Y, k and a list of hyperparameters h_i for <i>i</i> between 1 and <i>n</i>
$X_{polynomial} \leftarrow$ derive the polynomial features of X
Augment X with $X_{polynomial}$
$X_{TrainValidate}, X_{Test}, Y_{TrainValidate}, Y_{Test} \leftarrow$ Split and normalise X and Y based
on $TrainingSet$
$Best_score \leftarrow 0$
$Best_parameter \leftarrow 0$
for $i = 1 \dots n$ do
Model \leftarrow model with h_i hyperparameter
Scores $\leftarrow k$ -fold cross-val. scores using $X_{TrainValidate}, Y_{TrainValidate},$ Model
Score \leftarrow average of Scores for the i^{it} iteration
if Score $> Best_score$ then
$Best_score \leftarrow Score$
$Best_parameter \leftarrow h_i$
end if
end for
Model \leftarrow model with $Best_parameter$
return Model

3.6.4.2 Testing

As mentioned above, four different types of regression models are used, namely OLS single linear regression, multiple linear ridge regression, OLS single polynomial regression and multiple polynomial ridge regression. Also, during the training and validation, the value of α , the different predictor variables are assessed in the required cased. After the k-fold cross-validation, the mean validation score for each examined model is obtained. The validation performance is assessed using the R^2 score, and the model with the highest R^2 is selected for testing.

The model selected for testing is fully defined in terms of the regression type, the predictor variables and the value of α . That model is then retrained using the training and validation datasets, and its testing performance is evaluated using the R^2 score. Once the testing performance of the selected model is analysed the model is finalised (Figure 3.6).

3.6.5 Fault Detection

Following the selection of the ideal regression model and the identification of the optimum predictor variables, the FD step ensues. The output of the EB model is a prediction (for the EB) of a specifically selected variable of an engineering system. To facilitate fault detection, the aim is to monitor specific variables (y) and gauge any deviations from their expected value (\hat{y}) , as produced by the EB model. Moreover, any deviations are assessed as a function of the ship's operational profile. As shown in Figure 3.5, the fault detection process has two inputs. It uses incoming, previously unseen data and the EB estimate from the model for the monitored variable.

For each instance of the incoming database, the residuals (r) between the

expected value and the recorded value is calculated according to Equation 3.22.

$$r_k = \hat{y}_k - y_k \quad for \quad k = 1, \dots N$$
 (3.22)

Analysing the residuals is an effective method for detecting faults in engineering systems, as the comparison between the actual and the expected behaviour can uncover developing faults at an early stage. The residuals quantify the deviation of a variable from its expected value, given an operating profile (Harrou et al. 2015; Awad, AlHamaydeh, and Faris 2018; Holmes and Mergen 2000).

After the residuals are calculated, the EWMA control chart is constructed, which allows for the creation of visual models that enable the accurate detection of developing faults in their early stages. In Equation 3.23, z refers to the EWMA statistic, which is calculated for all of the k instances. For the particular case of z_0 , the mean value of the variable in the incoming data is used. The smoothing effect of the EWMA is attributed to the user-defined smoothing parameter, λ . The smoothing parameter is defined according to common practices. Lastly, the residual at each instance (r_k) is used.

$$z_{k} = \lambda r_{k} + (1 - \lambda) z_{k-1}$$

for $k = 1$ (3.23)
and $\lambda \in [0, 1)$

A crucial component of the EWMA fault detection is the Upper Control Limit (UCL) and Lower Control Limit (LCL). These two limits provide the basis for the detection of faults, as any point above the UCL or below the LCL signifies faults. These limits are calculated according to the following Equations 3.24 and 3.25.

$$UCL = \mu_0 + L\sigma \sqrt{\frac{\lambda}{2-\lambda} [1 - (1-\lambda)^{2i}]}$$
(3.24)

$$LCL = \mu_0 - L\sigma \sqrt{\frac{\lambda}{2-\lambda} [1 - (1-\lambda)^{2i}]}$$
(3.25)

In these equations, μ_0 is the mean value of the variable in the incoming data and σ is the standard deviation. Lastly, L represents the width of the control chart, and its value is assigned based on the application, discussed in more detail in Section 6.4. In essence, the UCL and LCL form the envelope of normal operations for the selected variable. As the choice of L affects this envelope, it must be appropriately selected so that it can correctly classify normal and faulty operating points. If the recorded data represent "healthy" operating points, the resulting UCL and LCL envelope must fully encase all the data points. On the other hand, if a known fault exists in the data, the ULC or LCL must be exceed at the point of the failure. If the resulting envelope does not exhibit this behaviour, the value of L must be altered. Consequently, assigning L, its value can be an iterative process.

3.7 Diagnostics

The diagnosis of faults is the concluding step of the methodology. The aim is to propagate the evidence of developing faults in a diagnostic network that allows the determination of the root cause of the detected fault. Consequently, any diagnostic efforts must utilise to some degree, a fault detection methodology. Moreover, the diagnostic methodology can be used to summarise the condition of the ship effectively. This part of the framework focuses on the creation of a knowledge-based diagnostic network which is integrated with the FD methodology described in the previous section. In more detail, the integration of the knowledge-based diagnostic network with the ML-driven EB-based FD model is novel. The use of the knowledge-based diagnostic offers several advantages compared to the physics-based and data-driven alternatives. As discussed in Section 3.6.4, knowledge-based diagnostics offer accurate performance without requiring lengthy set-up times or a large amount of data for training. Besides, they are modular and easier to expand in multiple engineering systems.

3.7.1 Overview

The diagnostics methodology commences with the collection of the appropriate data, as seen in Figure 3.8. During this part of the framework, operating manuals from the ME's manufacturer are collected together with failure statistics from reliability data banks. Also, ProMon data collected during the FD methodology are used in this part of the framework too. This is due to the lack of additional required data, as explained further in Section 5.4. Once the required data are gathered, the fault mapping, which is of paramount importance to this diagnostic methodology, is developed. Fault mapping is a process that pairs certain faults with the variables they can affect. Moreover, the behaviour of the affected variables is also described. Then, the results from the FD step, mainly from the EWMA control chart, are aggregated. The objective of the aggregation is to classify the condition of the described system in its appropriate operational state. The FD and diagnostic steps are closely developed, as the trigger for the diagnostic tasks is a function of the output of the FD step. The next process is the set-up of the diagnostic network. In this work, a BN is employed due to the advantages detailed in Section 3.6.4. The BN integrates the results of the FD with the identification of the root cause of the detected faults. The structure of the network is specified by combining the results of the fault-mapping process with engineering knowledge of the examined system. At this stage, the failure statistics collected from data banks are used to quantify part of the BN. Once the structure of the BN is specified, the BN is used to combine the results from FD, with inputs from the data banks to pinpoint the root-causes of specific faults. Also, the final BN can be used to summarise the condition of a ship, quantifying the probabilities of different faults and summarising the condition of each cylinder.



Figure 3.8: Layout of the diagnostic methodology of the proposed framework showing the different steps and demonstrating the input from the FD methodology

3.7.2 Data Collection

The data collection step of the diagnostic methodology supplements the data collected in the previous parts of the framework. Due to limited access to additional sources for the required ProMon data, the data gathered in the FD methodology are also used for the diagnostics. These data are supplemented with the operating manuals of a ME of a merchant ship together with detailed failure statistics from data banks. The operating manuals are collected from ship operators and include details regarding the baseline operation, maintenance and troubleshooting of the ship's ME. To that end, these manuals include alarm limits for various variables at different operating points and include detailed accounts of functional dependencies between different systems. However, the alarm limits in the operating manuals are single points and represent hard limits aimed at preventing very severe equipment damages. The overall aim of the FD and diagnostic methodology is to detect ship degradation or the development of a fault before the alarm limits are reached. Data banks, on the other hand, offers failure statistic for a plethora of shipboard equipment. They can include the failure rates of all the possible failure modes for shipboard systems, sub-systems and components.

3.7.3 Diagnostic Set-Up

Diagnostic set-up is the next phase of the methodology and follows the data collection. There are three processes that form the diagnostic set-up step. These processes include the mapping of faults, the aggregation of the results from the FD and finally, the network set-up. The goal of this phase is the ability to use real-time information to produce accurate probabilities, of different faults, occurring in the selected system.

3.7.3.1 Fault Mapping

Fault mapping is a crucial task as it identifies the potential faults that can be diagnosed in a selected system, together with the variables required for their diagnosis. Therefore, this step examines the diagnostic potential of the ProMon data gathered during the FD. This step essentially justifies the variables that are monitored in the EWMA control chart. Alongside with the required variables, the acceptable range of operation and the behaviour of each variable are specified. Lastly, any additional tests required for the diagnosis of specific faults are specified in this phase. Fault mapping is based on domain knowledge and by taking into consideration the operating manuals of the selected systems, provided after personal communications with the ship operator.

3.7.3.2 FD Results Aggregation

The first examined process of the diagnostic set-up step is the aggregation of the results from the FD step. The goal of this process is to classify the condition of a system to its appropriate operation state (i.e. normal, degraded, failed) based on the behaviour of an appropriate variable. The behaviour of the variable is assessed in the FD step and mainly during the EWMA control chart.

As discussed in Section 4.5.5, the EWMA makes use of the L hyperparameter, which forms the envelope of normal operation. Any data points positioned outside of that envelope represent a fault. Moreover, Equations 3.26 and 3.27 are used to define the UCL_{deg} and LCL_{deg} which define the envelope of degraded operation. Similarly, any points between the failed and degraded envelopes represent degraded operations. The degraded limits use the L_d hypeparameter, such that $L > L_d$. Laslty, any other points represent normal operations.

$$UCL_{deg} = \mu_0 + L_d \sigma \sqrt{\frac{\lambda}{2-\lambda} [1 - (1-\lambda)^{2i}]}$$
(3.26)

$$LCL_{deg} = \mu_0 - L_d \sigma \sqrt{\frac{\lambda}{2 - \lambda} [1 - (1 - \lambda)^{2i}]}$$
(3.27)

3.7.3.3 Network Set-up

Once the pairing between monitored variables and corresponding faults is completed, the structure of the diagnostic knowledge-based BN is determined. As discussed in Section 3.6.4, knowledge-based diagnostic models have the advantage of mimicking the reasoning of a specialist, while avoiding the restrictions of black-box approaches. Similarly, diagnostic BNs have high modularity and are easier to set-up and interpret, compared to the alternative knowledge-based approaches. The identified faults are represented in the primary and secondary fault nodes of the network. Depending on the application, the states of these nodes can vary between Normal and Abnormal, Working and Failed, etc. The variables required for the monitoring are used in the observable nodes. These variables are selected so that the primary and secondary faults can manifest through the behaviour of these variables. Also depending on the application, the states of the observable nodes can vary between Normal, Degraded, and Failed. Any additional tests required for the investigation of a fault are inserted in the test nodes. The states of the tests nodes can vary between Pass and Fail, Normal and Abnormal, etc. Lastly, any other nodes concerned with the inner-workings of the diagnostic tool can be inserted in the control nodes section (Figure 3.9). The states of the control nodes can very between True and False, Pass and Fail, etc.



Figure 3.9: General diagnostic network structure

BNs represent a joint probability distribution of a set of random variables. They consist of a qualitative part and a quantitative part. The qualitative part is defined by a probabilistic Directed Acyclic Graphical (DAG) model, where each variable is depicted as a node. The qualitative part also includes directed links between the nodes to define a causal relationship. Similarly, the quantitative part is defined by the conditional probability distribution in the Conditional Probability Table (CPT) of each node (variable) (Ruggeri, Faltin, and Kenett 2007). The observable nodes of Figure 3.9 are the parent nodes of the control nodes. The control nodes can also be called the child nodes of the observable nodes. Moreover, leaf nodes are defined as nodes with parent nodes and no child nodes (e.g. test nodes), whereas root nodes have child nodes but no parent nodes (e.g. observable nodes) (Pearl 1985). BNs are based on Baye's theorem, with the goal of calculating the posterior conditional probability distribution of a fault given some observable evidence, as shown in Equation 3.28 (Langseth and Portinale 2007; Horný 2014; Cai, Huang, and Xie 2017b).

$$P(A|B) = \frac{P(B|A)P(A)}{P(B)} \iff$$

$$P(fault|evidence) = \frac{P(evidence|fault)P(fault)}{P(evidence)}$$
(3.28)

Assuming a set (U) of n random variables $U = (X_1, \ldots, X_j, \ldots, X_n)$, a BN with n-nodes can be constructed. Moreover, X_j represents the j^{th} variable. The BN for the n variables can be represented by Equation 3.29, where $pa(X_j)$ denotes all the parent nodes of X_j .

$$P(X_1, X_2, \dots X_j, \dots X_n) = \prod_{j=1}^n P(X_j | pa(X_j))$$
(3.29)

For example, the case of a simple network is considered in Figure 3.10.



Figure 3.10: sample of a basic BN

In that network, we assume that each variable (X_1, X_2, X_3) has only two states, True (t) and False (f). Therefore, Equation 3.29 takes the form of Equation 3.30, by using the chain rule of probabilities and a conditional independence assumption. The conditional independence assumption dictates that a child node (X_j) is statistically dependent only to its parents $(pa(X_j))$.

$$P(X_1, X_2, X_3) = P(X_1)P(X_2)P(X_3|X_1, X_2)$$
(3.30)

Therefore, the probability $P(X_3 = t)$ is represented in Equation 3.31, which is also referred to as the prior probability of X_3 .

$$P(X_3 = t) = P(X_3 = t, X_1, X_2) = P(X_1)P(X_2)P(X_3 = t|X_1, X_2)$$
(3.31)

In addition, assuming that X_3 is observed to be at its True state, then the probability of X_2 occurring $(P(X_2 = t | X_3 = t))$ can be found by using Equation 3.32.

$$P(X_{2} = t | X_{3} = t) = P(X_{2} = t, X_{1} | X_{3} = t) =$$

$$\frac{P(X_{3} = t | X_{2} = t, X_{1}) P(X_{2} = t)}{P(X_{3} = t)} P(X_{1})$$
(3.32)

Equation 3.32 is also referred to as the posterior probability, and the first part of the product is due to Equation 3.28, while the second term i due to the joint probability distribution (Pearl 1988).

For this study, two types of evidence were used, namely Hard Evidence (HE) and Virtual Evidence (VE). HE represent the traditional type of evidence used in BNs. They are used to dictate the value or state of a variable. For example, HE shows that a variable X_j has a value x_j , with mathematical certainty, or that the variable $X_1 = True$. Under the premise of this thesis, HE was used for strict diagnostic tasks, to obtain the probabilities of examined faults based on monitored variables. However, HE can introduce troublesome assumptions,

Chapter 3

especially when the value of a variable is very close to a state's decision boundary. In other words, when the value of X_j is such that the variable could be in either of its states. To counter this issue, and to extend the capabilities of the diagnostic network, VE was used. VE represent evidence with uncertainty and was used to also obtain the fault profiles of the examined faults (Bilmes 2004; Korb and Nicholson 2010; Mrad et al. 2012). For instance, $X_1 = 0.7 True$ is considered as VE and represents that X_1 is almost in its true state. For instance, VE can be used to describe the deteriorating state of a variable.

The fully defined diagnostic BN can be used for two main applications. Initially, it is used to carry-out "pure" diagnostic tasks, following the detection of a developed fault. Once the aggregated results from FD indicate the presence of a fault, HEs are used in the appropriate nodes (observable and tests nodes). As a result, the probabilities of different faults are investigated until the root cause of the detected fault is identified. In addition, the BN can be used to summarise the condition of the modelled system by characterising the condition of the observable nodes and assessing the probabilities of different faults developing. This process is also performed aggregating results from the FD and by using Equations 3.26 and 3.27. In this application, the BN is used proactively, as it does not relly on a detected fault, rather than the overall condition of the observable nodes.

3.8 Chapter Summary

This chapter presented the proposed framework, together with its novel methodological components. This includes the novel combination of FTA with k-means clustering for the identification of critical components, the novel hybrid imputation method based on MICE and kNN, the novel EB and EWMA-based FD model and the novel combination of an ML-driven FD with a BN-based diagnostic network. The framework targets the concept of maritime predictive maintenance in a complete and holistic way while generating novelty in all of its methodological components. The framework initiates with the critical components identification methodology, which is based on the used of FTA and the k-means clustering algorithm. Then, the data preparation methodology is presented, which aims at controlling the impact of missing values in the data. As a result, an imputation methodology is developed by combining the benefits of kNN and MICE imputation algorithms. Subsequently, the FD methodology is presented, which aims at developing models for the detection of developing faults in ship systems. The FD methodology includes the application of the DBSCAN clustering algorithm for the identification of outliers. Afterwards, an EB model is developed by examining the use of different regression models and assessing different predictor variables and training datasets. Once the EB model is developed, it is used to obtain the residuals, which are finally analysed in an EWMA control chart for FD tasks. Finally, the diagnostics methodology is developed. This methodology uses as input the evidence of developing faults, using the FD methodology, and tries to identify their root cause. This diagnostic task is performed in a BN, which is developed by combining the results of a fault mapping process with engineering knowledge. In conclusion, this chapter established the theoretical background of the proposed framework, which is required prior to the presentation of the different case studies.

Chapter 4

Case Studies

4.1 Chapter Overview

Following the presentation of the novel maritime predictive maintenance framework, this chapter aims at presenting the case studies used. The case studies give a description of the systems and data used for the application of each part of the framework and they present the necessary inputs. As a result, the case studies describe the technical details of the application of each methodological component.

In total, three different vessels were used for the case studies. The use of three different vessels for the different parts of the framework is justified based on three main reasons. Firstly, the different parts of the framework are developed independently. Secondly, each part of the framework has different data requirements and therefore, different sources of data and information are used. Lastly, it is not practical to find a single source that can fulfil the data requirements of all the parts of the framework.

Despite the three different case studies, the developed methodologies can be evaluated as a single framework in a unique case study. To that end, it should be clarified that the evaluation of the methodologies in a single case study does not negate their verification, as achieved by the individual case studies. Instead, the single case study can act supplementary to showcase further the developed methodologies. However, for this to be achieved, all the data requirements, discussed in the following sections, must be met by a single vessel. Initially, the maintenance schedule and repair costs would need to be collected. From that point, the ship systems analysis would take place to identify the main systems of the ship. This step would be performed with assistance from the vessel's operator. Then, the remaining steps of the critical equipment selection would be performed to identify the critical equipment. Subsequently, parameters from the critical equipment would be collected. Once they are identified and collected, the data preparation methodology would be deployed to identify and impute missing data. As soon as the missing values are treated, the FD and diagnostic methodologies would be applied. However, the fault mapping process and the diagnostic network would be a function of the available data and the identified critical equipment.

Initially, the case study applied for the selection of the critical equipment is given. Afterwards, the case study for the data preparation methodology is presented. Lastly, the case study for the FD and diagnostic methodologies is shown. Even though the FD and diagnostic methodologies are developed separately, they are applied in the same system of the same vessel, due to limitations of the required data. Despite this, both methodologies focus on different goals and they are evaluated independently.

4.2 Critical Equipment Selection

This section describes the case study in which the critical equipment selection methodology is applied. In more detail, this section gives general information on the type of ship used, the collection of the required data, the analysis of the ship systems and the resulting FT structure This methodology was applied in a case study of a Liquefied Natural Gas (LNG) carrier. The LNG carrier was selected for the case study due to the availability of the required data, obtained by the ship's operating company.

4.2.1 Data Collection

The collection of the required data is the first step of the critical equipment selection methodology. As seen in Figure 3.2, the methodology requires a maintenance schedule and a list of repair costs, both representing the same equipment and components of the LNG carrier. The maintenance schedule is used to obtain the numerical inputs for the FTA, in the form of MTBF for different components. On the other hand, the repair costs are used to identify the cluster of critical components.

The vessel studied in this part of the framework is an electric powered LNG carrier, referred to as vessel "A". Vessel "A" is a 162000 m^3 carrier with a continuous deck, sunken stern, bulbous bow, transom stern, single rudder and single screw propeller. The cargo area has a double deck, double hull and double bottom with cofferdams between each cargo tank. It also has four cargo tanks with the GTT MARK III membrane system directly supported on the inner hull. Vessel "A" has four Main Generating Engines (MGEs) in two different generating sets, providing propulsive and ship service power. Two main propulsion motors connected via a single output gearbox drive the propeller. The main features of vessel "A" are shown in the following table.

For the maintenance schedule, the PMS of the examined LNG carrier, supplied by the ship's operator, was used. The PMS details the different tasks that need to be performed, while identifying various parameters, requirements and constrains. In detail, each task from the collected PMS included the following: task identification information (number, code, creator), involved components, tasks description, type of action (inspection, testing, maintenance), calendar related

Characteristic	Value	Unit
L_{BP}	275	m
L_{OA}	289	m
B_{MLD}	45.6	m
D_{MLD}	33.2	m
T_{DESIGN}	11.2	m
$V_{SERVICE}$	19.5	knots
Capacity	163700	m^3
DWT	81000	tons
2 MGEs - 12V50DF	11700@514	kW, rpm
$2~\mathrm{MGEs}$ - $8\mathrm{L50DF}$	7800@514	kW, rpm

Table 4.1: Main features of vessel "A"

information (start date, finish date, task frequency, last performed date), responsibility related information (responsible department and position) and other identifiers. After careful examination only the following information were kept for each task: involved component, type of action, task description and details and frequency of task. The selection was based on the usefulness of the information and its applicability for the quantification of the Fault Tree. Table 4.2 is a sample of the used PMS, providing information for some of the components of the examined LNG carrier. The information obtained from the PMS is found in Appendix C.

Involved	Type of	Task Description and Details	Frequency	Unit
Compo-	Action			
nent				
Main En-	Maintenance	1)Clean the charge air cool-	4000	Hours
gine No.1		ers. 2)Perform the pressure test.		
High Tem-		3)Look for corrosion. 4)Measure		
perature		the pressure difference over the		
Air Cooler		charge air cooler before and after		
		cleaning.		
Main Gen-	Maintenance	1)Carry out external inspec-	3	Months
erator		tion (check for vibration, bear-		
Engine		ing temperature, performance of		
No.1 High		the pump and obvious defects).		
Temper-		2)Grease bearings as per maker's		
ature		intervals. 3)Check that holding-		
Circulat-		down and end-cover bolts are		
ing Pump		correctly tightened. 4)Check		
		starter/control box. 5)Check cou-		
		pling wear/backlash. 6)Clean		
		suction filter if necessary and pos-		
		sible.		

Table 4.2: Sample of the PMS of vessel "A"

Continued on next page

Involved	Type of	Task Description and Details	Frequency	Unit
Compo-	Action			
nent				
Main	Maintenance	1)Carry out external inspec-	3	Months
Engine		tion (check for vibration, bear-		
No.1 Lu-		ing temperature, performance of		
bricating		the pump and obvious defects).		
Oil Puri-		2)Grease bearings as per maker's		
fier Feed		intervals. 3)Check that holding-		
Pump		down and end-cover bolts are		
		correctly tightened. 4)Check		
		starter/control box. 5)Check cou-		
		pling wear/backlash. 6)Clean		
		suction filter if necessary and pos-		
		sible.		
Main Gen-	Overhauling	1)Inspect big end bearing,	12000	Hours
erator		one/bank 2)Dismantle the big		
Engine		end bearing 3)Inspect the mating		
No.1 Con-		surfaces, if defects are found		
necting		open all big end bearings and		
Rod and		renew bearing shells if necessary		
Big End		4)Check small end bearing and		
Bearing		piston pin, one/bank. If defects		
		are found, open all and renew if		
		needed.		

Table 4.2 - Continued from previous page

After this preliminary screening of the PMS, the frequency of the different tasks is formatted in different units. Since the PMS represents the frequency of the tasks either in hours, months or years, a common unit must be selected. Consequently, all the frequencies are converted to operational hours, to ensure uniformity and compatibility with the FTA (PTC Windchill 2019).

Apart from the PMS, cost information regarding various ship components are obtained. This information includes the repair costs of the components presented in the PMS. A sample table containing the used repair costs is shown in Table 4.3. The table with all the costs used is located in Appendix B.

Table 4.3: Table of repair costs of components shown in Table 4.2

Component	Repair Cost (US Dollars)
Main Engine No.1 High Temper-	15000
ature Air Cooler	
Main Engine No.1 High Temper-	5000
ature Circulating Pump	
Main Engine No.1 Lub Oil Puri-	2500
fier Feed Pump	
Main Engine No.1 Connecting	2500
Rod and Big End Bearing	

4.2.2 Ship System Analysis

The ship system analysis follows the data collection and its main aim is the examination of the case study vessel in terms of its systems, sub-systems and components. Moreover, the ship system analysis assists with the specification of the structure of the FTA.

After examining the specifications and engineering drawings of vessel "A", the main systems of the vessel are identified. Moreover, the systems were identified with supplementary meetings and interactions with the superintendent engineer of vessel "A". Figure 4.1 shows the main identified systems, as identified in five main groups. The essential machinery group includes the MGEs, the steering gear and the steam generation system. The fuel group includes the Fuel Oil (FO) transfer system, the FO purification system, the FO feed system for the MGEs and the gas fuel system. Also, the Lubricating Oil (LO) group covers the LO services system and the LO purification system. Moreover, the cooling group describes the central cooling system for the MGEs and the auxiliary cooling system. Finally, the various group encompasses the inter gas generation system, the LNG cargo equipment and the bilge, fire and ballast system.



Figure 4.1: Identified systems for the studied vessel

4.2.2.1 Essential Machinery

The essential machinery group includes the most essential machinery of vessel "A". The MGE system includes the two 12V50DF and the two 8L50DF engines. These engines are Dual Fuel (DF) and as such they can burn both FO and gas fuel. These engines are used to generate the required electric output to cover all

the electric loads of the vessel. The steering gear covers the equipment needed to steer the vessel along predefined routes. Lastly, the steam generation system includes the two auxiliary boilers that are responsible for generating the steam required for the various heaters and heat-exchangers.

4.2.2.2 Fuel

The fuel group includes the various oil and gas systems within the vessel. The FO transfer system includes various pumps which ensure that the different FO tanks are properly supplied. The FO purification system comprises of different pumps and purifiers that clean (purify) the FO onboard prior to its use in the various consumers. The MGE FO system consists of several pumps tasked with delivering purified FO to the MGEs. Lastly, the gas fuel system uses various different components (pumps, compressors, heaters) to deliver gas from the ship's cargo tanks to the DF MGEs for consumption.

4.2.2.3 Lubricating Oil (LO)

The LO group includes two main systems, the LO services and the LO purification systems. The former system includes different pumps tasked with delivering LO to various components in the ship, including the MGEs and the reduction gearboxes. The latter system includes pumps and purifiers which clean (purify) the LO prior to consumption.

4.2.2.4 Cooling

The cooling group includes the central cooling system for the MGEs and the auxiliary cooling system. The central MGE cooling system has Sea Water (SW) cooling pumps which transfer cooling SW to the central coolers. From the central coolers the cooled Fresh Water (FW) is circulated in the required components. The auxiliary cooling systems SW cooling pumps which transfer cooling water to

auxiliary coolers. From there, the cooled FW is circulated to the various locations through different circulating and booster pumps.

4.2.2.5 Various

The last identified group includes the remaining systems of the vessel. One of these systems is the Inert Gas Generation (IGG) system, which creates and distributes inert gas to void and cargo spaces. This is done for safety reasons to mitigate the risks of fire and explosions. Another system listed in this group is the cargo equipment of the ship. The cargo equipment includes various pumps, heaters and compressors. The main aim of this system is to handle the loading and unloading of the cargo, as well as maintain it during passages. The last identified system is the bilge, fire and ballast system, which has several different pumps (ballast, bilge and fire), firefighting equipment and a ballast water treatment facility.

4.2.3 Fault Tree Analysis

To perform a FTA, the structure of the FT first needs to be identified. This process is based on the results of the ship systems analysis, and is a necessary step in order to obtain the required reliability IMs.

As identified in the previous section, the five main machinery groups are the essential machinery, fuel system, LO system, cooling system and the various remaining systems. In the following figures, each gate examines the failure of the modelled system. For the purpose of this work, a failure is considered the loss of a system's function, similarly with the failures considered in Lazakis, Turan, and Aksu (2010a). All the gates, apart from the top event, do not state this explicitly for simplicity reasons. Moreover, transfer gates (denoted as blue triangles) are used to group together extensive parts of the FT. These gates are used for representation reasons only and are not related with any calculations. The gates at

the lower levels, and the events are identified by taking into account the results of the ship systems analysis and input from the operating company of vessel "A", creating the ideal FT structure. However, certain basic events may be omitted in case of failure data limitations.

The resulting FT structure is shown in Figure 4.2, where the failure of vessel "A" is modelled as an "OR" gate (top event). In this structure, the systems in essential machinery group are modelled together under the common "essential machinery" gate and form an input to the top event. Also, the remaining systems (i.e. fuel, LO, cooling and various) are modelled under the common "auxiliary" gate and form the final input to the top event. The systems in the essential machinery group are modelled under the same gate, as they represent systems that are crucial for the operation of the vessel. Similarly, the auxiliary gate represents a larger set of systems that collectively have a high criticality for the vessel. Finally, the top event is modelled as an "OR" gate, as a loss of function in either the essential machinery, or the auxiliary gate will cause the ship to loss its functionality.



Figure 4.2: FTA representation of the systems of vessel "A"

Figure 4.3 shows the structure of the FT which models the essential machinery group. The failure of the essential machinery group is represented with an "OR" gate, having as input the steering gear, generating engines and steam generation systems. The "OR" gate is selected as a failure in any of the three input systems will cause the failure of the essential machinery group and will cascade, triggering the "ship failure" gate.



Figure 4.3: FTA representation of the essential machinery group

Similarly, Figure 4.4 shows the FT structure for the fuel, LO, cooling and various systems represented as inputs to the These systems are connected through a "VOTING" gate, which requires two of the six inputs to occur for the "auxiliary" gate to fail. The use of the "VOTING" gate represents a set of systems that are less likely to cause the entire ship to fail. In more detail, the "VOTING" gates is used as the failure of the cargo equipment would not cause the ship to stop its operation if the ship was sailing without cargo.



Figure 4.4: FTA representation of the remaining machinery groups

Figure 4.5 represents the FT structure that models the steering gear system.

This section of the FT has as inputs the machinery components and the control components of the system. The failure of the steering gear system is modelled with an "OR" gate, as any failures in either the machinery, or control components can cause the system to cease operating.



Figure 4.5: FTA representation of steering gear system

Figure 4.6 represents the FT structure of one of the four MGEs. Each MGE has as inputs the assembly of the engine (cylinders, liners, etc.) and the supporting systems of the engine (FO, LO and cooling system). The failure of each MGE is modelled with an "OR" gate, as a failure in either the assembly or the supporting systems will cause the engine to stop operating. As can be inferred from Figure 4.3, the system of the generating engines is modelled with a "VOTING" gate. The four MGEs offer a higher degree of redundancy and consequently, the vessel can still operate, albeit with a reduced performance, with only one MGE.

Figure 4.7 shows the FT structure for the steam generation system. This system includes the fuel and feed-water systems together with the internal components as inputs. The steam generation system comprises of two boilers, resulting in enhanced operability. The steam generation system is modelled as an "OR" gate, as any failure in its sub-system will stop steam generation.

Figure 4.8 shows the FT structure for the IGG system, comprising of the blowers and the various other sub-systems. The IGG system is also modelled



Figure 4.6: FTA representation of one of the four MGEs



Figure 4.7: FTA representation of the steam generation system



with an "OR" gate, as it is sensitive to failures.

Figure 4.8: FTA representation of the IGG system

Also, Figure 4.9 shows the FT representation for the cargo equipment system. This system has as input the cargo spray pumps, high duty compressors and the other various components. As with the previous cases, the cargo equipment system is modelled with an "OR" gate, as it sensitive to failures.

Next, Figure 4.10 presents the FT representation of the bilge, fire and ballast system. This system comprises mainly of pumps, a water treatment system and various firefighting systems. Due to its nature, the bilge, fire and ballast system has high redundancy and each of the respective sub-systems has multiple redundant pumps which increase the system's operability. For instance, the duties of a failed bilge pump can be replaced by a ballast pump. Even though this functional relationship exists, modelling it in FTA is a challenge. This is due, as the representation of multiple functional dependencies between different systems is an inherent limitation of FTA (Vesely 2002). As a result the bilge, fire and ballast system is modelled with a "VOTING" gate, failing when either two of the bilge, fire and ballast systems fail.

Figure 4.11 shows the FT representation of the cooling system. As can be



Figure 4.9: FTA representation of the cargo equipment



Figure 4.10: FTA representation of the bilge, fire and ballast systems

seen, this system comprises of the auxiliary cooling and the central MGE cooling sub-systems, shown in Figure 4.12 and Figure 4.13 respectively. The cooling system is modelled with an "OR" gate, as a failure in either of the inputs will cascade.



Figure 4.11: FTA representation of the cooling system and the two sub-systems

Similarly, Figure 4.14 shows the LO system, as modelled in a FT. This system comprises of the LO purification and the LO services systems, as seen in Figure 4.15 and Figure 4.16 respectively. The LO system is modelled using an "OR"



Figure 4.12: FTA representation of the auxiliary cooling systems

gate, as it will fail if either of its inputs fail.

Lastly, Figure 4.17 shows the FT representation of the fuel system. This system has four inputs (sub-systems) including, the gas fuel, MGE FO, FO pu-



Figure 4.13: FTA representation of the central MGE cooling system

rification and the FO transfer systems. The fuel system is modelled with a "VOT-ING" gate requiring two of the four inputs to fail. This due to the fact that the vessel may be able to operate even if the gas fuel system fails, since the MGE



Figure 4.14: FTA representation of the LO systems

can operate with FO only (dual fuel). However, the use of this "VOTING" gate represents the best possible representation of the system, without increasing the model's complexity which is limited by the available computational power. For instance, the FO purification and FO transfer systems may fail and the system could still be operational for a while, due to the existing fuel in the service tanks. Nonetheless, this may not always be the case, as purified oil may be required when the service tanks are almost depleted. Therefore, the ability to model such cases is traded for a more manageable model.



Figure 4.15: FTA representation of the LO purification systems



Figure 4.16: FTA representation of the LO services system



Figure 4.17: FTA representation of the fuel systems


Figure 4.18: FTA representation of the gas fuel system



Figure 4.19: FTA representation of the MGE FO system



Figure 4.20: FTA representation of the FO purification systems





4.3 Data Preparation

This section describes the case study in which the data preparation methodology is applied. Consequently, the vessel and the data used for the application of the data preparation methodology are described. Since the operating company for the LNG carrier was only able to provide the PMS and repair costs of the vessel, alternative sources were sought. Instead of vessel "A", this methodology is applied in the case of a chemical tanker. Consequently, the operator of the chemical tanker was able to provide general information about the vessel, together with the required performance and process describing data.

As discussed in Section 4.1, the implementation of the data preparation methodology would be slightly altered if a single case study is used to evaluate all the methodologies. In detail, this methodology would be applied to parameters collected from the critical equipment, before they are used in the FD and diagnostic methodologies.

The vessel used in this case study is a diesel powered chemical tanker, referred to as vessel "B". Vessel "B" is a 38396 tons carrier with a continuous deck, bulbous bow, transom stern, single rudder and single screw propeller. This vessel is also double hulled and has five pure epoxy coated cargo tanks. Vessel "B" has a single ME for the generation of propulsive power and three diesel-powered generating sets for electric services. The main features of this vessel are summarised in Table 4.4.

4.3.1 Data Collection

Data collection is the first step of the data preparation methodology, as presented in Section 4.4. As shown in Figure 3.4, this methodology is structured on the use of performance data and on various operational limits from vessel "B". This case study is applied in the TC and ME system of vessel "B", a selection based

Characteristic	Value	Unit
L_{BP}	173	m
L_{OA}	183	m
B_{MLD}	27	m
D_{MLD}	16.8	m
T_{DESIGN}	11.6	m
$V_{SERVICE}$	10.0	knots
DWT	38396	tons
ME - 6S50MC	9611@127	kW, rpm

Table 4.4: Main features of vessel "B"

on the criticality and overall importance of the system (Theotokatos et al. 2018; Baldi, Theotokatos, and Andersson 2015), together with the input of the vessel's operator. A schematic of the studied system, together with the various available parameters is shown in Figure 4.22.



Figure 4.22: Diagram of a Main Engine (ME) system showing the physical interconnections between the measured parameters (the compressor is represented with C and the turbine with T)

The required performance data are collected from a DAQ system installed onboard vessel "B". The installed DAQ system records one sample every ten minutes from the ME power, ME speed, ME scavenging air pressure, TC EG inlet temperature, TC EG outlet temperature, TC LO inlet pressure, TC LO outlet temperature and the TC speed. These variables are recorded for nine days during the first two weeks of February of 2018, resulting in 1336 instances per variable. The descriptive statistics of these variables are shown in Figure 4.5.

Table 4.5: Descriptive statistics of the collected dataset used in the data preparation methodology

	ME	ME	ME	TC	TC	TC	TC	TC
	Power	Speed	Scav.	\mathbf{EG}	\mathbf{EG}	\mathbf{LO}	\mathbf{LO}	Speed
	(kW)	(rpm)	Air	Inlet	Out-	Inlet	Out-	(rpm)
			Press.	Temp.	\mathbf{let}	Press.	\mathbf{let}	
			(bar)	$(^{o}\mathbf{C})$	Temp.	(bar)	Temp.	
					$(^{o}\mathbf{C})$		$(^{o}\mathbf{C})$	
count	1336.0	1336.0	1336.0	1336.0	1336.0	1336.0	1336.00	1336.0
mean	4029.0	98.0	0.9	325.0	290.0	2.0	55.0	9494.0
std	1004.0	13.3	0.3	32.0	26.0	0.1	4.0	1995.0
\min	9.0	5.0	0.0	54.0	142.0	1.6	35.0	68.0
25%	3810.0	96.8	0.7	319.0	279.0	1.9	53.00	9088.0
50%	4264.0	102.0	0.9	334.0	293.0	2.1	55.0	9879.0
75%	4680.20	105.0	1.2	340.0	305.0	2.2	59.0	10897.0
max	5876.0	126.0	1.6	365.0	346.0	2.8	63.00	12258.0

Moreover, the aforementioned operational limits are aggregated from various sources including the shop tests (commissioning tests) of the ME and technical information from the Original Equipment Manufacturers (OEM) of the ship's equipment, as seen in Table 4.6. The operational limits define the range of plausible operating values for the collected variables on Table 4.5. The operational limits are used to scan and filter the data to determine the points for imputation, as described in Section 4.4.3.

Table 4.6:	Ship	system	parameter	limits	used	in	${\rm the}$	data	preparation	methodol-
ogy										

Parameter	Units	Lower	Upper	Source
		Limit	Limit	
ME Power	kW	0	10600	100% load from ME shop test
ME Speed	rpm	0	131	100% load from ME shop test
ME Scav. Air Press.	bar	0	3.14	100% load from ME shop test
TC EG Inlet Temp.	$^{o}\mathrm{C}$	35	650	From ambient temperature and
				130% of the TC OEM limit
TC EG Outlet Temp.	$^{o}\mathrm{C}$	35	650	From ambient temperature and
				130% of the TC OEM limit
TC LO Inlet Press.	bar	0	3.6	150% of the ME shop test
TC LO Outlet Temp.	$^{o}\mathrm{C}$	35	123	130% of the TC OEM limit
TC Speed	rpm	0	17600	110% of the TC OEM limit

4.4 Fault Detection (FD) and Diagnostics

This section describes the case study in which the FD and diagnostics methodologies are applied. Even though these two methodologies are independent, they are applied in the same vessel and systems, due to restricted accesses to additional required data. Since the operator of vessel "A" could not supply any ProMon data, and the operator of vessel "B" could not provide the needed parameters for meaningful FD and diagnostics, an additional ship operator was selected. As a result, the FD and diagnostic methodologies are applied in a case study of a bulk carrier.

The ship used is a 64000 tons bulk carrier, referred to as vessel "C". Vessel "C" is a diesel-powered and self-loading carrier with a continuous deck, bulbous bow, transom stern, single rudder and single screw propeller. This ship has five cargo holds and four cranes for the loading and unloading of its cargo. Lastly, vessel "C" has a double-bottom, a single ME for the generation of propulsive power and three diesel-powered generating sets for electric services. The main features of this vessel are presented in Table 4.7.

Characteristic	Value	Unit
L_{BP}	194.5	m
L_{OA}	199.9	m
B_{MLD}	32.26	m
D_{MLD}	18.5	m
T_{DESIGN}	11.3	m
$V_{SERVICE}$	13.2	knots
DWT	64000	tons
ME - 5S60ME	8050@89	kW, rpm

Table 4.7: Main features of vessel "C"

Both of the methodologies are applied in the ME and TC system of vessel "C". Similarly to Section 5.3.1, the selection of this system was based on its overall importance, criticality and the input the ship's operator. The schematic of the studied system, depicting the various components and measurable variables is shown in Figure 4.23.



Figure 4.23: Diagram of a Main Engine (ME) system showing the physical interconnections between the measured parameters (the compressor is represented with C and the turbine with T)

There are numerous monitored parameters in the selected system, including pressure and temperature drop across the Air Filter (AF) and Air Cooler (AC), speed of the TC, scavenging air temperature and pressure for the ME, power and rotational speed of the ME and EG temperature from each cylinder of the ME, as seen in Table 4.8. From these parameters, the EG temperature of each cylinder of the ME is monitored and used for FD. This selection was based on the variables importance in performance and process monitoring. In detail, monitoring the ME cylinder EG temperature can help a) control the ME's emissions, b) understand the cylinders' combustion performance c) identify underlying and developing faults. More specifically, faults in the AC, TC and gas passages of the ME can manifest through the ME EG temperature, as further described in Chapter 6 (Woodyard 2009; MAN B&W 20017). Apart from the ME cylinder EG temperature, the ME speed, power, scavenging air temperature and pressure are used as predictor variables and also for the identification of the root-cause of detected faults, as further described in Chapter 6. Also, the pressure and temperature drop across the AF and AC and the speed of the TC are used for the identification of the root-cause of detected faults.

4.4.1 Data Collection

The application of the FD and diagnostic methodologies in this case study requires the collection of several different types of data. As shown in Figure 3.5 and Figure 3.8 these methodologies require the use of historic and incoming ProMon data from vessel "C", information from the vessel's ME shop tests, the ship's ME operating manual and information from data banks.

4.4.1.1 Historic Process Monitoring (ProMon) Data

Historic ProMon data from the vessel's system are used in both methodologies. In more detail, they are employed to train, validate and test the developed EB model used for FD. Moreover, since the same FD model triggers the diagnostic methodology (i.e. the diagnostic tasks initiate after the detection of a fault), the historic ProMon data influence the diagnostic methodology. These data are collected from the beginning April of 2017 until the end of June of 2017, with a

Table 4.8:	Information	for the	e variables	used in	the FD	and	diagnostics	method-
ologies								

Variable Name	Description	Units	System	Role
ME CYL 1 EGT	ME cylinder 1	$^{o}\mathrm{C}$	Main Engine (ME)	Target variable
	EG temperature			
ME CYL 2 EGT	ME cylinder 2	$^{o}\mathrm{C}$	Main Engine (ME)	Target variable
	EG temperature			
ME CYL $3 EGT$	ME cylinder 3	$^{o}\mathrm{C}$	Main Engine (ME)	Target variable
	EG temperature			
ME CYL 4 EGT	ME cylinder 4	$^{o}\mathrm{C}$	Main Engine (ME)	Target variable
	EG temperature			
ME CYL $5 EGT$	ME cylinder 5	$^{o}\mathrm{C}$	Main Engine (ME)	Target variable
	EG temperature			
SHAFT_PWR	ME shaft power	kW	Main Engine (ME)	Diagnostic and Predictor
ME_RPM_TM	ME shaft speed	rpm	Main Engine (ME)	Diagnostic and Predictor
SCAV_AIR_PRESS	ME scavenging	bar	Main Engine (ME)	Diagnostic and Predictor
	air pressure			
SCAV_AIR_TEMP	ME scavenging	bar	Main Engine (ME)	Diagnostic and Predictor
	air temperature			
ΔPC	Scavenging air	mmWC	Air Cooler (AC)	Diagnostic Test
	pressure drop			
	across air cooler			
ΔPT	Temperature	$^{o}\mathrm{C}$	Air Cooler (AC)	Diagnostic Test
	difference be-			
	tween air cooler			
	air outlet and			
	air cooler water			
	inlet			
TCS	Turbocharger	rpm	Turbocharger (TC)	Diagnostic Test
	speed			
$\Delta \mathrm{PF}$	Pressure drop	mmWC	Air Filter (AF)	Diagnostic Test
	across tur-			
	bocharger air			
	filter			

5-minutes sampling rate and their descriptive statistics are shown in Table 4.9. Lastly, these data are randomly split into 80% for training and validation and 20% for testing, according to common practices and empirical knowledge.

4.4.1.2 Incoming Process Monitoring (ProMon) Data

Once the EB-based FD model is established through the use of the historic ProMon data, incoming ProMon data are used to simulate different faults to evaluate the capabilities of the FD and diagnostic methodologies. As mention in

	SCAV	SCAV	SHAFT	ME	ME	ME	ME	ME	ME
	AIR	AIR	\mathbf{PWR}	\mathbf{RPM}	CYL 1	CYL 2	CYL 3	CYL 4	CYL 5
	TEMP	PRESS	(kW)	\mathbf{TM}	\mathbf{EGT}	\mathbf{EGT}	\mathbf{EGT}	\mathbf{EGT}	\mathbf{EGT}
	$(^{o}\mathbf{C})$	(bar)		(rpm)	$(^{o}\mathbf{C})$	$(^{o}\mathbf{C})$	$(^{o}\mathbf{C})$	$(^{o}\mathbf{C})$	$(^{o}\mathbf{C})$
count	26205	26205	26205	26205	26205	26205	26205	26205	26205
mean	44.9	0.8	2750.0	279.0	212.0	197.0	211.0	210.0	220.0
\mathbf{std}	3.1	0.7	3066.0	2783.0	112.0	100.0	109.0	108.0	114.0
min	0	0	0	0	0	0	0	0	0
25%	41.5	0	0	0	51.8	52.2	52.7	54.1	53.3
50%	43.7	0.7	2710	59.8	285.0	261.0	280.0	280.0	294.0
75%	46.0	1.5	4624	74.9	296.0	271.0	295.0	292.0	304.0
max	55.3	2.6	32768	65509	364.0	336.0	357.0	354.0	360.0

Table 4.9: Descriptive statistics of the historic ProMon data used in the FD and diagnostic methodologies

the previous section, the historic data were supplied first. Once they were analysed and used for model development, additional data were requested from the operating company. However, only data from before April 2017 were available, and to avoid unnecessary repetitions of the work, they were used for the faults simulation. Therefore, before the newly supplied, incoming data, were used for the simulated faults different checks had to be performed. This was necessary due to the unconventional chronological order of the data sets. To avoid issues, the distributions of the two datasets were checked to ensure their similarity. As a result, it was observed that both datasets have similar distributions, validating their order of use. The accurate results from the verification of the methodologies also verify that the chronological order did not have a negative impact.

The use of simulated fault is necessary due to the fault-free nature of the used data, which is a very common problem in applications from merchant vessels. The maritime industry can be reluctant in sharing performance and condition datasets, even more so when they contain faulty data. Even though the same incoming ProMon data are used for the evaluation of the two methodologies, each methodology is assessed in different faults. These data are recorded from the beginning of January of 2017 until the end of March of 2017 and they are recorded with a 5-minutes sampling rate. The descriptive statistics of the incoming ProMon data are shown in Table 4.10

	SCAV	SCAV	SHAFT	ME	ME	ME	ME	ME	ME
	AIR	AIR	PWR	RPM	CYL 1	CYL 2	CYL 3	CYL 4	CYL 5
	TEMP	PRESS	(kW)	\mathbf{TM}	EGT	EGT	EGT	EGT	EGT
	$(^{o}\mathbf{C})$	(bar)		(rpm)	$(^{o}\mathbf{C})$	$(^{o}\mathbf{C})$	$(^{o}\mathbf{C})$	$(^{o}\mathbf{C})$	$(^{o}\mathbf{C})$
count	9960	9960	9960	9960	9960	9960	9960	9960	9960
\mathbf{mean}	40.8	0.7	2876.0	64.8	292.0	272.0	286.0	285.0	282.0
\mathbf{std}	1.0	0.1	140.0	0.7	6.3	7.6	4.7	5.9	17.7
\min	39.3	0.6	2474.0	58.2	256.0	235.0	255.0	247.0	247.0
25%	40.4	0.7	2794.0	64.5	288.0	267.0	283.0	282.0	269.0
50%	40.8	0.7	2852.0	64.7	292.0	274.0	286.0	286.0	277.0
75%	41.1	0.7	2928.0	64.9	296.0	277.0	288.0	289.0	299.0
max	49.9	1.3	4101.0	72.3	310.0	287.0	298.0	303.0	314.0

Table 4.10: Descriptive statistics of the incoming ProMon data used in FD and diagnostic methodologies

4.4.1.3 Main Engine (ME) Shop Tests

The ME shop tests represent another source of data used during the development (training, validation and testing) of the EB-based FD model. The shop tests are machinery tests and represent a form of benchmark and commissioning test, which is widely used in the maritime industry for setting the standards of normal operations for each ship. During these tests, the ME operates at different predefined loads and several parameters are recorded. Since these tests are performed at the manufacturer's facilities, the recorded values are intended to represent a baseline for normal operations. A sample of the shop tests of the ME of vessel "C" is shown in Table 4.11

ME Power(kW)	ME Speed (rpm)	EG Tempera- ture (°C)	ME Scaveng- ing Air Pres-	ME Scaveng- ing Air Tem-
			sure (bar)	$perature (^{o}C)$
2013	56.1	215.0	0.4	34
4025	70.6	246.0	1.2	30
6038	80.9	281.0	2.1	32
6843	84.3	308.0	2.5	34
8050	89	346.0	3.0	36
8855	91.9	364.0	3.3	38

Table 4.11: Shop test sample for the ME of vessel "C"

4.4.1.4 Data Banks

The last source of information that was used during this case study is the OREDA data bank. In the context of reliability and PdM, data banks are repositories that contain failure information that is difficult, or too detailed, to obtain from other sources (e.g.ship operators).

OREDA is a major data bank that originated from the offshore oil and gas sector after the collaboration of major industrial stakeholder. It contains a plethora of failure information for most shipboard systems, including mechanical and electrical components. Moreover, it is one of the most extensive and detailed data banks for machinery and is commonly employed in PdM and reliability related projects (Goble, Bukowski, and Loren 2016).

In OREDA, each of the examined systems (e.g. generating engines) is divided in sub-systems and components and the failure rate of each component is given. Moreover, OREDA specifies the different failure rates for the various failure modes of each subsystem and component. Failure rates of the different components were used to quantify the BN that was developed for the diagnostic methodology, as further discussed in Chapter 6.

4.4.2 Chapter Summary

This chapter presented the different case studies of the methodological components of the proposed framework. The main goal was to describe the different inputs required for the developed methodologies. In more detail, the critical equipment selection methodology is applied in the case study of an LNG carrier, as allowed by the access to the ship's maintenance schedule and repair costs. Then, a chemical tanker is used for the case study of the data preparation methodology. This is facilitated through the use of the available information from the OEM and from access to data collected from a DAQ system. Lastly, the case study of the FD and diagnostic methodologies is shown. These methodologies are applied in the same case study, but are developed and assessed independently. This required access to different ProMon data, the ME's shop tests and information from data banks.

Chapter 5

Case Studies Results

5.1 Chapter Overview

This chapter describes the results of the novel maritime PdM framework as applied in different case studies. Initially, the results from the critical equipment selection methodology, applied in the case study of an LNG carrier (vessel "A"), are presented. These results include the FTA and the resulting IMs, together with the application of k-means for the clustering analysis. Subsequently, the results from the data preparation methodology are presented, as applied in the case study of a chemical tanker (vessel "B"). The results include the preliminary analysis, the imputation process and the operational analysis. Finally, the results of the FD and diagnostics methodologies are presented, which originate from the bulk carrier case study (vessel "C"). The results from these two methodologies include data checking, model development, FD and diagnostic network set-up and use.

5.2 Critical Equipment Selection

The selection of critical equipment is the initiating methodology of the proposed novel predictive maintenance framework. This process of the framework is performed through the developed methodology presented in Section 4.4. The main aim of the critical equipment selection methodology is to enhance maritime predictive maintenance by establishing of a novel methodology for the identification systems with high functional and economic risk. To quantify the functional risks, the I^B and I^{CR} for each component are used. Similarly, the economic risks are gauged by using the repair and replacement costs for the modelled components As outlined in Figure 3.2, this methodology initiates with the data collection, ship system analysis and FT structure specification; steps which are applied in vessel "A', as earlier presented in Section 5.2. The IMs, together with the repair costs, are then clustered using the k-means algorithm, to identify the cluster with the critical equipment.

5.2.1 Reliability Importance Measures (IM)

After the structure of the FT is specified, as seen in Figures 4.2-4.21, the modelled "BASIC" events are quantified using MTBFs. Table 5.1 shows a sample of the MTBFs used for the bilge, fire and ballast system. The particular MTBFs used for each "BASIC" event are detailed in Appendix C.

Bilge, Fire and	Ballast
Component	MTBF (Hours)
Ballast Pumps	2190
BWTS	6048
Fire Pumps	2190
Bilge Pumps	2190
CO2 System	6132
Dry Powder System	6132

Table 5.1: Table of MTBFs used for the bilge, fire and ballast systems

Once all the "BASIC" events are quantified, the exact calculation method is specified, as described in Section 4.4.4.3, and the analysis is performed. In this work, the output of the FTA is the calculation of the I^B and I^{CR} IMs. The calculated IMs measure how the failure of each "BASIC" event influences the occurrence of the modelled top-event. As discussed in Section 4.4.4.4, the I^B measures the probability of a component being reliable for the occurrence of the top-event, without taking into account the component's reliability. Similarly, the I^{CR} evaluates the probability of the top-event occurring due to the failure of an examined component.

A sample of events with the highest I^B is shown in Table 5.2. The I^B for all of these components, show the probability of the component being reliable for the occurrence of the top-event, by only taking into account functional and systemic dependencies.

Table 5.2: Sample of the obtained I^B for vessel "A"

Component	I^B (Probability)
MGE 4 Cylinder Head	3.0307
MGE 1 FO Injector	2.0
FO Feed Pump 1	2.7061
MGE 3 LO Pump	2.02

As observed, the MGE Cylinder Head of vessel "A" has a very high I^B . Consequently, the MGE Cylinder Head has the biggest influence on the rate of change of the probability of the top-event. From a practical point of view, the failure of a cylinder head could cause either a slow-down or a shut-down of the MGE of the ship, crippling the operation of the vessel. Likewise, a failure in either a FO Injector or a FO Feed Pump could restrict the supply of FO to the engine, negatively influencing the ship's operation. The same adverse effects can be caused by a failure in an MGE LO pump, as the absence of LO can be disastrous for an engine. The I^B for all the modelled components can be found in Appendix C. Table 5.3 gives a sample of the events with the highest I^{CR} . The I^{CR} measures how critical each component is to the system, by also taking into consideration its reliability. Therefore, as it can be observed, the events in Table 5.3 are different from those in Table 5.2. This originates from the different formula each IM has, showcases their difference and further solidifies the need to examine both of them for the selection of the critical equipment. Therefore, the MGE FO Injection Pump and MGE FO Injector have the highest probability of causing a ship-wide failure. This is followed by the MGE LO Pump and the FO Feed Pump. The I^{CR} for all the modelled components can be found in Appendix C.

Table 5.3: Sample of the obtained I^{CR} for vessel "A"

Component	I^{CR} (Probability)
MGE 2 FO Injection Pump	0.18
MGE 3 FO Injector	0.15
MGE 1 LO Pump	0.13
FO Feed Pump	0.0874

By observing Tables 5.2 and 5.3, it is becoming apparent that even though both of the examined IMs assess the importance of components, the results can be different. For instance, according to the I^B , the most important is the MGE Cylinder Head. However, the most critical component, according to the I^{CR} , is the MGE FO Injection Pump. Also, this discrepancy between the ranked components is repeated. Therefore, additional analysis is required to select the most critical equipment of the modelled systems subjectively.

5.2.2 Clustering Analysis

The clustering analysis is the final step of the critical equipment selection methodology, developed as an integral part of the proposed novel predictive maintenance framework. During this step, the obtained IMs and repair costs are clustered together, using the k-means algorithm. Combining both of the obtained IMs with the repair costs allows for a more systematic selection of critical equipment.

Before the initiation of the clustering analysis, the k hyperparameter of the k-means algorithm must be specified. As discussed in Section 4.4.5, the k hyperparameter controls the ultimate number of the clusters; in this work is assumed that k = 3. As a result, the clustering algorithm will group components, based on their functional and economic risks, in three groups. In this work, it is assumed that the overall criticality of the equipment ranges. Therefore, the equipment can be divided into critical, partially critical and not critical components. The value of k is also a function of the shape of the data. In detail, the selected k value must correspond with distinct groups of the data, its value should be reduced.

Figure 5.1 shows the modelled components in a three-dimensional scatter plot, including the obtained IMs and repair costs detailed in Section 5.2.1 (data-space). The left plot of this Figure presents the modelled components in the examined data-space (IMs and repair costs). As also observed, the k value fits the shape of the data-space. In addition, the used axes are scaled, in accordance with standard practices, and to create an unbiased model. In this work, the common min-max scaling approach is used, as described in Necoara et al. (2008) , resulting in a dimensionless analysis. All of the axes range from 0 to 1, with 0 denoting the smallest value of the respective axis prior to the scaling and 1 indicating the largest values prior to the scaling. The right plot of this Figure shows the result of the k-means algorithm. In that plot, the different clusters are marked with green, blue and orange colours. Moreover, the centroids of the different clustered are denoted with an x-mark.



Figure 5.1: Three dimensional scatter plot showing the initial data-space (left) and the results of the k-mean algorithm (right).

As mentioned in Section 4.5.5, the critical equipment is defined as the components that belong in the cluster, which has the greatest distance from the beginning of the axes. That cluster contains components that have high I^B , I^{CR} and repair costs. As a result, the identified critical equipment has all of the following characteristics:

- They have the biggest influence on the rate of change of the probability of a ship failure, due to systemic dependencies.
- They are the most critical in the modelled system, by also considering the components' reliability and failure rate.
- They are the most costly to maintain after a failure, as they have the highest repair and replacement costs.

The cluster that contains the critical equipment is coloured in orange and is shown in Figure 5.2. On the other hand the green cluster in Figure 5.2 contains the least critical (i.e. safest) components.



Figure 5.2: Three dimensional scatter plot showing the least critical components (left) and the most critical components (right).

It is interesting to note that the green cluster includes the two most expensive components, including the BWTS and the CO_2 fire containment system. Especially in the case of the later, there are safety barriers and redundant systems in place (e.g. localised fire fighting systems), which reduce the overall risk of the system. On the other hand, it is observed that the orange cluster contains components with high IMs and relatively small costs.

In detail, the identified critical components are FO Feed Pump 1, FO Feed Pump 2, FO Feed Pump 3, FO Feed Pump 4, MGE 1 FO Injection Pump, MGE 2 FO Injection Pump, MGE 3 FO Injection Pump, MGE 4 FO Injection Pump, MGE 1 Injector, MGE 2 Injector, MGE 3 Injector, MGE 4 Injector, MGE 1 LO Pump, MGE 2 LO Pump, MGE 3 LO Pump and MGE 4 LO Pump. The I^B and I^{CR} of the identified critical equipment are shown in Figure 5.3 and Figure 5.4 respectively. Interestingly, even though the FO Feed Pumps and MGE FO Injectors have a relatively low I^{CR} they are included in the critical equipment. As previously discussed, the presented critical components belong in low order cut-sets (i.e. the smallest set of events that can lead to a failure). This justifies the absence of main engine components from the critical equipment, as these belong in cut-sets of higher order (i.e. more events required for a failure). Also, the failure statistics used to quantify the FT (provided by the ship operator) may have affected this outcome. This demonstrates the need for considering more than one IMs and taking into account both systemic interdependencies and individual failure statistics. Lastly, Figure 5.5 shows the Euclidean distance of



Figure 5.3: Bar chart showing the I^B of the identified critical equipment.

the critical equipment to the origin of the axes. This distance can consolidate and quantify the overall performance of the critical equipment in the used data-space. Figure 5.5 also shows for reference, denoted with a dotted line, the distance of the centroid of the critical equipment cluster.



Figure 5.4: Bar chart showing the I^{CR} of the identified critical equipment.



Figure 5.5: Distance of critical equipment to the origin of the axes

5.3 Data Preparation

Following the completion of the critical equipment selection methodology, the proposed predictive maintenance framework addresses the issue of data preparation. The data preparation part of the proposed framework is performed through the methodology presented in Section 4.5. The main aim of the data preparation methodology is to enhance maritime predictive maintenance by establishing a systematic and novel approach that ensures datasets reach their full knowledge-extracting potential. This is achieved by identifying and filtering-out missing and illogical values and subsequently using imputation to replace them. As outlined in Figure 3.4, this methodology initiates with the collection data from vessel "B", as described in Section 5.3. Following the data collection, the preliminary analysis follows, which includes the form handling, data synchronisation, data filtering and correlation examination processes. Then, the imputation step is performed, which includes the implementation of a novel imputation method and its comparison with other prominent algorithms. Lastly, an operational analysis is applied, which adjusts variables to account for ambient conditions.

5.3.1 Preliminary Analysis

The first process in the preliminary analysis is the form handling of the data, which ensures that data are in tabulated formats. Afterwards, the tabulated data are synchronised according to Equation 3.9 and filtered according to Algorithm 1. The final process of the preliminary analysis is the correlation examination of the variables. For that reason, the Pearson correlation coefficient is used. The Pearson correlation coefficient is selected due to its simplicity and universal applicability (Hastie, Tibshirani, and Friedman 2006). Figure 5.6 shows a heatmap of the correlation coefficient between the examined variables. Finally, the results of the Figure 5.6 are cross-referenced with the engineering knowledge of



the ME and TC systems.

Figure 5.6: Heat-map showing the Pearson correlation coefficient of the ME and TC variables.

From an engineering point of view, it is known that the TC LO inlet pressure and TC LO outlet temperature are amongst the uncorrelated variables. This is due to the fact that the TC LO system is independent and does not come in contact with areas of the ME or the TC where the ignition processes occurs. On the other hand, the TC speed and the ME power have the largest correlation. More specifically, the TC speed is influenced by many factors. It is correlated with the temperature drop of the exhaust gases in the TC. The TC speed is also correlated with the ME power and ME speed, which in turn influences the temperature of the exhaust gases. It should be noted that there are other variables that influence the TC speed, ME power and the temperature of the exhaust gases. For example, such variables are the combustion pressure and the back-pressure of the TC. The combustion pressure influences the power output and subsequently, the temperature of the exhaust gases. On the other hand, the back-pressure of the TC can affect (reduce) the TC speed, as the back-pressure can restrict the flow of the gases (Hountalas et al. 2014; Guan, Theotokatos, and Chen 2015). Even though these parameters are identified for their importance, they are not included in the selected dataset for the analysis. This is due to the fact that the DAQ used for the measuring of the parameters is not capable of recording them. Comparing the results from Figure 5.6 and the engineering knowledge of the ME and TC systems, the final correlation between variables is obtained. (Table 5.4). Table 5.4 presents in a concise manner the presence of correlation

Table 5.4: Resulting correlation based on the integration of the data-driven Pearson coefficient and the first-principle domain knowledge.

	Main	Engine pa	rameters					
	Power (kW)	Speed (rpm)	Scav. air press. (bar)	EG inlet temp. (°C)	EG outlet temp. (°C)	LO inlet press. (bar)	LO outlet temp. (°C)	Speed (rpm)
ME Power (kW)		1	1	1	X	×	×	1
ME Speed (rpm)	1		×	1	×	×	×	1
ME Scav. air press. (bar)	1	×		×	×	×	×	1
TC EG inlet temp. (°C)	1	1	X		1	×	X	X
TC EG outlet temp. (°C)	X	×	X	1		x	x	1
TC LO inlet press. (bar)	X	X	x	X	×		x	X
TC LO outlet temp. (°C)	X	×	X	X	×	x		X
TC Speed (rpm)	1	1	1	1	1	×	×	

between the collected variables (Cheliotis et al. 2019). This Table is also used in the application of the novel hybrid imputation approach, as described in the following sections.

5.3.2 Imputation Process

Following the completion of the preliminary analysis, the imputation process takes place. The filtered values from each of the collected variables from vessel "B", are replaced with predictions obtained by applying the MICE, and the kNN algorithms and the hybrid novel imputation approach. The imputation performance of the different algorithms is assessed with simulated missing values. More

specifically, random points from the used dataset (150 instances) are removed and the imputation algorithms predict the missing (removed) value. As a result, the comparison of the removed values with the predicted values evaluates the different algorithms (Cheliotis et al. 2019). The results from the discussed imputation approaches are presented in Figures 5.7-5.14. To evaluate the performance of the different imputation approaches and according to common practices of similar research, two graphs are generated for each variable. Initially, a scatter plot is provided, which presents the performance of each prediction. The x-axes of each scatter plot show the actual values of each prediction, whereas the y-axis shows the predicted values. Lastly, each scatter plot includes a dashed y = x line, which represents perfect accuracy and is provided to help determine the performance of each prediction. Also, a histogram is supplemented, which depicts the APE of each prediction. The histogram enumerates the APE of each prediction (distance from y = x) and is used to visualise the performance of each imputation approach. This representation of the results offers and concise and descriptive way of visualising the performance of the different imputation algorithms. These two graphs include the information that would be presented in a simple table format, however, they create a more engaging visualisation.

Figure 5.7 shows the imputed (predicted) values and APE for the TC LO inlet pressure. Observing the histogram, it is noted that all the imputation approaches produce equally good predictions. Also from the histogram, it is observed that the approaches produce predictions with APE ranging from 0-35%. This is a large range for the APE, indicating the probability of a model predicting inaccurate values. Through the scatter plot, it is observed that the kNN algorithm produces the results with the biggest standard deviation (sparsity), which can introduce outliers in the dataset. As aforementioned, all three approaches produce results with errors. This behaviour is attributed to the fact that TC LO inlet pressure does not have a substantial correlation with the other variables, as discussed in the previous section, and all of the imputation approaches are correlation-based. The errors produced are justified, as the TC LO inlet pressure is a variable with low correlation (Figure 5.6).



Figure 5.7: TC LO inlet pressure imputation performance comparing the MICE, kNN and hybrid methods.

Figure 5.8 shows the imputed values and APE for the TC LO outlet temperature. Observing the histogram, it is noted that the hybrid method has the best performance, as the majority of the predictions have less than 1% APE. The low APE of the hybrid method demonstrates its consistency in predicting very accurate results. Also from the histogram, it is observed that all the approaches produce predictions with APE ranging from 0-8%. This range of the APE indicates the relatively low probability of a model making inaccurate predictions. Through the scatter plot, it is observed that both kNN and MICE algorithms produce results with large sparsity, which can introduce outliers in the dataset. The hybrid approach performs the best, as it closely follows the y = x line.



Figure 5.8: TC LO outlet temperature imputation performance comparing the MICE, kNN and hybrid methods.

Figure 5.9 shows the imputed values and APE for the ME power. Observing the histogram, it is noted that the hybrid method has the best performance with the majority of the predictions having less than 1% APE. The low APE of the hybrid method demonstrates its consistency in predicting very accurate results. Also from the histogram, it is observed that all the approaches produce predictions with APE ranging from 0-12%. This range of the APE indicates the relatively low probability of a model making inaccurate predictions. Through the scatter plot, it is observed that the MICE algorithm produces results with large sparsity, which can introduce outliers in the dataset. It is observed that the hybrid approach performs the best, as it closely follows the y = x line. As the ME power is a highly correlated variable, the FP component of the hybrid method contributes to its overall positive performance.



Figure 5.9: ME Power imputation performance comparing the MICE, kNN and hybrid methods.

Figure 5.10 shows the imputed values and APE for the ME speed. Observing the histogram, it is noted that the hybrid method has the best performance with the majority of the predictions having less than 1% APE. The low APE of the hybrid method demonstrates its consistency in making very accurate predictions. Also from the histogram, it is observed that all the approaches produce predictions with APE ranging from 0-8%. This range of the APE indicates the low probability of a model making inaccurate predictions. Through the scatter plot, we observe that the MICE algorithm produces results with large sparsity, which can introduce outliers in the dataset. It is observed that all the tools produce more accurate predictions from 100 rpm and above. At lower speeds, many of the predictions are relatively inaccurate, with the MICE tool predicting possible outliers. As with the previous cases, the hybrid approach follows the y = x line the closest. Since the ME speed is also a highly correlated variable, the FP component of the hybrid approach has a positive effect.

Chapter 5



Figure 5.10: ME Speed imputation performance comparing the MICE, kNN and hybrid methods.

Figure 5.11 show the imputed values and APE for the ME scavenging air pressure. Observing the histogram, it is noted that the hybrid method has the best performance with the majority of the predictions having less than 1% APE. The low APE of the hybrid method demonstrates its consistency in making very accurate predictions. Also from the histogram, it is observed that all the approaches produce predictions with APE ranging from 0-35%. This is a large range for the APE, indicating the probability of a model making inaccurate predictions. Interestingly, the ME scavenging air pressure and TC LO inlet pressure have the largest range for APE. This is attributed to the uncorrelated nature of the TC LO inlet pressure and the poor performance of the kNN algorithm. Through the scatter plot, it is noted that the kNN algorithm produces results with large sparsity, which can introduce outliers in the dataset. In this variable, all of the imputation methods produce results very close to the actual values. It is also observed that both the MICE algorithm and the hybrid approach have comparable



performance and both follow the y = x line closely.

Figure 5.11: ME scavenging air pressure imputation performance comparing the ME, kNN and hybrid methods.

Figure 5.12 shows the imputed values and APE for the TC EG inlet temperature. Observing the histogram, it is noted that the hybrid method has the best performance with the majority of the predictions having less than 1% APE. The low APE of the hybrid method demonstrates its consistency in making very accurate predictions. Also from the histogram, it is observed that all the approaches produce predictions with APE ranging from 0-10%. This range of the APE indicates the relatively low probability of a model making inaccurate predictions. Through the scatter plot, it is observed that the Multiple Imputation by Chained Equations (MICE) algorithm produces results with large sparsity, which can introduce outliers in the dataset. The TC EG inlet temperature exhibits similar behaviour with the ME Speed. The predictions of all the tools are relatively inaccurate at lower temperatures. However, this behaviour is reverted at higher temperatures, with the hybrid method following the y = x line the closest.



Figure 5.12: TC EG inlet temperature imputation performance comparing the MICE, kNN and hybrid methods.

Figure 5.13 shows the imputed values and APE for the TC EG outlet temperature. Observing the histogram, it is noted that the hybrid method has the best performance with the majority of the predictions having less than 1% APE. The low APE of the hybrid method demonstrates its consistency in making very accurate predictions. Also from the histogram, it is observed that the approaches produce predictions with APE ranging from 0-10%. This range of the APE indicates the relatively low probability of a model making inaccurate predictions. Through the scatter plot, it is noted that the MICE algorithm produces results with large sparsity, which can introduce outliers in the dataset. In general, the hybrid method displays consistently good predictions in the temperature range following the y = x line the closest.



Figure 5.13: TC EG outlet temperature imputation performance comparing the MICE, kNN and hybrid method.

Finally, Figure 5.14 shows the imputed values and APE for the TC speed. Observing the histogram, it is noted that the hybrid method has the best performance with the majority of the predictions having less than 1% APE. The low APE of the hybrid method demonstrates its consistency in making very accurate predictions. Also from the histogram, it is observed that all the approaches produce predictions with APE ranging from 0-18%. This range of the APE indicates the moderate probability of a model making inaccurate predictions. Through the scatter plot, it is noted that the MICE algorithm produces results with large sparsity, which can introduce outliers in the dataset. Similarly with the previous cases, the hybrid method displays consistently good predictions in the speed range following the y = x line closely.



Figure 5.14: TC Speed imputation performance comparing the MICE, kNN and hybrid methods.

Summarising the above, Table 5.5 encapsulates the overall performance of the three imputation approaches for the examined variables. Table 5.5 shows the MAPE and mean σ for each approach and for each variable. Also, an overall MAPE and mean σ are shown to summarise the general performance of each approach. As it is observed, even though the MICE and kNN approaches perform relatively well, the hybrid method outperforms them. It has the lowest overall mean error of 2.21% and the smallest overall σ of 2.64%. The hybrid tool makes accurate predictions without running the risk of generating outliers. The worst performing tool is the kNN with an overall mean error of 5.55% and an overall σ of 8.9%. In addition, Figure 5.15 provides a graphical representation of the MAPE and mean σ of the different approaches across the examined variables. As visually confirmed, the hybrid approach tends to have the lowest error metrics (MAPE, mean σ). Observing the results, it becomes clear that in correlated variables (ME power, ME speed, TC speed, TC EG inlet temperature, TC EG

outlet temperature, ME scavenging air pressure, TC LO outlet temperature) the novel imputation method has superior performance. It is observed that the FP component of the hybrid model makes a positive influence on the prediction. By understanding the systemic interdependencies of the system under examination, the performance of the predictions is enhanced. Therefore, the integration of the knowledge of the system to any predictive effort is encouraged and should be preferred to purely data driven approaches.

Case		MAPE		Mean σ			
	kNN	MICE	Hybrid	kNN	MICE	Hybrid	
ME Power	3.16%	2.44%	2.29%	5.48%	4.06%	4.98%	
ME Speed	1.15%	1.02%	0.65%	1.12%	1.23%	1.05%	
ME Scav. Air Press.	17.15%	2.34%	1.92%	23.72%	3.63%	3.60%	
ME TC EG Inlet Temp.	1.63%	2.15%	0.92%	1.69%	1.96%	1.16%	
TC EG Outlet Temp.	2.25%	3.08%	1.19%	1.96%	2.60%	1.64%	
TC LO Inlet Press.	14.92%	8.46%	7.97%	32.11%	5.24%	4.96%	
TC LO Outlet Temp.	2.25%	2.33%	1.29%	3.65%	2.31%	1.73%	
TC Speed	1.96%	1.92%	1.42%	1.49%	2.61%	1.97%	
Average	5.55%	2.97%	2.21%	8.9%	2.96%	2.64%	

Table 5.5: Summary table of imputation approaches performance


Figure 5.15: Summary plot of imputation approaches performance

5.3.3 Operational Analysis

Following the imputation process step, the resulting dataset is adjusted to account for the influence of the environmental conditions. As also mentioned in Section 3.5.5, this is a common step in similar research efforts as it accounts for the influence ambient conditions have on the collected data. In general, accounting for ambient conditions improves the accuracy of the subsequent developed models (Sodré and Soares n.d.). For that reason, the TC speed, ME scavenging air pressure and the TC EG inlet temperature were corrected (MAN B&W 2014; Tsitsilonis and Theotokatos 2018) according to the manufactures guides and the ISO 3046-1:2002 standards (ISO 2008). The measured ME scavenging air pressure, P_{scav} , was adjusted to its corrected figure, $P_{scav,corr}$ according to Equation 5.1. In Equations 5.1, 5.2 and 5.3 K, F_1 and F_2 are correction constants, while T_{air} and T_{sea} are the ambient temperatures of the air and the sea respectively.

$$P_{scav,corr} = P_{scva} + (T_{air} - 25)F_1(K + P_{scav}) + (T_{sea} - 25)F_2(K + P_{scav})$$
(5.1)

The measured TC speed, N, was adjusted to its corrected value, N_{corr} according to equation 5.2.

$$N_{corr} = \frac{N}{\sqrt{\frac{(K+T_{air})}{(K+25)}}} \tag{5.2}$$

The measured TC EG inlet temperature, T_{egin} , was adjusted to its corrected value, $T_{eqin,corr}$ according to equation 5.3.

$$T_{egin,corr} = T_{egin} + (T_{air} - 25)F_1(K + T_{egin}) + (T_{sea} - 25)F_2(K + T_{egin})$$
(5.3)

5.4 Fault Detection (FD)

Following the completion of the data preparation methodology, the proposed predictive maintenance framework addresses the issue of FD. The FD part of the proposed framework is performed through the methodology presented in Section The FD methodology aims to enhance maritime predictive maintenance 4.6. by establishing a novel approach for the early detection of developing faults. During this work, the considered faults are the result of gradual degradation and wear-and-tear. In other words, sudden breakages and shock loads are not considered. As outlined in Figure 3.5, the FD methodology initiates with data collection, described in Section 5.4. Then, it proceeds with the development of a regression-based EB model, which uses as input the pre-processed historic ProMon data. The selected model is the optimal result from a systematic model evaluation process, which examines different regression types, model inputs and model hyperparameters. Once the generalisation performance of the selected model is evaluated, the fault detection steps proceeds, utilising the pre-processed incoming ProMon data. For FD, the residuals between the predicted values and the actual values of the target variable (ME EG temperature of each cylinder) are assessed in an EWMA control chart. Lastly, as discussed in Section 5.4.1, the performance of the FD process is evaluated using simulated faults. The simulated faults are divided into four different cases, with each case representing groups of different possible faults.

5.4.1 Data Checking

The first process of the data checking step is to ensure that the units of the data are in the correct form and that the dataset is in a tabulated form. Following that, the DBSCAN algorithm is deployed to remove transient states of operation and outliers from each variable. The application of the DBSCAN algorithm requires the specification of the ϵ and minP hyperparameters. The ϵ hyperparameter dictates the maximum distance between points for them to be considered in the same cluster. Also, the minP hyperparameter controls the number of neighbouring points required to form a cluster. As discussed in Section 4.6.3, the value of ϵ is determined after iterative attempts. Different values are iteratively used until the transient states are removed, and consequently, the outliers are filtered out. Finally, the selected value was $\epsilon = 0.25$. Considering the dataset used in the FD methodology has 9 dimensions, as seen in Table 4.9, and keeping in mind the restrictions suggested by Chen and Li (2011) the final value for minP was determined to by 9. As the sampling rate of the data is 1 recording per 5 minutes and minP = 9, samples from 45 minutes are required to form a cluster. This is realistic and reasonable time-frame for the operation of the ME when the ship in on voyage, as confirmed by the operators of vessel "B".

The last process of the data checking step is to filter-out any points collected when the ship was not operational. For that purpose a simple value-based filter was created, as seen in Equation 5.4.

$$Power_{ME} > 10kW \tag{5.4}$$

Figure 5.16 shows two plots depicting the initial state and the result of the data checking process for the ME shaft power and ME shaft speed variables respectively. In each of the two plots, the top graph represents the "checked data", whereas the bottom graph represents the "raw" data, as supplied by the DAQ system of vessel "C". As observed, the spikes and dips int the "raw" data graph are filtered out by the DBSCAN clustering algorithm, while the flat-lines are removed by Equation 5.4. The data checking results for all the variables of vessel "C" i are shown in Appendix D.



Figure 5.16: Data checking result for ME shaft power (top) and ME shaft speed (bottom)

5.4.2 Model Development

The purpose of model development is to identify the optimum EB model, which is then used to predict the ideal behaviour of a selected target variable. The target variable that is used in this part of the methodology is the ME EG temperature for each cylinder. This variable was selected due to its great importance in performance and process monitoring. Monitoring the ME cylinder EG temperature can help 1) control the ME's emissions, 2) understand the cylinders' combustion performance 2) identify underlying and developing faults. In more detail, faults in the air cooler, turbocharger and gas passages of the ME can manifest through the ME EG temperature.

During the model development, the available historical data are divided into a training and validation set and a testing set. The former is used to fit the different models, tune the different hyperparameters, compare and ultimately select the best performing model. The best model is selected by primarily assessing the validation score and taking into account the standard deviation (σ) of the prediction errors. Once the best performing model is selected, its generalisation capabilities are assessed in the training set.

In total, four different types of regression models are examined, including OLS single linear regression, multiple linear ridge regression, OLS single polynomial regression and multiple polynomial ridge regression, discussed in Section 4.6.4. Apart from the different types of models, different inputs (predictor variables) are considered for each model. For each of the ridge regression models, the α hyperparameter ranges from 0.1 to 0.6. Also, the examined polynomial models are assumed to be of 6th order. The value of α and the order of the polynomial models depends purely on each application, and there is no standard guide on their selection. Instead, different ranges of α and different values for the polynomial order can be examined. In case none of the models exhibits good behaviour with the selected α and order, these values are changed, and the analysis is repeated.

The various resulting models are detailed in Tables 5.6-5.8. These tables detail the different models with their IDs together with the inputs used in each case. The inputs refer to the predictor variables used in each model and can include the ME power, ME speed, scavenging air pressure and scavenging air temperature. Table 5.6 contains the different models based on multiple linear ridge regression and Table 5.7 contains the models based on multiple polynomial ridge regression. For instance, model M1 uses as predictors the ME power, ME speed, scavenging air pressure and scavenging air temperature and has the α hyperparameter ranging from 0.1 to 0.6 with 0.1 increments. Consequently, model M16 has the predictors of model M1, and the α hyperparameter has a value of 0.6. Lastly, Table 5.8 has the single input OLS models that are used as a benchmark.

Multiple Linear Ridge Regression			
Variables	Model ID		
Power, Pressure, Temperature, Speed	M1		
Power, Pressure, Temperature	M2		
Speed, Pressure, Temperature	M3		
Speed, Power, Temperature	M4		
Power, Pressure, Speed	M5		
Power, Pressure	M6		
Pressure, Temperature	M7		
Power, Temperature	M8		
Speed, Temperature	M9		
Speed, Power	M10		
Speed Pressure	M11		

Table 5.6: List of examined multiple linear ridge regression models.

Multiple Polynomial Ridge Regression			
Variables	Model ID		
Power, Pressure, Temperature Speed	N1		
Power, Pressure, Temperature	N2		
Speed, Pressure, Temperature	N3		
Speed, Power, Temperature	N4		
Power, Pressure, Speed	N5		
Power, Pressure	N6		
Pressure, Temperature	N7		
Power, Temperature	N8		
Speed, Temperature	N9		
Speed, Power	N10		
Speed Pressure	N11		

Table 5.7: List of examined multiple polynomial ridge regression models

Table 5.8: List of examined single regression models

Remaining Models			
\mathbf{Mode}	Model ID		
OLS Single Linear Regression	L		
OLS Single Polynomial Regression	Р		

5.4.2.1 Training and Validation

From the historic ProMon data, 80% of them are randomly selected for training and validation, according to empirical knowledge and common practices. The different models need a substantial amount of the collected data (e.g. 80%) in order to identify patterns and develop good generalisation capabilities. The random selection of the data controls the variance of the model, and it enhances its generalisation capabilities, as during the training, data from the entirety of the operational profile are used (Kirk 2017).

During this stage, k-fold cross-validation is used to train and validate the different models discussed above. For this work, the value of k = 7 was assumed. Similarly with the previous hyperparameters (e.g. α), the selection of k was based on empirical knowledge, as its value is application-specific. Consequently,

each model is trained and validated 7 times, each time using a different segment of the data for validation. The validation performance of each model is then assessed by averaging the 7 different validation scores resulting from the 7-fold cross-validation.

Figure 5.17 shows the validation performance of the different models in terms of their average R^2 score during the k-fold cross-validation. This Figure examines different regression models and with varying inputs and ranging α values. The upper bar chart examines the average score of each model (multiple linear ridge regression) shown in Table 5.6. To further evaluate the performance of the different models, the maximum score is shown as a dashed blue line. Similarly, the performance of the single OLS linear regression, using the ME power as an input, is shown as a solid black line. Models M11, M12 and M13 which use as inputs the ME power, speed, scavenging air temperature and pressure with $\alpha = 0.1$, $\alpha = 0.2$ and $\alpha = 0.3$ respectively have the best validation score of nearly 0.93. The performance of the OLS linear regression is lower than the other multiple linear ridge regression models. The lower bar chart examines the average score of each model (multiple polynomial ridge regression) shown in Table 5.7. The maximum score and the performance of the OLS single polynomial regression are also shown as a dashed and a solid line respectively. Models N53, N54, N55 and N56 which use as inputs ME power, speed, scavenging air pressure with $\alpha = 0.3$, $\alpha = 0.4, \alpha = 0.5$ and $\alpha = 0.6$ respectively have the best validation score of nearly 0.96. Interestingly, the performance of the OLS single polynomial regression is preferable to most of the multiple polynomial ridge regression models, as seen by the high R^2 score. Finally, it is observed that the polynomial models have a superior performance in terms of the mean validation score. As summarised in Table 5.9, the polynomial models have a higher mean validation score and a smaller range compared to the linear models.



Figure 5.17: R^2 validation performance of the linear (upper chart) and polynomial models (lower chart)

Table 5.9: Performance of linear and polynomial models in terms of \mathbb{R}^2

	Linear Models	Polynomial Models
Mean Validation Score	0.83	0.94
Validation Score Range	0.11	0.091

Figure 5.18 shows the validation performance of the different models in terms of the standard deviation (σ) of the average R^2 scores during the k-fold crossvalidation. This Figure examines the same models as Figure 5.17 and is supplementary for the evaluation of the different models. Similarly, the minimum σ of the average R^2 scores is shown as a solid line and the σ of the R^2 scores from the single linear and single polynomial models are shown as dashed lines. The standard deviation of the validation performance describes the consistency of each model in their predictions during the k-fold cross-validation. It is observed that the linear models have a superior performance in terms of the σ of the mean validation score. As summarised in Table 5.10, the linear models have a lower mean σ and a smaller range compared to the polynomial models.

Table 5.10: Performance of linear and polynomial models in terms of σ

	Linear Models	Polynomial Models
σ of Mean Validation Score	0.046	0.07
σ of Mean Validation Score Range	0.07	0.24



Figure 5.18: σ of the performance of the linear (upper chart) and polynomial models (lower chart)

As previously discussed, the model development aims to identify the model with the highest R^2 and a small σ . For that reason, the models with the highest R^2 and the models with the lowest σ are identified and compared. The identified models are presented in Table 5.11. As highlighted in the Table, model N54 has the best mean validation R^2 and a σ comparable with the remaining models and for these reasons is identified as the optimum choice. In summary model N54 uses multiple polynomial ridge regression has $\alpha = 0.4$ and uses as input the ME Power, pressure, and speed.

	Mean Validation R^2	σ of Mean Validation R^2
N54	0.96	0.03
M31	0.89	0.01
M32	0.89	0.01
M33	0.89	0.01
M34	0.89	0.01
$\mathbf{M35}$	0.89	0.01
M36	0.89	0.01
M91	0.88	0.01
M92	0.88	0.01
M93	0.88	0.01
M94	0.88	0.01
M95	0.88	0.01
$\mathbf{M96}$	0.88	0.01

Table 5.11: Performance of models with the highest R^2 and performance of models with lowest σ

Finally, Figure 5.19 shows the learning curves of model N54 having as target variable the EG temperature of the cylinders of the ME and the average EG temperature of all the cylinders of the ME. These curves show the training (red) and validation (black) scores for each case as a function of the number of folds in k-fold cross-validation. In effect, increasing the number of folds also increases the training data. Thus, the learning curves aim to evaluate if the model is either overfitting or underfitting the data. In other words, the learning curves are used to gauge the model's generalisation capabilities. In Figure 5.19, across all the graphs, as the number of fold increases the training performance reaches a plateau, indicating that the training performance can no longer improve by increasing the amount of training data. Similarly, the validation score increases, indicating that the generalisation capabilities of the model are satisfactory. In general, the convergence of the training and validation learning curves indicate the presence of a model with a good fit on the data. For example, the upper left chart of Figure 5.19 shows the learning curves for the model predicting the average EG temperature of all the cylinders of the ME. As seen, the training score reaches a plateau of around 0.977, and the validation score reaches a maximum value of nearly 0.968.



Figure 5.19: Learning curves of model N54 showing the training and validation scores as a function of increasing folds during the cross-validation process

5.4.2.2 Testing

Model N54 is identified as the optimal and fully defined choice for the prediction of the EG temperature of the cylinders of the ME. Following the completion of the training and validation process, model N54 is trained using the whole training set (no validation set is used). The trained model is then evaluated on the previously unseen test set. Figure 5.20 shows the training and testing scores of this process. As observed, the testing performance is satisfactory as the R^2 ranges from 0.93 to 0.966.



Figure 5.20: Final training and testing scores of the selected model

5.4.3 Fault Detection

This step of the methodology uses the incoming ProMon data, which are checked according to the process described in Section 4.6.3. The identified and evaluated model (N54) is used to obtain the expected (predicted) values for the ME EG temperature of each cylinder of vessel "C". Once these values are calculated, they are compared with the actual EG temperature from the incoming ProMon data, resulting in the residuals.

Once the calculation of the residuals is completed, the EWMA control chart is constructed, which requires the specification of the λ and L hyperparameters. The former is the smoothing parameter and is obtained according to common practices. In this work, it is assumed that $\lambda = 0.3$, according to Badodkar and Dwarakanath (2017). On the other hand, the L hyperparameter controls the width of the control chart (distance between UCL and LCL), and its value is assigned after iterations. As discussed in Section 5.4.1, the incoming data represent healthy operating conditions, as confirmed by the operator of vessel "C". Therefore, the value of L was selected so that the residuals on the control chart do not exceed the UCL and LCL. Figure 5.21 shows the residuals of the average EG temperature of cylinders of the ME of vessel "C", plotted in an EWMA control chart. The average EG temperature is used for simplicity reasons, as it summarises the behaviour of the individual cylinders. In this Figure, the obtained residuals are shown with grey, and the EWMA statistic for each residual is shown as blue. Lastly, the UCL, LCL and the Center Line (CL) are also shown. In Figure 5.21, L = 3 was used, since the resulting EWMA statistic for the residuals lie between UCL and LCL. It should be specified that L = 3 is the first value correctly classifying the EWMA residuals.



Figure 5.21: EWMA control chart for the selection of the L hyperparameter

The evaluation of the methodology transpires by using the developed EB model and analysing the residuals in an EWMA chart for fault detection. To examine the detection capabilities of the methodology, and by considering the fault-free nature of the available data, four different fault cases are examined through simulated data in the form of a sensitivity analysis (Law 2009; Saltelli 2004)

These four cases are presented in Table 5.12 and represent failure modes that can affect the target variable. In detail, according to domain knowledge and by considering the publications from Hountalas (2000) and Theotokatos and Tzelepis (2015), the examined cases represent specific failure modes in a ship, further explained in the remaining section. In addition, the limits presented in Table 5.12 are selected from the ME manufacturer guide (MAN B&W 20017) and represent the alarm limits set by the manufacturer. In each case, the appropriate variables in the incoming ProMon data are adjusted linearly, to reach and exceed the presented limits, simulating the faulty conditions.

Case ID	Variable	Alteration	Limit	Value
Case 1	ME scavenging air pressure	Increased	Upper	$3.3 \mathrm{bar}$
Case 2	ME scavenging air pressure	Decreased	Lower	0.4 bar
Case 3	ME cylinder EG temperature	Increased	Upper	420 $^o\mathrm{C}$
Case 4	ME cylinder EG temperature	Decreased	Lower	214 $^o\mathrm{C}$

Table 5.12: Verification cases description

These adjustments take place, across all cases, from 10/01/2017 to 11/01/2017 and is assumed that after this period, rectifying actions take place. Figures 5.22 -5.25 show the EWMA fault detection results for case 1 to case 4 respectively. All cases show the residuals of the average EG temperature across all the cylinders. This is due to simplicity reasons, as the examples of the examined faults affect the EG temperature of all the cylinders.

Figure 5.22 shows the EWMA smoothed residuals signal for case 1. As it can be observed, the points from the first few days have residuals with an error close to 0°C. However, from the 10/01/2017 the residuals begin to increase and reach more than 200°C. This surge is attributed to the increase of the ME scavenging air pressure to more than 3.20 bar. The residuals (as defined in Equation 3.22) increase, as the expected value of the target variable, increases with a higher rate. As it can be observed, the simulated fault is successfully detected as the EWMA exceeds the *UCL*. A typical example of a fault described in case 1 is the overloading of the ME caused by fouling of the ship's hull.



Figure 5.22: EWMA control chart of the average EG temperature for case 1

Figure 5.23 shows the EWMA smoothed residuals signal for case 2. Similarly with the previous case, the residuals from the first few days fluctuate around 0°C. However, from 10/01/2017 the residuals begin to drop and reach a value of more than -350° C. This decrease is attributed to the controlled drop of the ME scavenging air pressure to nearly 0.20bar. In this case, the residuals drop, as the expected value of the target variable declines with a sharper rate. As it can be observed, the simulated fault is successfully detected by the *LCL*. For instance, such behaviour can be attributed to fouling and corrosion in the TC of the ship, and fouling and corrosion in the nozzle ring of the TC.



Figure 5.23: EWMA control chart of the average EG temperature for case 2

Figure 5.24 shows the EWMA smoothed residuals signal for case 3. As it can be observed, the points from the first few days fluctuate around 0°C. However, from 10/01/2017 the residuals begin to decrease and reach a value of approximately 150°C. This drop is attributed to the simulated rise in the ME cylinder EG temperature to nearly 420°C. The residuals in case 3 decline, as the actual value of the EG temperature drops decoupled from the expected value. As it can be observed, the simulated fault is successfully detected, as the *LCL* is successfully exceeded. Typical examples of fault described in case 3 include the fouling of the main cooler of the ship and fouling in the AC of the ME (both air and water sides).



Figure 5.24: EWMA control chart of the average EG temperature for case 3

Figure 5.25 shows the EWMA smoothed residuals signal for case 4. As it can be observed, the points from the first few days fluctuate around 0°C. However, from 10/01/2017 the residuals begin to increase and exceed the value of approximately 200°C. This surge is attributed to the simulated drop in the ME cylinder EG temperature to nearly 214°C. Following the same underlying reasoning as case 3, the decoupled decrease of the actual values of the target variable increases the residuals. As it can be observed, the simulated fault is successfully detected by the *UCL*. A typical example of a fault described in case 4 includes the improperly maintained or improperly configured engine room conditions.



Figure 5.25: EWMA control chart of the average EG temperature for case 4

5.5 Diagnostics

Following the completion of the FD methodology, the proposed predictive maintenance framework addresses the issue of diagnostics. The diagnostics part of the proposed framework is performed through the methodology presented in Section 4.7. The diagnostics methodology aims to enhance maritime predictive maintenance by developing a novel approach to pinpoint the root cause of certain detected fault. More specifically, this methodology includes the novel integration of an ML-based EB model (multiple polynomial ridge regression) for FD with a knowledge-based BN for diagnostics. As discussed in Section 5.4, the diagnostic methodology uses the same data as the FD methodology, due to restrictions in obtaining additional required data. The FD and the diagnostics methodologies were developed sequentially, as any diagnostic efforts must initiate after the detection of a fault. As a result, the diagnostic methodology uses the identified EB model (model N54) as input. As outlined in Figure 3.8, the diagnostics methodology initiates with data collection, described in Section 5.4. The next step includes the fault mapping process, which allows the determination of the different possible causes of the faults identified with model N54. Then, model N54 is used to detect developing faults, and the results are subsequently aggregated. After the different fault causes are identified, the structure of the BN is determined, and the resulting network is used for diagnostic tasks.

5.5.1 Diagnostic Set-up

The diagnostic set-up is the primary step of this methodology and forms the link between the FD and the diagnostics methodologies. This step includes the fault mapping, the FD results aggregation process and the network set-up.

5.5.1.1 Fault Mapping

Fault mapping is a crucial task as it identifies the potential faults that can be diagnosed in a selected system, together with the variables required for their diagnosis. During this step, faults are paired-up with incoming ProMon variables and also with additional required tests. The monitored variables are selected so that any deviations indicate the development of a specific fault.

As previously discussed, the main monitored variables are the ME CYL 1-6 EGT (Table 4.8) The behaviour of these variables is accessed in terms of any abnormal increments in all of the cylinders simultaneously. Such behaviour indicates faults in the supporting systems of the ME, namely the AC and Air and Gas (AG) handling system. Table 5.13 shows the outcome of the fault mapping process, assuming a simultaneous increase in the EG temperature of all the cylinders.

Table 5.13: Mapped faults for increases in EG temperature in all cylinders, adapted from (MAN B&W 20017)

Primary Fault	Secondary	Diagnostic Parameters	Diagnostic Test
	Fault		
AC	AC air-side foul-	$\Delta PC, SCAV_AIR_PRESS$	Pressure drop test
	ing		
AC	AC water-side	$\Delta PT, SCAV_AIR_PRESS$	Temperature drop test
	fouling		
AG handling	AF fouling	$\Delta PF, SCAV_AIR_PRESS$	Pressure drop test
system			
AG handling	Corroded TC	$TCS, SCAV_AIR_PRESS$	Speed drop test
system	mechanical		
	components		
AG handling	TC Fouling	SHAFT_PWR, SCAV_AIR_PRESS	Scavenge air pressure drop test
system			

The faults in Table 5.13 are divided into primary and secondary. The primary faults refer to the system in which a fault is developing. The secondary faults refer to the specific components of a system in which the fault is developing. Moreover, each fault is given a specific diagnostic test to assist with its identification. The diagnostic tests are used once a simultaneous abnormal increase in the EG temperature is detected, in order to pinpoint the specific root cause of the detected fault. The fault-mapping process and the diagnostic tests were identified by taking into account the operating manuals of the ME (MAN B&W 20017; Woodyard 2009). The diagnostic tests are shown in Figures 5.26 - 5.30. Figure 5.26 shows the diagnostic test (chart) used to identify fouling in the air-side of the AC. The ΔPC is examined in terms of the $SCAV_AIR_PRESS$, and if its value is beyond the highlighted envelope, the test is failed. If this tests fails and all the cylinders have increased EG temperature fouling in the air-side of the AC is detected.



Figure 5.26: Diagnostic test for the detection of AC air-side fouling

Similarly, Figure 5.27 shows the diagnostic test identified from the ME manufacturer, used to identify fouling in the water-side of the AC. The ΔPT is examined in terms of the $SCAV_AIR_PRESS$, and if its value is beyond the highlighted envelope, the test is failed. If this tests fails and all the cylinders have increased EG temperature fouling in the water-side of the AC is detected.

Ship Maintenance



Figure 5.27: Diagnostic test the detection of AC water-side fouling

Likewise, Figure 5.28 shows the diagnostic test (chart) used to identify fouling in the AF of the ME. The ΔPF is examined in terms of the $SCAV_AIR_PRESS$, and if its value is beyond the highlighted envelope, the test is failed. If this tests fails and all the cylinders have increased EG temperature fouling in the AF of the ME is detected.



Figure 5.28: Diagnostic test for the detection of ME AF fouling

Figure 5.29 shows the diagnostic chart used to identify fouling in the mechanical components of the TC (turbine blades, nozzle ring). The TCS is examined in terms of the $SCAV_AIR_PRESS$, and if its value is beyond the highlighted envelope, the test is failed. If this test fails and all the cylinders have increased EG temperature fouling in the turbine blades, or nozzle ring of the TC is detected.



Figure 5.29: Diagnostic test for the detection of fouling in the mechanical components of the TC

Lastly, Figure 5.30 provides the diagnostic chart used to identify fouling in the turbine or compressor of the TC. The SCAV_AIR_PRESS is examined in terms of the ME power, and if its value is beyond the highlighted envelope, the test is failed. If this tests fails and all the cylinders have increased EG temperature fouling in either the compressor, or the turbine is detected.



Figure 5.30: Diagnostic test for the fouling in the turbine or compressor of the TC

Apart from the simultaneous increase in the EG temperature in all the cylinders, it is possible to observe increased EGs in only one cylinder. This behaviour alters the fault-mapping shown in Table 5.13 and instead the faults in Table 5.14 are investigated. An isolated increase in the EG temperature indicates the presence of faults within that specific cylinder, as demonstrated by the primary and secondary faults in Table 5.14.

Table 5.14: Mapped faults for increase in EG temperature of one cylinder, adapted from (MAN B&W 20017)

Primary	Secondary Fault	Diagnostic Param-	Diagnostic Test
Fault		eters	
Cylinder	Leaking Exhaust	$SHAFT_PWR, P_{max}$	Power drop and P_{max}
head	Gas Valve (EGV)		drop
Cylinder	Blocked fuel valve	$SHAFT_PWR, P_{max}$	Power drop and P_{max}
head	or fuel injector		drop
Combustion	Chamber blow-by	$P_{comp}, \qquad P_{max},$	P_{comp} drop,
chamber		SCAV_AIR_TEMP	P_{max} drop and
			$SCAV_AIR_TEMP$
			increase

For instance, to investigate whether an isolated EG temperature increase is caused by a leaking Exhaust Gas Valve (EGV), the power output and the maximum combustion pressure (P_{max}) of the suspected cylinder are examined. If the power output and P_{max} are reduced, a leak in the EGV of the cylinder is possible (Papagiannakis and Hountalas 2004). However, the same behaviour can be observed when the temperature rise is caused by a blocked fuel valve or fuel injector. It is often challenging to distinguish these two faults without visually inspecting the cylinders. However, the drops in the power output and P_{max} are more prominent in the former cases (Hountalas 2000). Lastly, the case of a blow-by, as defined by (Mobley, Higgins, and Wikoff 2008), is mapped. If the P_{max} and compression pressure P_{comp} are reduced, and the SCAV_AIR_TEMP is increased, the presence of a blow-by is detected (Woodyard 2009).

Based on the above, it is becoming apparent the faults diagnostics are heavily reliant on a variety of data. To successfully identify the faults mapped in Tables 5.13 and 5.14, both ProMon and PeMon data are required. Since the DAQ system installed onboard vessel "C" is not able to record PeMon data, only the faults shown in Tables 5.13 are pursued.

5.5.1.2 FD Results Aggregation

Once the fault-mapping process is completed, the results from the FD are aggregated to summarise the condition of each cylinder. During this step, the incoming ProMon data are used, together with the residuals obtained by using model N54 (described in Section 6.4.3). To evaluate the diagnostic capabilities of the developed methodology and since the incoming ProMon data represent fault-free operations, artificial faults are also used. The artificial fault is introduced in terms of increased residuals for each cylinder, which are caused by a gradual increase in the ME scavenging air pressure (predictor variable) until the alarm limit (3.30 bar) for the variable is exceeded (Table 5.12).

The FD results aggregation aims to introduce the necessary artificial faults, reflected on each cylinder. Then, the residuals of each cylinder are plotted in an EWMA control chart, covering in duration up until the end of the introduced fault. Once the EWMA control chart is plotted, the states of each cylinder are summarised (aggregated) in Failed, Degraded and Normal states. In detail, points situated above the UCL, or below the LCL, contribute to the Failed state. Point between the UCL_{deg} and UCL, or between the LCL and LCL_{deg} (Section 4.7.3.1), contribute to the Degraded state. The remaining points contribute to the Normal state.

Figures 5.31 - 5.35 show the resulting EWMA control charts for the introduced faults, for all the cylinders (cylinder 1 - cylinder 5). From these figures, it is observed that the artificial faults that are introduced on the 18th of January 2017 are successfully detected, as the EWMA of the residuals for all cylinders exceed the UCLs. Similarly, Figure 5.36 shows the produced aggregated results, in terms of Normal, Degraded and Failed states for each cylinder.



Figure 5.31: EWMA control chart for Cylinder 1 during the simulated fault



Figure 5.32: EWMA control chart for Cylinder 2 during the simulated fault



Figure 5.33: EWMA control chart for Cylinder 3 during the simulated fault



Figure 5.34: EWMA control chart for Cylinder 4 during the simulated fault



Figure 5.35: EWMA control chart for Cylinder 5 during the simulated fault



Figure 5.36: Aggregated results during the simulated fault for the ME cylinders
5.5.1.3 Network Set-Up

Once the fault-mapping is concluded and the aggregated results for each cylinder (under the effects of the introduced faults) are obtained, the network set-up process takes place. The network set-up includes the specification of the structure of the BN, as discussed in Section 4.7.3.3, and the quantification of the BN using the aggregated results for each cylinder. For the period of interest, the CPTs of the network are populated. Afterwards, evidence (VE and HE) regarding the states of the observable and test nodes are used as input, and the probabilities of the primary and secondary faults, together with the profile of each fault are obtained.

In Figure 5.37, the resulting layout of the diagnostic BN is shown. The top nodes (i.e. Cyl 1- Cyl 5) represent the state of each cylinder (Failed, Degraded, Normal), based on the residuals' location in the EWMA control chart for that cylinder. For example, during the period of interest the condition of cylinder 1 is summarised as 26% in the Failed state, 2% in the Degraded state and 73%in the Normal state, as also summarised in the upper left chart of Figure 5.36. The nodes representing the cylinders are the observable nodes, as discussed in Section 4.7.3.3. The next layer represents the control nodes, which are tasked with accessing if a simultaneous increase in the ME EGT of all the cylinders takes place. The state of these nodes is binary (True or False) and their purpose is to propagate and evaluate the information from the observable nodes. The next two layers represent the primary and secondary fault nodes, each of which has a Normal or Abnormal state. Lastly, the lowest layer represents the test nodes which help to quantify the probability of a specific fault occurring and have Pass or Fail states. The CPTs of the test and the control nodes are populated based on logical rules depicting functional dependencies in the network. Lastly, the CPTs of the primary and secondary nodes are populated by obtaining failure statistics from the OREDA data-bank and using logical rules. For instance,

using the OREDA statistics and without using evidence in the test nodes, it is observed that the AC function is 97% normal. The 3% abnormal state of the AC is attributed to the propagation of the increased Failed state of the cylinders, demonstrating the network's ability in evaluating the condition of certain ship systems. The structure of the BN in Figure 5.37 corresponds to output of the fault-mapping process shown in Table 5.13 As can be observed, the BN in Figure



Figure 5.37: Initial diagnostic BN

5.37 creates a visual model summarising the condition of the vessel. This graph covers the period up until the end of the introduced artificial fault. Without using evidence to specify the presence of a fault, the condition of the cylinders is shown in the observable nodes. Each cylinder has a high percentage in the Failed state, which is attributed to the presence of the artificial fault.

However, on the 23rd of January 2017, the EWMA control charts indicate that

the ME cylinders are on the "Failed" state, as presented in Figures 5.31 - 5.35, due to the presence of the simulated fault. Consequently, HEs are used to specify the Failed state in each observable (cylinder) node. As a result, each cylinder is 100% at the failed state. Comparing Figures 5.37 and 5.38, this has a great effect in the probabilities of primary and secondary fault nodes. For example, using the information from above and without utilising the test nodes the abnormal state of the AC increases to 20%.



Figure 5.38: Diagnostic BN with with observable nodes at Failed state

To investigate the root-cause of the detected fault, the diagnostic tests must be performed and their outcomes used as input in the test nodes. Assuming a simultaneous increase in the EG temperature and a failed pressure drop test (Table 5.13) in the AC, fouling in the air-side of the AC is detected (Figure 5.39).



Using HE, the state of the DP test node is specified at 100% Failed. Therefore, the

Figure 5.39: Fully investigated diagnostic BN

network identifies that the air-side of the AC is at 100% Abnormal state, whilst also increasing the abnormal state of the AC to 86% Abnormal, identifying the root-cause of the abnormal simultaneous increase in the EG temperature.

To further investigate the behaviour of the mapped faults (fault profile), the gradual transitions between each state of the observable and test nodes take place. The gradual transition between states is represented using VE, whereas the fully observed states are described using HE. This transition concludes at the Failed state in all the observable nodes and the Fail state in the test nodes. In more detail, HEs are used to demonstrate the ability of the BN to calculate the probability of each fault. To achieve this, the following two conditions are met: a) the observable nodes are at the Failed state, due to the artificial fault and b)

Case Number	Failure Modes
AC Fault 1	AC Water-side fouling
AC Fault 2	AC Air-side fouling
AC Fault 3	AC Air-side fouling and
	AC Water-side fouling
AG Handling System Fault 1	AF Fouling
AG Handling System Fault 2	Corroded TC mechanical components
AG Handling System Fault 3	TC fouling
	AF Fouling Corroded and
AG Handling System Fault 4	TC mechanical components and
	TC fouling
AG Handling System Fault 5	AF Fouling and
	Corroded TC mechanical components
AG Handling System Fault 6	Corroded TC mechanical components and
	TC fouling
AG Handling System Fault 7	AF Fouling and
	TC fouling

Table 5.15: Failure modes summary for the primary faults

appropriate test nodes are set to the Fail state which simulates the cause for the artificial fault. The application of the VE follows the same principle, but their use allows to capture the profile of each fault and examine the rate with which it develops.

The resulting fault profiles for the primary and secondary faults are shown in Figure 5.40 and Figure 5.42 respectively. Moreover, the primary faults are examined in terms of all the different failure modes. These failure modes are shown in Table 5.15, and represent different combinations which can cause either the AC or the AG handling systems to fail.

Regarding the primary faults, Figure 5.40 shows their fault profiles based on all the possible causing combinations (failure modes), as detailed in Table 5.15. In this case, the lower three lines represent failure modes of the AC, whereas the remaining represent the failure modes of the AG handling system. As it can be observed, the failure of the AG handling system is more likely than the failure of the AC. As a consequence, any simultaneous deviation in the ME EG temperature is more likely to be caused by a fault in the AG handling system. In particular, the most likely failure mode corresponds to the simultaneous failure of all the components of the AG handling system. Figure 5.41 shows the average rate of change of the primary faults. In that plot the y-axis denotes the average rate of change of the probability of a fault and the x-axis shows the considered faults. The faults of the AC are colour-coded in teal and the faults of the AG handling system in blue. From that chart, it is observed that the simultaneous fouling in both the air-side and the water-side of the AC has the largest rate of increase and therefore can develop the quickest.



Figure 5.40: Fault profiles for the primary faults

In Figure 5.42, the fault profiles for the secondary faults are shown. In that case, it can be observed that the highest fault profiles belongs to the corroded TC mechanical components, which is then followed by the AF fouling. Therefore, the most likely secondary fault is the corrosion in the mechanical components of the TC. Consequently, any faults manifested through the simultaneous increase of the ME EGT, are most likely attributed to the corrosion of the mechanical components of the TC. Figure 5.43 shows the average rate of change of the sec-



Figure 5.41: Average rate of change of primary faults

ondary faults. From that chart, it is observed that the corrosion of the mechanical components of the TC has the largest rate of increase and therefore, can develop the quickest.



Figure 5.42: Fault profiles for the secondary faults



Figure 5.43: Average rate of change of secondary faults

5.6 Discussion of Results

This section aims to summarise the primary outcomes of the case studies and discuss the impact of the proposed framework. The different methodologies of the proposed predictive maintenance framework are applied in different casestudies, each aiming at demonstrating the effectiveness of the different parts of the framework.

The proposed predictive maintenance framework enhances maritime maintenance by combining reliability-based and data-driven tools in a novel way. The proposed framework examines the areas of critical equipment selection, data preparation, FD and diagnostics in unique methodologies, assessed in the presented case studies.

The first case study that is considered is the critical equipment selection part of the framework, which is applied in vessel "A". During this case study, a ship system analysis is performed, which creates a useful base for the implementation of the FTA. Translating the ship systems analysis in an FT structure creates a visual model demonstrating various functional and systemic interdependencies. Quantifying the resulting FT structure and using the exact calculation method, the I^B and I^{CR} are obtained, which allow the ranking of the modelled components based on two different metrics. Calculating both the I^B and the I^{CR} enables the criticality examination of the components both in terms of their reliability and in terms of their functional dependencies.

By combining the obtained IMs with the repair costs of the modelled components, an initial scatter plot of the data-space is created. Then, the use of the k-means clustering algorithm allows for the segmentation of plotted components in three groups, including a) critical components, b) medium criticality components and c) safest components. The critical components are located in the cluster whose centroid has the biggest distance from the beginning of the axes. Based on the above, it is suggested that the maintenance actions are focused on the critical components, as they have the following characteristics. They have the biggest influence on the rate of change of the probability of a ship failure, due to systemic dependencies. Also, they are the most critical in the modelled system, by considering the components' reliability and failure rate. Lastly, they are the most costly to maintain after a failure, as they have the highest repair and replacement costs. Focusing maintenance actions on the critical components have a positive effect on the daily operations of ships, by improving their overall availability and reliability. Similarly, this part of the framework can be used as a maintenance prioritisation tool for ship owners and operators.

Once the critical equipment is identified, the case study of the data preparation methodology of the proposed framework is applied. This part of the framework focuses on imputations from a holistic view, including all the necessary preand post-imputation steps. The drive of the data preparation methodology of the framework is to improve the quality and knowledge extracting potential of data.

The data preparation methodology establishes a hybrid imputation approach based on the MICE and kNN imputation algorithms. The superior performance of the proposed novel imputation method is compared against the existing kNN and MICE methods. It is demonstrated in the case of a ME and TC system of vessel "B". In total, eight variables are examined including the ME power, ME speed, ME scavenging air pressure, TC EG inlet temperature, TC EG outlet temperature, TC LO outlet temperature, TC LO inlet pressure and TC speed.

A key outcome of this case study is the investigative comparison between the kNN, MICE and the proposed hybrid method for imputation purposes, which exhibits superior performance. In detail, the proposed approach has a mean error of 2.21% compared to the MICE, kNN algorithms with errors of 3.3% and 5.6%, respectively, highlighting that the small error of the proposed hybrid method improves the quality of data. Since the proposed approach is based on a combination

of data-driven tools and FP knowledge, the small error of the method highlights the importance of using FP knowledge in the prediction of measurements from an engineering system, compared to pure data-driven approaches.

The use of the data preparation methodology of the developed framework can enhance the accuracy of maritime predictive maintenance. Ensuring the quality of data, while maximising their knowledge extracting potential can improve various aspects of ship operation. It can improve maintenance planning by providing accurate estimates for the condition of the vessel, which also has a positive effect on ship safety.

Following the data preparation case study, the results FD part of the framework applied on vessel "C" are presented. This case study provides an application for an ML and data-driven FD methodology, based on EB modelling and EWMA control charts. The ultimate goal of the FD is to allow for preemptive rectifying actions and maintenance scheduling. The EB modelling approach is used to predict the EB of the ME cylinder EG temperature. The ME EG temperature is selected, as several faults in the ME's supporting systems can manifest through this variable. Then, the residuals between the expected and recorded ME cylinder EG temperature are analysed in an EWMA control chart to detect developing faults by means of the UCL and LCL.

During the development of the EB model, several regression models are examined, and the optimal model is selected, using k-fold cross-validation. In more detail, multiple linear ridge regression and multiple polynomial ridge regression models are compared. In addition to that, the optimal value of the α hyperparameter is examined, together with different predictor variables, including the ME power, speed, scavenging air pressure and scavenging air temperature. From the analysis, it is seen that on average, the polynomial models have mean validation R^2 score of 0.94 and linear models have a mean validation R^2 of 0.83. Lastly, the identified optimal model is based on multiple polynomial ridge regression with $\alpha = 0.4$ and inputs the ME power, speed and scavenging air pressure.

The identified model is used to obtain the residuals of the target variable, which are then analysed in an EWMA control chart. Healthy operating data are used to fine-tune the L hyperparameter of the control chart before 4 different artificial faults are simulated to assess the detection capabilities of the proposed method.

The use of the FD methodology of the developed framework can improve maritime predictive maintenance. The early detection of developing faults has a profound effect on the safety of vessels, their availability and operational efficiency. By identifying faults, extensive failures are avoided while allowing for efficient maintenance planning. Also, this part of the framework can mitigate the risk of inefficient ship operations, as degraded machinery operations are detected.

The last case study that is presented is on the developed diagnostics methodology of the proposed framework. This part of the framework is also applied on vessel "C". The developed methodology combines the use of ML for FD and BNs for diagnostics. As seen in this case study, the diagnostic network uses the evidence of detected faults in the ME cylinders and seeks possible root-causes in the AC and AG handling system of the ME.

The diagnostics case study demonstrated the creation of a practical diagnostic network which allows for the real-time assessment of operational data to compute accurate probabilities of different faults. Moreover, fault probabilities are used to better understand the operation state of the ME cylinders, ME AC and ME AG handling system.

The case study of the diagnostics frameworks results that the fastest developing faults (largest mean rate of change) are the simultaneous fouling in both the air-side and the water-side of the AC and the corrosion of the mechanical components of the TC. Similarly, the most likely faults are the simultaneous failure of all the components of the AG handling system and the corrosion in the mechanical components of the TC.

The use of the diagnostics methodology of the developed framework can augment maritime predictive maintenance. The diagnostics of developing faults has a positive effect on the operational efficiency of ships. By identifying the root-cause of a fault, targeted maintenance actions are enabled, reducing the downtime of ship systems. Also, this part of the framework can mitigate the risk of inefficient ship operations, as degraded machinery operations are detected.

5.7 Chapter Summary

This chapter discusses the main results of the novel predictive maintenance framework applied in the cases studies of vessels "A", "B" and "C". Initially, the results of the critical equipment selection methodology are presented. This methodology is applied in vessel "A", and combines FTA with k-means clustering. In more detail, the methodology integrates reliability IMs repair costs to select critical components systematically. Then, the outcomes of the data preparation methodology are presented. This methodology is applied in the case of vessel "B" and showcases the effectiveness of a hybrid imputation method, based on MICE and kNN, in predicting values from missing instances. The results from the FD methodology then follow, which are obtained from vessel "C". The FD methodology combines the DBSCAN algorithm for data checking, with a regression-based EB model and the use of EWMA control charts for FD. This methodology includes the systematic investigation for the optimal regression model, which results in a multiple polynomial ridge regression model using as inputs the ME power, speed and scavenging air pressure. Lastly, the results of the diagnostic methodology are shown. This methodology is also applied in vessel "C" and demonstrates the integration of ML-based FD with BN diagnostics.

Chapter 6

Discussion and Conclusions

6.1 Chapter Overview

This is the final chapter of this thesis and discusses the conclusions drawn from the establishment of the novel predictive maintenance framework. The fulfilment of the aim and objectives is deliberated first, before presenting the generated novelty. Then, the concluding remarks and reflections are presented, followed by recommendations for future work and the discussion of the main assumptions and challenges.

6.2 Fulfilment of Aim and Objectives

The main drive of this research is to enrich, in a practical and theoretical manner, the area of maritime predictive maintenance for ship machinery systems. This was accomplished by focusing on the research question presented in Section 2. As a result, this work was directed towards developing a framework encompassing the areas of critical equipment selection, data preparation, detection and diagnostics of developing faults. In detail, the set objectives of this work and how they were fulfilled are discussed in depth in this Section.

Objective 1: The investigation of the relevant literature regarding maintenance strategies, reliability assessment and data-drive predictive modelling in a factual and critical manner, in order to identify gaps and direct the novelty of the present research. This objective was achieved through the factual and critical review of the relevant literature, presented in Section 3, and the generation of the novelties, detailed in Section 4.2 and summarised in Section 7.3. The review of the literature identified several gaps in the maritime industry regarding the applications of PdM. It was uncovered that there is a gap in addressing the topics of critical equipment selection, data imputation, FD and diagnostics in a single research effort. In more detail, maintenance concepts and frameworks are examined. Then, the different categories of tools used in predictive maintenance are presented, including reliability assessment and data-driven approaches. Based on the identified tools, the resulting predictive maintenance process are explored, which are aligned with the remaining objectives. Lastly, by comparing the predictive maintenance status quo between the maritime sector and other industries, several gaps are identified. These gaps are presented in detail in Section 3.9 and stem from the limitations of the maritime predictive maintenance process, in terms of data preparation, fault detection and diagnostics. Moreover, the main identified gap is the absence of a complete predictive maintenance framework taking into account the particular needs of the maritime industry and providing data-driven and knowledge-based solutions. The identified gaps are subsequently used to direct the novelty of this research.

Objective 2: The proposal of a novel, data-driven and reliability-based predictive maintenance framework, tailor-made for the needs of the maritime industry. The completion of the objective was based on the development of the novel compound framework, together with its methodological components. The novel framework is presented in Section 4, and its flow is illustrated in Figure 3.1. The development of this novel framework addresses the gap for a complete predictive maintenance framework taking into account the particular needs of the maritime industry. In more detail, it includes the novel integration critical selection, data preparation, fault detection and diagnostic methodologies in a single framework. From a holistic point of view, this work initiates by identifying high-risk (functional and economical) ship equipment, which can be used as a starting point for predictive maintenance framework. Then, the data preparation methodology improves the quality and knowledge extracting potential of the data used in the fault detection and diagnostic tasks. Lastly, the fault detection and diagnostic methodologies identify developing anomalies, together with their root causes, by using treated data from specially identified equipment.

Objective 3: The development of a novel methodology for the identification of the critical equipment of ship systems, aimed at prioritising maintenance efforts. This objective was completed by developing a novel methodology for the identification and selection of critical equipment and components of ship systems. Through this objective, it was uncovered that it is necessary to combine different reliability metrics for the identification of critical components, as the same items rank differently with different metrics. Also, reliability metrics must be combined with cost information for practical maintenance prioritisation. Moreover, this objective highlighted the importance of clustering different components as it can effectively group and summarise them based on their importance. This methodology is detailed in Section 4.4 and illustrated in Figure 3.2. The novelty of this part of the frameworks stems from addressing the gap of combining data-driven and reliability-based tools for the identification of critical components. Similarly, it also stems from addressing the gap in combining cost aspects with reliability characteristics for identifying critical components. Following the collection maintenance and repair cost data, this methodology performs a ship systems analysis which leads to a fault tree analysis. The results from the fault tree analysis are combined with additional cost information in a clustering analysis which leads to the identification and selection of the critical equipment. The creation of a methodology able to identify high-risk ship equipment allows for targeted and efficient maintenance prioritisation.

Objective 4: The establishment of a novel data preparation methodology, concerned explicitly with handling missing values from data sets used on condition monitoring tasks. The completion of the objective was based on the development of a novel data preparation methodology, aiming at improving the quality and value of data. This objective uncovered that the performance of the developed hybrid imputation approach (2.2% mean error) is superior compared against the state-of-the-art MICE (3.3% mean error) and widely used kNN (5.6% mean error) algorithms. Also, it was concluded that imputation methods that take into account the correlation between variable perform better. Moreover, a major advantage of the developed imputation approach is that it can be used as a virtual sensor enhancing the accuracy of data-driven models and preserving otherwise lost information from DAQ systems. This methodology is detailed in Section 4.5 and illustrated in Figure 3.4. The novelty of this part of the framework originates from establishing a formalised data preparation approach for the maritime industry. Also, the novelty of this part of the framework stems from the novel combination of data-driven solutions with domain knowledge for the imputation of missing values. The starting point of this methodology is the form handling, synchronisation and filtering of the data. Then, a new imputation approach is used to predict missing values from the dataset. The effectiveness of the new imputation approach is established through the comparison with prominent MICE and kNN imputations algorithms. Enhancing the quality of data can have a positive effect on the resulting predictive maintenance framework, as misdirecting information are removed and missing trend-describing points are restored.

Objective 5: The development of a novel fault detection methodology, lessening the amount of data-associated assumptions, and tailored to the needs of the mar*itime industry.* This objective was completed by establishing an innovative fault detection methodology by modelling and monitoring the expected behaviour of systems. This objective resulted in the creation of an FD model that can allow for preemptive rectifying actions and maintenance scheduling, greatly improving ship operations. Also, another benefit for the created model is that the selected monitored parameter (ME EG temperature) can be used to detect various faults from different systems. Similarly, the completion of this objective required the investigation of the optimal regression EB model, resulting in an optimised multiple polynomial ridge regression model with a testing R^2 score of 0.96. This methodology is detailed in Section 4.6 and illustrated in Figure 3.5 and Figure 3.6. The novelty of this part of the framework stems from addressing the gap of the application of an FD model taking into account the particular needs of the maritime industry and addressing the limitations imposed by the available data. The fault detection process starts with a supplementary data filtering and outlier detection process, exploiting the benefits of application-agnostic tools. Then, the optimal expected-behaviour model is identified through a structured comparison of different regression models, under the effect of different inputs. The resulting model is used to gauge the deviation between the expected behaviour of a selected signal and its recorder behaviour from an incoming dataset. To facilitate this comparison, a signal-smoothing control chart is used. The early detection of developing faults can significantly enhance safety and operational efficiency, through monitoring specially selected signals with fault indicative potential.

Objective 6: The establishment of a diagnostic methodology combining in a novel way, machine learning applications with domain knowledge for practical applications of ship systems. This objective is accomplished by developing a novel diagnostic methodology, integrated with a fault detection module, for practical applications of ship systems. This objective resulted in one of the few diagnostic models that are integrated with ML-based FD for ship applications. Another benefit is that the resulting model uses evidence of developed faults to identify potential root-cause, thus improving ship operations This results in a practical network that allows for the real-time assessment of the condition of specific systems. Also, this objective examined the rate of change of the probabilities of different faults, proving insight on how certain faults develop. This methodology is detailed in Section 4.7 and is illustrated in Figure 3.8. The novelty of this part of the framework originates from addressing the gap in the application of a diagnostic module combining ML-based FD, with knowledge-based BN diagnostics. A detrimental step in this methodology is the fault-mapping process, in which monitored variables are paired with potential faults that can manifest through trends in their signal. Once the monitored signals are paired with potential faults, the method for summarising the results from the FD step is developed. The last step of the diagnostic methodology is the specification of a diagnostic network, which draws as input the outcome of the fault-mapping process. The creation of the diagnostic network allows for the creation of a visual model that is easy to interpret. It provides with the capability of assessing in real-time the operational status of the examined model.

Objective 7: The demonstration and validation of the effectiveness of the proposed predictive maintenance framework through different case studies, such as the main engine of a bulk carrier. The requirements of this objective are met by evaluating the performance of the different methodologies in different case studies. These case studies are described in Section 5, and their outcomes are presented in detail throughout Section 6. The different case studies relate to various vessels, but in each case, they are mainly structured around the main engine of the respective ship. Moreover, all the case studies aim at showcasing the capabilities of the different methodologies by also taking into account potential limitations during development. In detail, the case studies evaluate the framework's capabilities in a) identifying critical equipment, b) adequately preparing data for subsequent analysis, c) detecting in a timely manner developing faults and d) identifying the root-cause of detected faults. The case study for the critical equipment selection is described in Section 5.2, expanded in Section 6.2 and applied in an 81000 DWT LNG carrier. During this case study, maintenance information is used to quantify the developed fault tree. The obtained reliability IMs together with repair costs are clustered together to identify the critical equipment. The case study for the data preparation is described in Section 5.3, expanded in Section 6.3 and is applied in a 38000 DWT chemical tanker. During this case study operating parameters from the vessel's ME are used to develop an imputation approach and compare it against common practices. The developed approach, based on domain knowledge and data-driven approaches, can predict missing values with the biggest accuracy and consistency. The fault detection case study is described in Section 5.4, detailed in Section 6.4 and is applied in a 64000 DWT bulk carrier. Throughout this case study, variables from the vessel's ME are used. The collected variables are used to develop and select the optimal regression model, from a pool of compared models. The selected model is used to produce the expected behaviour of a selected variable, which is then compared with its recorded behaviour. Analysing the deviation between these two values, a signal-smoothing control chart allows for the detection of developing faults. Lastly, the case study for the diagnostics is described in Section 5.5, detailed in Section 6.5 and is applied in the same 64000 DWT bulk carrier. In this case study, variables from the vessel's ME are also used. Combing domain knowledge with information for the ship's ME manufacturer a fault mapping process is carried, which heavily influences the developed Bayesian diagnostic network. The developed diagnostic network is used to evaluate in real-time the condition of the ship by examining the probabilities of different faults in the supporting systems of the ME.

Objective 8: The discussion of the main outcomes of the developed framework

together with suggestions for future work. This objective is fulfilled in the current chapter, which outlines the generated novelty, provides the main reflections and conclusions of this research and directs any potential future work.

6.3 Generated Novelty

The novelty of the present research stems from the development of the compound maritime predictive maintenance framework. Keeping in line with the stated research aim, the research and development related objectives are summarised in four distinct categories, including critical equipment selection, data preparation, fault detection, and diagnostics. The established maritime predictive maintenance framework introduces novel aspects in the areas of critical equipment selection of ship systems, data preparation for maritime applications, and fault detection and diagnostics of ship systems and components. Similarly, the established framework combines, in a novel manner, reliability-based and data-driven models. The novel combination of these tools is innovative and address practical problems of the maritime industry. Due to the use of established reliability tools, the developed framework benefits in terms of robustness. Similarly, the datadriven aspects incorporate innovative applications which future-proof the developed framework. The data-driven methodologies of the proposed framework can be implemented in a variety of different settings. They can be integrated with different applications due to the inherent interoperable nature of these tools.

A novel aspect of this work is the combination of fault tree analysis with kmeans clustering for the identification of critical components. This methodology uses as input maintenance information from vessel "A" to obtain reliability IMs, by performing a fault tree analysis. Then, the obtained IMs are clustered together with cost information to identify critical equipment. This methodology combines in a novel manner the use of fault tree analysis with k-means clustering. Moreover, the consideration of cost information for the identification of critical equipment is ship systems is also novel. In general, the application of this methodology in LNG ship systems represents a novel application. Apart from the generated novelties, this methodology has a practical impact on the shipping industry. It can be used to prioritise maintenance actions in high-risk equipment, improving the ship's availability and mitigating the risks of breakdowns. From a similar standpoint, this framework can be used to optimise the procurement and distribution of spare parts, to reduce the risks of breakdowns further. Lastly, this methodology can provide a starting point for more in-depth reliability analysis of ship systems.

Another novel part of the present research is the implementation of a hybrid imputation approach, developed as a part of the data preparation methodology. This methodology uses as input performance and process variables from vessel's "B" ME. The collected data pass through a basic pre-processing step before they are used to develop the hybrid imputation tool, which is based on the MICE and kNN algorithms. Using these two algorithms enables the combination of datadriven solutions with domain knowledge in a single approach. Once the hybrid imputation approach is developed, its performance is evaluated against the MICE and kNN imputation algorithms. This methodology integrates in a novel way the MICE and kNN imputation algorithms in an approach with superior performance. From a theoretical point of view, this methodology includes the novel combination of data-driven models with domain knowledge in a single imputation approach. Moreover, the application of this methodology in ship systems is also novel. Apart from the discussed novel aspects, this methodology has practical significance in shipping. Initially, it can be used as a virtual sensor, ensuring the continuous collection of required data. It enriches the literature and addresses a practical problem encounter by maintenance planners. This methodology can be used to improve the quality of data sets used in predictive maintenance, enhancing the accuracy of subsequent models. As a result, decision-making processes are more accurate and lead to more effective actions.

Another novel aspect of this work is the combination of regression-based expected behaviour modelling with the exponential moving average control chart for fault detection. This particular combination allows for the creation accurate regression models that are not depend on very large training sets and detect developing faults while filtering out noise (Cheliotis, Lazakis, and Theotokatos 2020). This part of the framework considers the use of performance and process variables from vessel's "C" ME. The DBSCAN algorithm is deployed to remove outliers from the dataset before the data are used to train and validate different regression models. Once the optimal model is selected and tested, the residuals of a chosen variable are obtained and analysed in the exponential moving average control chart for fault detection. This methodology combines in a novel way the investigation of different regression models for expected behaviour modelling with control charts for fault detection. Moreover, the use of the DBSCAN algorithm for outlier detection, coupled with the presented fault detection approach for ship applications is also novel. The expected behaviour approach has the distinct advantage of detecting faulty operating conditions without requiring scarce faulty labelled data for model training. Also, this approach has enhanced applicability when compared with traditional classification approaches. This is due to the time-series output of the expected behaviour approach, which is more useful and interpretable. From a practical point of view, this model can be used to improve the safety of ship systems and to reduce the number of breakdowns.

The final novel part of this research is the combination of machine learningbased fault detection with a Bayesian diagnostic network. This part of the framework also uses as input performance and process variables from vessel's "C" ME. By aggregating the results from fault detection and by carrying out a faultmapping process, the structure of the Bayesian diagnostic network is specified. This methodology includes the novel combination of a machine learning-based fault detection module with a Bayesian diagnostic network, in a novel application for ship systems. From a theoretical point of view, this methodology combines in a novel manner, data-driven approaches with knowledge-based models in an integrated diagnostics application. The use of this methodology has several practical benefits for the maritime industry. It can enable ship operators to get a realtime assessment of the condition of their vessels. As a result, better maintenance planning is allowed, which increase the safety of ships and ship systems.

6.4 Reflections

The following statements contain the concluding reflections of this work:

- The presented framework is established through the examination of the relevant literature investigating trends and common practices of predictive maintenance in the maritime and other sectors. The established requirements for a holistic framework encompassing the individual methodologies strengthen the framework presented in this work.
- In line with the above, this thesis presented a reliability-based and datadriven framework addressing a multitude of issues of predictive maintenance. The resulting framework combines different data-driven and reliability based tools in a practical schema with proven accuracy. Maritime predictive maintenance issues can be tackled by either using the framework as a stand-alone solution or by employing the individual methodologies.
- The developed maritime predictive maintenance framework initiates with the critical equipment selection methodology. The methodology includes a ship system analysis, fault tree analysis, reliability IMs assessment and data clustering. It is possible to identify critical equipment systematically, using objective criteria and incorporating various aspects. This method-

ology enhances the manner in which critical components are identified by combining reliability indices with cost information. The use of additional criteria is enabled by using easy to expanded partitioning algorithms. As a result, ship operators can identify critical components in a tailor-made manner, by including parameters and metrics based on their requirements and goals.

- The data preparation methodology is the subsequent step in tackling mar-• itime predictive maintenance. The steps of this methodology include data form handling and synchronisation, data filtering, imputation of missing values and correction to ambient conditions. The development of the data preparation methodology enriches the relevant literature, which is very limited for maritime applications. Imputation in shipping is overlooked, but it remains essential. This is particularly true with the increasing application of machine learning algorithms, which are sensitive to missing values. In essence, the proposed imputation approach acts as a virtual sensor, ensuring the uninterrupted collection of required data. From the developed methodology, it is observed that when attempting to predict missing values from highly correlated variables, imparting domain knowledge in datadriven modes can have a positive effect. On the other hand, when performing imputation of uncorrelated values, increasing the model's complexity can have a limited influence.
- The fault detection methodology is the following step in addressing maritime predictive maintenance. This methodology incorporates the investigation of the optimum expected behaviour model, the evaluation of the identified model, the calculation of the residuals of a target variable, and the analysis of the residuals in a signal-smoothing control chart. The developed methodology allows for the prediction of faults without requiring

faulty labelled data during model training. However, the selection of the model's inputs should be thoroughly and systematically investigated. This expedites the developing process, resulting in models that are quicker to deploy and can offer accurate predictions decoupled from operating conditions. By detection faults using an expected behaviour approach, the need for traditional one-class classifiers is bypassed. This increases flexibility and eases the model's integration with subsequent diagnostic tasks. Due to the reduced number of assumptions, the developed methodology can be expanded and implemented in different engineering systems, increasing the framework's impact on safety.

• The developed diagnostics methodology is the final step of maritime predictive maintenance framework. The steps of this methodology are, the aggregation of the results from fault detection, the fault mapping structure, the specification of the structure of the diagnostic network, and the investigation of different faults. This methodology demonstrates the integration of a machine learning-based fault detection module, with a knowledge-based diagnostic network. As a result, this methodology enriches limited maritime literature regarding integrated fault detection and diagnostics. The developed network is also expandable and can be used to model additional systems; however, this process depends heavily on additional failure statistics which can be hard to obtain. Shipowners and operators can use the developed diagnostic network in order to improve their understanding, in real-time, in terms of the condition of their vessels.

6.5 Recommendation for future work

The developed framework, together with the different methodologies contained within, contribute to the area of maritime predictive maintenance. Future research directions that can further enhance the impact of this work are presented below. These directions reflect both to the individual methodologies and the framework as a whole:

- The fault tree structure developed for the critical equipment selection can be expanded to account for greater modelling detail and include dynamic gates. These gates can model with greater accuracy the dynamic dependencies between components. Moreover, additional gates and events can be used to model with greater accuracy the ship systems, by incorporating additional components, currently omitted for simplicity reasons. However, this process should always be a function of the available computational power.
- The reliability and cost criteria used to identify the critical equipment can be expanded to include additional information, both from a practical and from a theoretical standpoint. For example, the availability of spares could be factored in order to provide application-specific maintenance prioritisation.
- Additional clustering algorithms can be explored when partitioning the critical equipment data-space. The used k-means algorithm creates hard clusters, where each component has a binary membership in each cluster. Exploring algorithms which create soft clusters, where each component has a gradual membership in different clusters could be beneficial. For instance, the *c*-means clustering algorithm can be used to improve the representation of components which are currently located near the boundary of each cluster.
- To further establish the effectiveness of the developed imputation approach, it would be beneficial to compare it with additional imputation methods, apart from the MICE and kNN algorithms. For instance, vertical imputation approaches, or additional machine learning tools (e.g. random forests)

can provide an extended basis for comparison. Similarly, the effectiveness of the developed approach can be evaluated through the use of additional variables during the comparison of the different imputation approaches.

- The data filtering process used during the data preparation methodology can be replaced with application-agnostic tools. For example, instead of using a value-based filter, the DBSCAN algorithm, or other similar algorithms, could be employed.
- Regarding the data preparation methodology, it would be beneficial to holistically assess its impact on ship condition monitoring (e.g. fault detection, diagnostics). In more detail, this would involve performing condition monitoring tasks with treated and untreated datasets and evaluating the obtained results.
- The fault detection methodology could be enhanced by exploring additional expected behaviour models. Even though the selected expected behaviour model obtained excellent training, validation and testing scores, its performance could be compared with random forest, or support vector machine regression models.
- The outlier detection algorithm used during the fault detection methodology (DBSCAN) could be compared with additional density-based clustering techniques (e.g. OPTICS algorithm).
- Since the aim of the expected behaviour models is to classify each point in terms of its state, a direct comparison between this approach and the traditional classification approach would be beneficial. For example, comparing the regression-based expected behaviour model with a one-class classifier (e.g. support vector machine) would offer an insight into the performance of these two approaches.

- The effectiveness of the expected behaviour model could be expanded in predicting additional signals (e.g. ME speed).
- The effectiveness of the exponentially weighted moving average in analysing the residuals, during fault detection, could be compared with additional control charts (e.g. cumulative sum charts).
- The structure of the diagnostic Bayesian Network can be expanded in future research to include additional observable nodes (e.g. ME speed and ME power). However, this process is a function of the available data.
- Future research efforts could also compare the performance of the developed diagnostic network with alternative knowledge-based approaches.
- Future research efforts could bypass the use of data banks for the nonobservable nodes. Instead, combining reliability tools with Monte Carlo simulations could be used as an alternative.
- Finally, future research efforts could examine the overall effectiveness of the developed framework in a single, continuous, case study.

6.6 Assumptions and Challenges

To meet the main research objectives and establish the novelties of this thesis, several assumptions were made, and different challenges had to be overcome. These two topics are presented and discussed in this Section.

The main assumptions of this thesis represent the basic premises based on which this work is developed. The first assumption reflects the manner in which the various methodologies of the framework are evaluated. Due to the data limitations discussed in Section 5, different case studies are used to evaluate the capabilities of the individual parts of the framework. Also, the individual methodologies of the proposed framework were not developed simultaneously. Regarding the critical equipment selection aspect of the framework, it is assumed that the maintenance schedule of the examined vessel is used to obtain information for the quantification of the FT structure. Similarly, regarding the data preparation part of the framework, operating manual supplied for ship operators are used for data filtering. The main assumption of the FD methodology of the proposed framework is based on the retraining of the EB models after major modifications and repairs on the examined vessel. Lastly, the diagnostics methodology is based on the assumption that data banks can be used for failure statistics in the absence of required data.

The majority of the challenges of this thesis are relating to the required resources. Initially, there was an absence of a single dataset containing all the required information necessary to develop and evaluate the different components of the framework in a single case study. Moreover, some of the individual datasets were still missing some information, resulting in the use of alternative sources (e.g. data banks). Lastly, the size of the analysis is some parts of the framework (e.g. during the FTA) was restricted due to the limited computational power.

6.7 Chapter Summary

This chapter presented concluding remarks and reflections for the established predictive maintenance framework. Initially, the fulfilment of the main aim and objectives is elaborated. Then, the novelty generated from this work is summarised and outlined before presenting concluding reflections and remarks. Lastly, this chapter presented recommendations and directions for future work and discussed the main assumptions and challenges. List of references and appendices are provided next, supplementing this work.

Bibliography

- ABS (2016). Guide for Surveys Based on Machinery Reliability and Maintenance Techniques. Tech. rep. Houston: ABS. URL: www.eagle.org.
- Adegoke, Nurudeen A., Saddam Akber Abbasi, Abdaljbbar B.A. Dawod, and Matthew D.M. Pawley (June 2019). "Enhancing the performance of the EWMA control chart for monitoring the process mean using auxiliary information". In: Quality and Reliability Engineering International 35.4, pp. 920–933. ISSN: 10991638. DOI: 10.1002/qre.2436. URL: https://onlinelibrary.wiley. com/doi/abs/10.1002/qre.2436.
- Afzali, Peyman, Farshid Keynia, and Masoud Rashidinejad (2019). "A new model for reliability-centered maintenance prioritisation of distribution feeders". In: *Energy* 171, pp. 701–709. ISSN: 03605442. DOI: 10.1016/j.energy.2019.01.
 040. URL: https://doi.org/10.1016/j.energy.2019.01.040.
- Ahmed, Nesreen K., Amir F. Atiya, Neamat El Gayar, and Hisham El-Shishiny (Aug. 2010). "An empirical comparison of machine learning models for time series forecasting". In: *Econometric Reviews* 29.5, pp. 594–621. ISSN: 07474938. DOI: 10.1080/07474938.2010.481556. URL: http://www.tandfonline. com/doi/abs/10.1080/07474938.2010.481556.
- Ahn, Junkeon, Yeelyong Noh, Sung Ho Park, Byung Il Choi, and Daejun Chang (Oct. 2017). "Fuzzy-based failure mode and effect analysis (FMEA) of a hybrid molten carbonate fuel cell (MCFC) and gas turbine system for marine propulsion". In: Journal of Power Sources 364, pp. 226–233. ISSN: 03787753. DOI:

10.1016/j.jpowsour.2017.08.028. URL: https://www.sciencedirect. com/science/article/pii/S0378775317310522.

- Ahuja, I. P.S. and J. S. Khamba (2008). "Total productive maintenance: Literature review and directions". In: International Journal of Quality and Reliability Management 25.7, pp. 709–756. ISSN: 0265671X. DOI: 10.1108/ 02656710810890890.
- Alsyouf, Imad (Sept. 2009). "Maintenance practices in Swedish industries: Survey results". In: International Journal of Production Economics 121.1, pp. 212– 223. ISSN: 09255273. DOI: 10.1016/j.ijpe.2009.05.005.
- Amin, Tanjin, Faisal Khan, and Syed Imtiaz (2018). "Fault Detection and Pathway Analysis using a Dynamic Bayesian Network". In: *Chemical Engineering Science* October. ISSN: 0009-2509. DOI: 10.1016/j.ces.2018.10.024. URL: https://doi.org/10.1016/j.ces.2018.10.024.
- Anantharaman, Mohan (2013). "Using Reliability Block Diagrams and Fault Tree circuits, to develop a Condition Based Maintenance Model for a Vessel's Main Propulsion System and Related Subsystems". In: *TransNav, the International Journal on Marine Navigation and Safety of Sea Transportation* 7.3, pp. 409–413. ISSN: 2083-6473. DOI: 10.12716/1001.07.03.13.
- Anantharaman, Mohan, F. Khan, V. Garaniya, and B. Lewarn (2014). "A Step by Step Approach for Evaluating the Reliability of the Main Engine Lube Oil System for a Ship's Propulsion System". In: *TransNav, the International Journal on Marine Navigation and Safety of Sea Transportation* 8.3, pp. 367– 371. ISSN: 2083-6473. DOI: 10.12716/1001.08.03.06.
- Arca, Jesús García and J. Carlos Prado Prado (2008). "Personnel participation as a key factor for success in maintenance program implementation: A case study". In: International Journal of Productivity and Performance Management 57.3, pp. 247–258. ISSN: 17410401. DOI: 10.1108/17410400810857257.

- Armina, Roslan, Azlan Mohd Zain, Nor Azizah Ali, and Roselina Sallehuddin (Sept. 2017). "A Review on Missing Value Estimation Using Imputation Algorithm". In: Journal of Physics: Conference Series. Vol. 892. 1. IOP Publishing, p. 012004. DOI: 10.1088/1742-6596/892/1/012004. URL: http: //stacks.iop.org/1742-6596/892/i=1/a=012004?key=crossref. 4ae7f5f8c76d6b171256642de95c6fcf.
- Arunraj, N. S. and J. Maiti (Apr. 2007). "Risk-based maintenance-Techniques and applications". In: *Journal of Hazardous Materials* 142.3, pp. 653–661. ISSN: 03043894. DOI: 10.1016/j.jhazmat.2006.06.069.
- Assaf, A. George, Mike Tsionas, and Anastasios Tasiopoulos (Apr. 2019). "Diagnosing and correcting the effects of multicollinearity: Bayesian implications of ridge regression". In: *Tourism Management* 71, pp. 1–8. ISSN: 02615177. DOI: 10.1016/j.tourman.2018.09.008. URL: https://www.sciencedirect.com/science/article/pii/S0261517718302164.
- Atoui, Mohamed Amine, Sylvain Verron, and Abdessamad Kobi (Sept. 2015). "Fault detection and diagnosis in a Bayesian network classifier incorporating probabilistic boundary". In: *IFAC-PapersOnLine*. Vol. 28. 21, pp. 670–675. DOI: 10.1016/j.ifacol.2015.09.604.
- Awad, Mahmoud I., Mohammad AlHamaydeh, and Ahmed Faris (Oct. 2018). "Fault detection via nonlinear profile monitoring using artificial neural networks". In: *Quality and Reliability Engineering International* 34.6, pp. 1195– 1210. ISSN: 10991638. DOI: 10.1002/qre.2318. URL: http://doi.wiley. com/10.1002/qre.2318.
- Ayodele, Taiwo Oladipupo (2010). "Types of Machine Learning Algorithms".
 In: New Advances in Machine Learning, p. 366. ISSN: 9789533070865. DOI: 10.5772/711. URL: www.intechopen.com.
- Azadeh, Ali and Saeed Abdolhossein Zadeh (2016). "An integrated fuzzy analytic hierarchy process and fuzzy multiple-criteria decision-making simulation ap-

proach for maintenance policy selection". In: *Simulation* 92.1, pp. 3–18. ISSN: 17413133. DOI: 10.1177/0037549715616686.

- Aziz, Abdul, Salim Ahmed, Faisal Khan, Chris Stack, and Annes Lind (2019).
 "Operational risk assessment model for marine vessels". In: *Reliability Engineering and System Safety* 185.January, pp. 348–361. ISSN: 09518320. DOI: 10.1016/j.ress.2019.01.002. URL: https://doi.org/10.1016/j.ress.2019.01.002.
- Azur, Melissa J, Elizabeth A Stuart, Constantine Frangakis, and Philip J Leaf (2011). "Multiple imputation by chained equations: what is it and how does it work?" In: International Journal of Methods in Psychiatric Research 20.1, pp. 40-49. ISSN: 1557-0657. DOI: 10.1002/mpr. URL: http://www.scopus. com/inward/record.url?eid=2-s2.0-79951999327&partnerID=tZ0tx3y1.
- Badodkar, D N and T A Dwarakanath (2017). "Machines, Mechanism and Robotics".
 In: Proceeding of INACOMM 2017. Ed. by D N Badodkar and T A Dwarakanath.
 Mumbai: Springer, p. 841. ISBN: 978-981-10-8596-3. DOI: 10.1007/978-981-10-8597-0. URL: http://link.springer.com/10.1007/978-981-10-8597-0.
- Bae, Suk Joo, Byeong Min Mun, Woojin Chang, and Brani Vidakovic (Apr. 2019). "Condition monitoring of a steam turbine generator using wavelet spectrum based control chart". In: *Reliability Engineering and System Safety* 184, pp. 13–20. ISSN: 09518320. DOI: 10.1016/j.ress.2017.09.025.
- Balaraju, Jakkula, Mandela Govinda Raj, and Chivukula Suryanarayana Murthy (2019). "Fuzzy-FMEA risk evaluation approach for LHD machine-A case study".
 In: Journal of Sustainable Mining 18.4, pp. 257–268. ISSN: 23003960. DOI: 10.1016/j.jsm.2019.08.002. URL: https://doi.org/10.1016/j.jsm.2019.08.002.
- Baldi, Francesco, Gerasimos Theotokatos, and Karin Andersson (2015). "Development of a combined mean value-zero dimensional model and application for

a large marine four-stroke Diesel engine simulation". In: *Applied Energy* 154, pp. 402-415. ISSN: 03062619. DOI: 10.1016/j.apenergy.2015.05.024. URL: http://dx.doi.org/10.1016/j.apenergy.2015.05.024.

- Bangalore, P. and M. Patriksson (Jan. 2018). "Analysis of SCADA data for early fault detection, with application to the maintenance management of wind turbines". In: *Renewable Energy* 115, pp. 521-532. ISSN: 18790682. DOI: 10. 1016/j.renene.2017.08.073. URL: https://linkinghub.elsevier.com/ retrieve/pii/S0960148117308340.
- Basurko, Oihane C. and Zigor Uriondo (Apr. 2015). "Condition-based maintenance for medium speed diesel engines used in vessels in operation". In: Applied Thermal Engineering 80, pp. 404-412. ISSN: 13594311. DOI: 10.1016/ j.applthermaleng.2015.01.075. URL: https://www.sciencedirect.com/ science/article/pii/S1359431115001040.
- Batista, Gustavo E.A.P.A. and Maria Carolina Monard (2003). "An analysis of four missing data treatment methods for supervised learning". In: Applied Artificial Intelligence 17.5-6, pp. 519–533. ISSN: 10876545. DOI: 10.1080/ 713827181.
- Ben-Daya, Mohamed, Jezdimir Knezevic, Abdul Raouf, and Daoud Ait-Kadi (2009). Handbook of Maintenance Management and Engineering. Ed. by Mohamed Ben-Daya, Jezdimir Knezevic, Abdul Raouf, and Daoud Ait-Kadi. 1st. Vol. 1. 4. London: Springer, p. 745. ISBN: 9781848824713. DOI: 10.1007/978-1-84882-472-0.
- Bengtsson, Marcus (2004). Condition Based Maintenance Systems: An Investigation of Technical Constituents and Organizational Aspects. Västerås. URL: http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.122. 2614&rep=rep1&type=pdf%5Cnhttp://citeseerx.ist.psu.edu/ viewdoc/download?doi=10.1.1.122.2614&rep=rep1&type=pdf.

- Bernal, Esteban, Maksym Spiryagin, and Colin Cole (2019). "Onboard Condition Monitoring Sensors, Systems and Techniques for Freight Railway Vehicles: A Review". In: *IEEE Sensors Journal* 19.1, pp. 4–24. ISSN: 1530437X. DOI: 10.1109/JSEN.2018.2875160.
- Bertsche, Bernd (2007). Reliability in Automotive and Mechanical Engineering.
 1st. Berlin: Springer, p. 502. ISBN: 9788578110796. DOI: 10.1017/CB09781107415324.
 004.
- Bilmes, Jeff (2004). On Virtual Evidence and Soft Evidence in Bayesian Networks. Tech. rep. 206. Seattle: University of Washington. URL: http://www.ee. washington.edu.
- Bishop, Christopher M (2006). Pattern Recognition and Machiner Learning. Ed. by Michael Jordan, Jon Kleinberg, and Bernhard Scholkopf. 1st. Vol. 27. 1. Singapore: Springer, p. 758. ISBN: 9780387310732.
- Borjalilu, Naimeh and Mahmoud Ghambari (2018). "Optimal maintenance strategy selection based on a fuzzy analytical network process: A case study on a 5-MW powerhouse". In: International Journal of Engineering Business Management 10, pp. 1–10. ISSN: 18479790. DOI: 10.1177/1847979018776172.
- Borunda, Mónica, O. A. Jaramillo, Alberto Reyes, and Pablo H. Ibargüengoytia (2016). "Bayesian networks in renewable energy systems: A bibliographical survey". In: *Renewable and Sustainable Energy Reviews* 62, pp. 32–45. ISSN: 18790690. DOI: 10.1016/j.rser.2016.04.030. URL: http://dx.doi.org/ 10.1016/j.rser.2016.04.030.
- Boullosa-Falces, David, Miguel Angel Gomez-Solaetxe, Zaloa Sanchez-Varela, Sergio García, and Alfredo Trueba (Apr. 2019). "Validation of CUSUM control chart for biofouling detection in heat exchangers". In: Applied Thermal Engineering 152, pp. 24–31. ISSN: 13594311. DOI: 10.1016/j.applthermaleng. 2019.02.009.
- Bowerman, Bruce L., Richard T. O'Connell, and Emily S. Murphree (2015). Regression Analysis Unified Concepts, Practical Applications, and Computer Implementation, p. 280. ISBN: 9781412918084. DOI: 10.4135/9781412963947. n461.
- Bradley, David and Peter Hehenberger (2016). *Mechatronic futures*, pp. 1–15. ISBN: 9783319321561. DOI: 10.1007/978-3-319-32156-1{_}1.
- Braglia, Marcello, Davide Castellano, and Mosè Gallo (2019). "A novel operational approach to equipment maintenance: TPM and RCM jointly at work".
 In: Journal of Quality in Maintenance Engineering 25.4, pp. 612–634. ISSN: 13552511. DOI: 10.1108/JQME-05-2016-0018.
- BSI Standards Publication (2010). Maintenance Maintenance Terminology BS EN 13306:2010. London.
- Buuren, Stef van and Karin Oudshoorn (1999). Flexible multivariate imputation by MICE. 1st. Leiden: TNO Institute Prevention and Health, pp. 1–20. ISBN: 9067436232.
- Byrne, Robert (2012). "Beyond Traditional Time-Series: Using Demand Sensing to Improve Forecasts in Volatile Times". In: *The Journal of Business Forecasting* 31.2, p. 13. ISSN: 0278-6087.
- Cai, Baoping, Lei Huang, and Min Xie (2017a). "Bayesian Networks in Fault Diagnosis". In: *IEEE Transactions on Industrial Informatics* 13.5, pp. 2227–2240. ISSN: 15513203. DOI: 10.1109/TII.2017.2695583.
- (2017b). "Bayesian Networks in Fault Diagnosis". In: *IEEE Transactions on Industrial Informatics* 13.5, pp. 2227–2240. ISSN: 15513203. DOI: 10.1109/TII.2017.2695583.
- Cai, Baoping, Xiangdi Kong, Yonghong Liu, Jing Lin, Xiaobing Yuan, Hongqi Xu, and Renjie Ji (2019). "Application of Bayesian Networks in Reliability Evaluation". In: *IEEE Transactions on Industrial Informatics* 15.4, pp. 2146–2157. ISSN: 15513203. DOI: 10.1109/TII.2018.2858281.

- Cai, Baoping, Hanlin Liu, and Min Xie (2016a). "A real-time fault diagnosis methodology of complex systems using object-oriented Bayesian networks".
 In: Mechanical Systems and Signal Processing 80, pp. 31-44. ISSN: 10961216.
 DOI: 10.1016/j.ymssp.2016.04.019. URL: http://dx.doi.org/10.1016/j.ymssp.2016.04.019.
- (Dec. 2016b). "A real-time fault diagnosis methodology of complex systems using object-oriented Bayesian networks". In: *Mechanical Systems and Signal Processing* 80, pp. 31–44. ISSN: 10961216. DOI: 10.1016/j.ymssp.2016.04.
 019.
- Campora, Ugo, Carlo Cravero, and Raphael Zaccone (2018). "Marine gas turbine monitoring and diagnostics by simulation and pattern recognition". In: International Journal of Naval Architecture and Ocean Engineering 10.5, pp. 617– 628. ISSN: 20926790. DOI: 10.1016/j.ijnaoe.2017.09.012. URL: https: //doi.org/10.1016/j.ijnaoe.2017.09.012.
- Caruana, Rich and Alexandru Niculescu-Mizil (2006). "An empirical comparison of supervised learning algorithms". In: ACM International Conference Proceeding Series 148, pp. 161–168. DOI: 10.1145/1143844.1143865.
- CDNSWC (2010). Handbook of Reliability Prediction Procedures for Mechanical Equipment. Ed. by Naval Surface Warfare Center Carderock Division. West Bethesda: Naval Surface Warfare Center Carderock Division. ISBN: 3012274383.
- Çelik, Mete, Filiz Dadaşer-Çelik, and Ahmet Şakir Dokuz (2011). "Anomaly detection in temperature data using DBSCAN algorithm". In: *INISTA 2011*2011 International Symposium on INnovations in Intelligent SysTems and Applications, pp. 91–95. DOI: 10.1109/INISTA.2011.5946052.
- Cem Kuzu, Ali, Emre Akyuz, and Ozcan Arslan (May 2019). "Application of Fuzzy Fault Tree Analysis (FFTA) to maritime industry: A risk analysing of ship mooring operation". In: Ocean Engineering 179, pp. 128–134. ISSN:

00298018. DOI: 10.1016/j.oceaneng.2019.03.029. URL: https://www. sciencedirect.com/science/article/pii/S0029801818308254.

- Cevasco, D., M. Collu, and Z. Lin (2018). "O&M Cost-Based FMECA: Identification and Ranking of the Most Critical Components for 2-4 MW Geared Offshore Wind Turbines". In: *Journal of Physics: Conference Series* 1102.1. ISSN: 17426596. DOI: 10.1088/1742-6596/1102/1/012039.
- Chan, F. T.S., H. C.W. Lau, R. W.L. Ip, H. K. Chan, and S. Kong (Jan. 2005).
 "Implementation of total productive maintenance: A case study". In: *International Journal of Production Economics* 95.1, pp. 71–94. ISSN: 09255273.
 DOI: 10.1016/j.ijpe.2003.10.021.
- Cheliotis, M. and I. Lazakis (2018). "Ship machinery fuzzy condition based maintenance". In: Smart Ship Technology 2018. Ed. by RINA. London: Royal Institution of Naval Architects. ISBN: 9781909024632.
- Cheliotis, Michail, Christos Gkerekos, Iraklis Lazakis, and Gerasimos Theotokatos (Sept. 2019). "A novel data condition and performance hybrid imputation method for energy efficient operations of marine systems". In: Ocean Engineering 188. ISSN: 00298018. DOI: 10.1016/j.oceaneng.2019.106220.
- Cheliotis, Michail and Iraklis Lazakis (2015). "Assessing Key Reliability and Criticality Aspects of Ship Systems for Improved Energy Efficiency". In: SCC 2015: Shipping in Changing Climates Conference 2015. Glasgow: University of Strathclyde.
- Cheliotis, Michail, Iraklis Lazakis, and Gerasimos Theotokatos (Nov. 2020). "Machine learning and data-driven fault detection for ship systems operations". In: Ocean Engineering 216, p. 107968. ISSN: 00298018. DOI: 10.1016/j. oceaneng.2020.107968. URL: https://linkinghub.elsevier.com/retrieve/ pii/S0029801820309203.
- Chen, Chong, Xuan Zhou, Zhuo Li, Zhiheng He, Zhengtian Li, and Xiangning Lin (Dec. 2018). "Novel complex network model and its application in identifying

critical components of power grid". In: *Physica A: Statistical Mechanics and its Applications* 512, pp. 316–329. ISSN: 03784371. DOI: 10.1016/j.physa. 2018.08.095.

- Chen, Hua Peng and Mehrdad Bahari Mehrabani (May 2019). "Reliability analysis and optimum maintenance of coastal flood defences using probabilistic deterioration modelling". In: *Reliability Engineering and System Safety* 185, pp. 163–174. ISSN: 09518320. DOI: 10.1016/j.ress.2018.12.021.
- Chen, Zhenguo and Yong Fei Li (Jan. 2011). "Anomaly Detection Based on Enhanced DBScan Algorithm". In: *Proceedia Engineering* 15, pp. 178–182. ISSN: 1877-7058. DOI: 10.1016/J.PROENG.2011.08.036. URL: https://www.sciencedirect.com/science/article/pii/S1877705811015372.
- Chojnacki, E., W. Plumecocq, and L. Audouin (2019). "An expert system based on a Bayesian network for fire safety analysis in nuclear area". In: *Fire Safety Journal* 105.October 2018, pp. 28–40. ISSN: 03797112. DOI: 10.1016/j. firesaf.2019.02.007. URL: https://doi.org/10.1016/j.firesaf. 2019.02.007.
- Choudhary, Khilawan and P. Sidharthan (2016). "Failure mode effects and criticality analysis (FMECA) of Electronic Power Conditioner (EPC)". In: 2016 5th International Conference on Reliability, Infocom Technologies and Optimization, ICRITO 2016: Trends and Future Directions, pp. 343–346. DOI: 10.1109/ICRITO.2016.7784978.
- Cipollini, Francesca, Luca Oneto, Andrea Coraddu, Alan John Murphy, and Davide Anguita (2018). "Condition-based maintenance of naval propulsion systems: Data analysis with minimal feedback". In: *Reliability Engineering and Systems Safety* 177, pp. 12–23. ISSN: 09518320. DOI: 10.1016/j.ress.2018.04.015.
- Claramunt, C., C. Ray, L. Salmon, E. Camossi, M. Hadzagic, A.-L. Jousselme, G. Andrienko, N. Andrienko, Y. Theodoridis, and G.A. Vouros (2017). "Maritime

data integration and analysis: Recent progress and research challenges". In: *Advances in Database Technology - EDBT* 2017-March, pp. 192–197. ISSN: 23672005. DOI: 10.5441/002/edbt.2017.18.

- Cocquempot, Vincent and Roozbeh Izadi-zamanabadi (2006). "Pump Application". In: 14.2, pp. 204–215.
- Coraddu, Andrea, Luca Oneto, Aessandro Ghio, Stefano Savio, Davide Anguita, and Massimo Figari (2016). "Machine learning approaches for improving conditionbased maintenance of naval propulsion plants". In: Proceedings of the Institution of Mechanical Engineers Part M: Journal of Engineering for the Maritime Environment 230.1, pp. 136–153. ISSN: 20413084. DOI: 10.1177/ 1475090214540874. URL: https://journals.sagepub.com/doi/pdf/10. 1177/1475090214540874.
- Davarifar, M., A. Rabhi, A. El-Hajjaji, and M. Dahmane (2013). "Real-time model base fault diagnosis of PV panels using statistical signal processing". In: *Proceedings of 2013 International Conference on Renewable Energy Research* and Applications, ICRERA 2013. IEEE Computer Society, pp. 599–604. DOI: 10.1109/ICRERA.2013.6749826.
- De Faria, Haroldo, João Gabriel Spir Costa, and Jose Luis Mejia Olivas (2015).
 A review of monitoring methods for predictive maintenance of electric power transformers based on dissolved gas analysis. DOI: 10.1016/j.rser.2015.
 02.052.
- Deepak Prabhakar, P. and V.P. Jagathy Raj (2014). "CBM, TPM, RCM and A-RCM A Qualitative Comparison of Maintenance Management Strategies".
 In: International Journal of Management & Business Studies 4.3, pp. 49–56.
- Dehghanian, Payman and Mahmud Fotuhi-Firuzabad (2012). "A reliability-oriented outlook on the critical components of power distribution systems". In: IET Conference Publications 2012.616 CP. DOI: 10.1049/cp.2012.2159.

- Dehghanian, Payman, Mahmud Fotuhi-Firuzabad, Saeed Bagheri-Shouraki, and Ali Asghara Razi Kazemi (2012). "Critical component identification in reliability centered asset management of power distribution systems via fuzzy AHP". In: *IEEE Systems Journal* 6.4, pp. 593–602. ISSN: 19328184. DOI: 10.1109/JSYST.2011.2177134.
- Diakaki, C., N. Panagiotidou, A. Pouliezos, G.D. Kontes, G.S. Stavrakakis, K. Belibassakis, T.P. Gerostathis, G. Livanos, N. Pagonis D., and G. Theotokatos (2015). "A decision support system for the development of voyage and maintenance plans for ships". In: *International Journal of Decision Support Systems* 1, pp. 42–71. DOI: 10.1504/ijdss.2015.067274.
- Dikis, F., I. Lazakis, and G. Theotokatos (2015). "Dynamic reliability analysis tool for ship machinery maintenance". In: *Towards Green Marine Technology* and *Transport*, pp. 619–626. DOI: 10.1201/b18855.
- Dikis, Konstantinos (2017). "Establishment of a novel predictive reliability assessment strategy for ship machinery". PhD thesis. Glasgow: University of Strathclyde, p. 374. URL: http://digitool.lib.strath.ac.uk/webclient/ StreamGate?folder_id=0&dvs=1542893816921~871&usePid1=true& usePid2=true.
- Dikis, Konstantinos and Iraklis Lazakis (Mar. 2019). "Dynamic predictive reliability assessment of ship systems". In: International Journal of Naval Architecture and Ocean Engineering. ISSN: 20926782. DOI: 10.1016/j.ijnaoe. 2019.01.002. URL: https://www.sciencedirect.com/science/article/pii/S2092678217302996?via%3Dihub.
- Dikis, Konstantinos, Iraklis Lazakis, and Osman Turan (2014a). "Probabilistic Risk Assessment of Condition Monitoring of Marine Diesel Engines". In: International Conference on Maritime Technology 2014, 7-9 July 2014, Glasgow, United Kingdom July, pp. 1–9.

- Dikis, Konstantinos, Iraklis Lazakis, and Osman Turan (2014b). "Probabilistic Risk Assessment of Condition Monitoring of Marine Diesel Engines". In: International Conference on Maritime Technology 2014, 7-9 July 2014, Glasgow, United Kingdom July, pp. 1–9.
- Dinmohammadi, F and M Shafiee (2013). "A Fuzzy-FMEA Risk Assessment Approach for Offshore Wind Turbines". In: International Journal of Prognostics and Health Management 4.4, pp. 1–10. ISSN: 10990801. DOI: 10.1002/bmc. 3797.
- Dobrkovic, Andrej, Maria-Eugenia Iacob, and Jos van Hillegersberg (2018). "Maritime pattern extraction and route reconstruction from incomplete AIS data".
 In: International Journal of Data Science and Analytics 5.2-3, pp. 111–136.
 ISSN: 2364-415X. DOI: 10.1007/s41060-017-0092-8. URL: http://link.
 springer.com/10.1007/s41060-017-0092-8.
- Domingos, Pedro (2012). "A few useful things to know about machine learning". In: *Communications of the ACM* 55.10, p. 78. ISSN: 00010782. DOI: 10.1145/2347736.2347755. URL: http://dl.acm.org/citation.cfm? doid=2347736.2347755.
- Donders, A. Rogier T., Geert J.M.G. van der Heijden, Theo Stijnen, and Karel G.M. Moons (2006). "Review: A gentle introduction to imputation of missing values". In: *Journal of Clinical Epidemiology* 59.10, pp. 1087–1091. ISSN: 08954356. DOI: 10.1016/j.jclinepi.2006.01.014.
- Duran, Vanessa, Zigor Uriondo, and Juan Moreno-Gutiérrez (2012). "The impact of marine engine operation and maintenance on emissions". In: *Transportation Research Part D: Transport and Environment* 17.1, pp. 54–60. ISSN: 13619209.
 DOI: 10.1016/j.trd.2011.09.001.
- EMSA (2017). Annual Overview of Marine Casualties and Incidents 2017. Tech. rep. Lisbon: EMSA. URL: http://emsa.europa.eu/emsa-documents/

latest/item/3156-annual-overview-of-marine-casualties-andincidents-2017.html.

- EMSA (2018). Annual Overview of Marine Casualties and Incidents 2018. Tech. rep. Lisbon: EMSA, p. 175. URL: http://www.emsa.europa.eu/publications/ technical-reports-studies-and-plans/download/5425/3406/23.html.
- Enders, C. K. (2001). "The Performance of the Full Information Maximum Likelihood Estimator in Multiple Regression Models with Missing Data". In: *Educational and Psychological Measurement* 61.5, pp. 713–740. ISSN: 0013-1644. DOI: 10.1177/0013164401615001.
- Erozan, Ihsan (July 2019). "A fuzzy decision support system for managing maintenance activities of critical components in manufacturing systems". In: Journal of Manufacturing Systems 52, pp. 110–120. ISSN: 02786125. DOI: 10.1016/ j.jmsy.2019.06.002.
- Ersdal, Gerhard and Arne Kvitrud (2000). "Green water on Norwegian production ships". In: Proceedings of the International Offshore and Polar Engineering Conference 4, pp. 211–218.
- Erto, Pasquale, Antonio Lepore, Biagio Palumbo, and Luigi Vitiello (2015). "A Procedure for Predicting and Controlling the Ship Fuel Consumption: Its Implementation and Test". In: *Quality and Reliability Engineering International* 31.7, pp. 1177–1184. ISSN: 10991638. DOI: 10.1002/qre.1864.
- Eruguz, Ayse Sena, Tarkan Tan, and Geert Jan van Houtum (Sept. 2017). A survey of maintenance and service logistics management: Classification and research agenda from a maritime sector perspective. DOI: 10.1016/j.cor. 2017.03.003.
- Ester, Martin, Hans-Peter Kriegel, Jiirg Sander, and Xiaowei Xu (1996). A Density-Based Algorithm for Discovering Clusters in Large Spatial Databases with Noise. Tech. rep. URL: www.aaai.org.

- Fan, Cheng, Fu Xiao, Zhengdao Li, and Jiayuan Wang (Jan. 2018). Unsupervised data analytics in mining big building operational data for energy efficiency enhancement: A review. DOI: 10.1016/j.enbuild.2017.11.008.
- Fang, Liu and Huang Zhaodong (2015). "System Dynamics Based Simulation Approach on Corrective Maintenance Cost of Aviation Equipments". In: *Procedia Engineering*. Vol. 99. Elsevier Ltd, pp. 150–155. DOI: 10.1016/j.proeng. 2014.12.519.
- Fearnleys Research (2019a). Fearnleys Weekly Report Week 48 2019. Tech. rep. London: Astrup Fearnley, p. 7.
- (2019b). *Fearnleys Weekly Report Week 9 2019*. Tech. rep. London: Astrup Fearnley, p. 3.
- (2019c). Fearnleys Weekly Week 41 2019. Tech. rep. London: Astrup Fearnley,
 p. 7.
- Fedele, Lorenzo (2011). Methodologies and Techniques for Advanced Maintenance.
 Ed. by Lorenzo Fedele. 1st. Vol. 1. 4. London: Springer, p. 228. ISBN: 9780857291028.
 DOI: 10.1007/978-0-85729-103-5.
- Fog, T. L., L. K. Hansen, J. Larsen, H. S. Hansen, L. B. Madsen, P. Sorensen,
 E. R. Hansen, and P. S. Pedersen (1999). "On condition monitoring of exhaust valves in marine diesel engines". In: *Neural Networks for Signal Processing -Proceedings of the IEEE Workshop*. IEEE, pp. 554–564. ISBN: 0-7803-5673-X.
 DOI: 10.1109/nnsp.1999.788175. URL: http://ieeexplore.ieee.org/
 document/788175/.
- Ford, B. L. (1983). "An overview of hot-deck procedures." In: Incomplete Data in Sample Surveys. Ed. by W.G Madow, I Olkin, and DB Rubin. Vol. 2. 1st. New York Academic Press, pp. 185–207.
- Fruth, Markus and Frank Teuteberg (2017). "Digitization in maritime logistics—What is there and what is missing?" In: Cogent Business and Manage-

ment 4.1, pp. 1–40. ISSN: 23311975. DOI: 10.1080/23311975.2017.1411066. URL: http://doi.org/10.1080/23311975.2017.1411066.

- Fu, Shanshan and Di Zhang (2016). "A Fuzzy Event Tree Model for Acciden Scenario Analysis of Ship Stuck in Ice in Arctic Waters". In: International Conference on Ocean, Offshore and Arctic Engineering. Busan: ASME, pp. 1– 8.
- Fussell, J B (1975). A Review of Fault Tree Analysis with Emphasis on Limitations. Idaho.
- Galar Pascual, Diego (2015). Artificial Intelligence Tools: Decision Support Systems in Condition Monitoring and Diagnosis. 1st. Boca Raton: CRC Press,
 p. 136. ISBN: 9781466584068. DOI: 10.1201/b18384.
- Gandhare, Balasaheb S and Milind Akarte (2012). "Maintenance strategy selection". In: Proc. of Ninth AIMS Int. Conf. on Management April, pp. 1330– 1336.
- Gaonkar, MN and K Sawant (2013). "Auto Eps DBSCAN: DBSCAN with Eps automatic for large dataset". In: International Journal on Advanced Computer Theory and Engineering 2.2, pp. 11–16.
- García Márquez, Fausto Pedro, Andrew Mark Tobias, Jesús María Pinar Pérez, and Mayorkinos Papaelias (2012). "Condition monitoring of wind turbines: Techniques and methods". In: *Renewable Energy* 46, pp. 169–178. ISSN: 09601481. DOI: 10.1016/j.renene.2012.03.003. URL: http://dx.doi.org/10.1016/j.renene.2012.03.003.
- Garg, Amik and S. G. Deshmukh (2006). "Maintenance management: Literature review and directions". In: Journal of Quality in Maintenance Engineering 12.3, pp. 205–238. ISSN: 13552511. DOI: 10.1108/13552510610685075.
- Garoudja, Elyes, Fouzi Harrou, Ying Sun, Kamel Kara, Aissa Chouder, and Santiago Silvestre (2017). "Statistical fault detection in photovoltaic systems". In:

Solar Energy 150, pp. 485-499. ISSN: 0038092X. DOI: 10.1016/j.solener. 2017.04.043. URL: http://dx.doi.org/10.1016/j.solener.2017.04.043.

- Gibert, K. (Jan. 2014). "Mixed intelligent-multivariate missing imputation". In: International Journal of Computer Mathematics 91.1, pp. 85–96. ISSN: 0020-7160. DOI: 10.1080/00207160.2013.783209. URL: http://www.tandfonline. com/doi/abs/10.1080/00207160.2013.783209.
- Giorgi, Maria Grazia De, Stefano Campilongo, and Antonio Ficarella (2018). "A diagnostics tool for aero-engines health monitoring using machine learning technique". In: *Energy Procedia* 148.Ati, pp. 860–867. ISSN: 18766102. DOI: 10.1016/j.egypro.2018.08.109. URL: https://doi.org/10.1016/j.egypro.2018.08.109.
- Giorgio, Massimiliano, Maurizio Guida, and Gianpaolo Pulcini (2015). "A conditionbased maintenance policy for deteriorating units. An application to the cylinder liners of marine engine". In: Applied Stochastic Models in Business and Industry 31.3, pp. 339–348. ISSN: 15264025. DOI: 10.1002/asmb.2029.
- Goble, William M, Julia Bukowski, and Stewart Loren (2016). Comparing FMEDA Predicted Failure Rates to OREDA Estimated Failure Rates for Sensor and Valve Assemblies. Sellersville.
- Goossens, Adriaan J.M. and Rob J.I. Basten (2015). "Exploring maintenance policy selection using the Analytic Hierarchy Process; An application for naval ships". In: *Reliability Engineering and System Safety* 142, pp. 31–41. ISSN: 09518320. DOI: 10.1016/j.ress.2015.04.014. URL: http://dx.doi.org/ 10.1016/j.ress.2015.04.014.
- Groenen, Patrick J F and Krzysztof Jajuga (2001). Fuzzy clustering with squared Minkowski distances. Tech. rep., pp. 227–237. URL: www.elsevier.com/ locate/fss.
- Guan, Cong, Gerasimos Theotokatos, and Hui Chen (2015). "Analysis of two stroke marine diesel engine operation including turbocharger cut-out by using

a zero-dimensional model". In: *Energies* 8.6, pp. 5738–5764. ISSN: 19961073. DOI: 10.3390/en8065738.

- Guo, Fumin, Xuemei Ren, Zhijun Li, and Cunwu Han (May 2018). "Fault detection and fault-tolerant control over signal-to-noise ratio constrained channels".
 In: International Journal of Robust and Nonlinear Control 28.7, pp. 2806–2827. ISSN: 10498923. DOI: 10.1002/rnc.4050. URL: http://doi.wiley.com/10.1002/rnc.4050.
- Gupta, G. and R.P. Mishra (2018). "Identification of Critical Components Using ANP for Implementation of Reliability Centered Maintenance". In: *Procedia CIRP* 69, pp. 905–909. ISSN: 22128271. DOI: 10.1016/j.procir. 2017.11.122. URL: https://linkinghub.elsevier.com/retrieve/pii/ S2212827117309071.
- Haasl, D F, N H Roberts, W E Vesely, and F F Goldberg (1981). Fault Tree Handbook. Tech. rep. Washington DC: Office of Nuclear Regulatory Research.
- Han, Jiawei, Micheline Kamber, and Jian Pei (2012). Data Mining: Concepts and Techniques. 3rd. Waltham: Morgan Kaufmann, p. 745. ISBN: 978-0-12-381479-1. DOI: 10.1016/B978-0-12-381479-1.00001-0. URL: http: //myweb.sabanciuniv.edu/rdehkharghani/files/2016/02/The-Morgan-Kaufmann-Series-in-Data-Management-Systems-Jiawei-Han-Micheline-Kamber - Jian - Pei - Data - Mining. - Concepts - and - Techniques - 3rd -Edition-Morgan-Kaufmann-2011.pdf%OAhttp://scholar.google.com/ schol.
- Harrou, Fouzi, Mohamed N Nounou, Hazem N Nounou, and Muddu Madakyaru (2015). "PLS-based EWMA fault detection strategy for process monitoring". In: Journal of Loss Prevention in the Process Industries 36, pp. 108–119. ISSN: 09504230. DOI: 10.1016/j.jlp.2015.05.017. URL: http://dx.doi.org/10.1016/j.jlp.2015.05.017.

- Hastie, Trevor, Robert Tibshirani, and Jerome Friedman (2006). The Elements of Statistical Learning. 2nd. New York: Springer, p. 764. DOI: 10.1007/978-1-4419-9863-7{_}941.
- Hegde, Jeevith and Børge Rokseth (Feb. 2020). Applications of machine learning methods for engineering risk assessment – A review. DOI: 10.1016/j.ssci. 2019.09.015.
- Henriques de Gusmão, Ana Paula, Maisa Mendonça Silva, Thiago Poleto, Lúcio Camara e Silva, and Ana Paula Cabral Seixas Costa (2018a). "Cybersecurity risk analysis model using fault tree analysis and fuzzy decision theory". In: *International Journal of Information Management* 43.January, pp. 248–260. ISSN: 02684012. DOI: 10.1016/j.ijinfomgt.2018.08.008. URL: https://doi.org/10.1016/j.ijinfomgt.2018.08.008.
- (2018b). "Cybersecurity risk analysis model using fault tree analysis and fuzzy decision theory". In: International Journal of Information Management 43.January, pp. 248–260. ISSN: 02684012. DOI: 10.1016/j.ijinfomgt.2018.
 08.008. URL: https://doi.org/10.1016/j.ijinfomgt.2018.08.008.
- Hilber, Patrik, Vladimiro Miranda, Manuel A. Matos, and Lina Bertling (2007).
 "Multiobjective optimization applied to maintenance policy for electrical networks". In: *IEEE Transactions on Power Systems* 22.4, pp. 1675–1682. ISSN: 08858950. DOI: 10.1109/TPWRS.2007.908468.
- Hodge, Victoria and Austin Jim (2004). "A Survey of Outlier Detection Methodologies". In: Artificial Intelligence Review 22: 22.85, pp. 85–126. DOI: 10. 4324/9781315744988.
- Holmes, Donald S. and A. Erhan Mergen (2000). "Exponentially weighted moving average acceptance charts". In: Quality and Reliability Engineering International 16.2, pp. 139–142. ISSN: 07488017. DOI: 10.1002/(SICI)1099– 1638(200003/04)16:2<139::AID-QRE312>3.0.CO;2-C.

- Homik, Wojciech (2010). "Diagnostics, maintenance and regeneration of torsional vibration dampers for crankshafts of ship diesel engines". In: *Polish Maritime Research* 17.1, pp. 62–68. ISSN: 12332585. DOI: 10.2478/v10012-010-0007-2.
- Hong, Y.S., Y.M. Cho, Z Hameed, C.K. Song, and S.H. Ahn (2007). "Condition monitoring and fault detection of wind turbines and related algorithms: A review". In: *Renewable and Sustainable Energy Reviews* 13.1, pp. 1–39. ISSN: 13640321. DOI: 10.1016/j.rser.2007.05.008. URL: www.elsevier.com/ locate/rser.
- Horný, Michal (2014). "Bayesian networks: A Technical report". In: Communications of the ACM 53.5, p. 15. ISSN: 00010782. DOI: 10.1145/1859204. 1859227. URL: http://www.bu.edu/sph/files/2014/05/bayesiannetworks-final.pdf%OAhttp://portal.acm.org/citation.cfm?doid= 1859204.1859227.
- Hoseyni, Seyed Mojtaba, Francesco Di Maio, and Enrico Zio (Nov. 2019). "Conditionbased probabilistic safety assessment for maintenance decision making regarding a nuclear power plant steam generator undergoing multiple degradation mechanisms". In: *Reliability Engineering and System Safety* 191. ISSN: 09518320. DOI: 10.1016/j.ress.2019.106583.
- Hosseini, Seyedmohsen and Kash Barker (2016). "Modeling infrastructure resilience using Bayesian networks: A case study of inland waterway ports". In: *Computers and Industrial Engineering* 93, pp. 252–266. ISSN: 03608352. DOI: 10.1016/j.cie.2016.01.007. URL: http://dx.doi.org/10.1016/j.cie. 2016.01.007.
- Hountalas, Dimitrios T. (2000). "Prediction of marine diesel engine performance under fault conditions". In: Applied Thermal Engineering 20.18, pp. 1753– 1783. ISSN: 13594311. DOI: 10.1016/S1359-4311(00)00006-5.
- Hountalas, Dimitrios T, Nikolaos F Sakellaridis, Efthimios Pariotis, Antonis K Antonopoulos, Leonidas Zissimatos, and Nikiforos Papadakis (2014). "Effect

of turbocharger cut out on two-stroke marine diesel engine performance and NOx emissions at part load operation". In:

- Huang, Jianglin, Jacky Wai Keung, Federica Sarro, Yan Fu Li, Y. T. Yu, W. K. Chan, and Hongyi Sun (2017). "Cross-validation based K nearest neighbor imputation for software quality datasets: An empirical study". In: *Journal of Systems and Software* 132, pp. 226–252. ISSN: 01641212. DOI: 10.1016/j. jss.2017.07.012.
- IACS (2015). Classification societies what, why and how? Tech. rep., pp. 1-16. DOI: 10.1007/BF03195078. URL: http://www.iacs.org.uk/media/3785/ iacs-class-what-why-how.pdf.
- (2017). "Requirements concerning Survey and Certification". In: International Association of Classification Societies. URL: http://www.iacs.org.uk/ document/public/Publications/Unified_requirements/PDF/UR_Z_ pdf1890.pdf.
- Ijaz, Muhammad Fazal, Ganjar Alfian, Muhammad Syafrudin, and Jongtae Rhee (2018). "Hybrid Prediction Model for type 2 diabetes and hypertension using DBSCAN-based outlier detection, Synthetic Minority Over Sampling Technique (SMOTE), and random forest". In: Applied Sciences (Switzerland) 8.8. ISSN: 20763417. DOI: 10.3390/app8081325.
- IMO (2013). IMO: What it is. London.
- (2015). International Management Code for the Safe Operation of Ships and for Pollution Prevention (International Safety Management (ISM) Code).
 Tech. rep. London, London:IMO.
- International Institute of Forecasters (2020). M3 Competition Data. Medford. URL: https://forecasters.org/resources/time-series-data/m3competition/.

- Iphar, C, A Napoli, and C Ray (2015). "Data Quality Assessment for Maritime Situation Awareness". In: Isprs Geospatial Week 2015 II-3.W5, pp. 291–296.
 ISSN: 2194-9034. DOI: 10.5194/isprsannals-II-3-W5-291-2015.
- Irvine, University of California (2020). UC Irvine Machine Learning Repository. Irvine. URL: http://archive.ics.uci.edu/ml/index.php.
- Isermann, Rolf (2006). Fault-Diagnosis Systems. An Introduction from Fault Detection to Fault Tolerance. Ed. by Rolf Isermann. Darmstadt: Springer, p. 478. ISBN: 978-3-540-24112-6. URL: https://link.springer.com/content/pdf/ 10.1007%2F3-540-30368-5.pdf.
- ISO (2008). ISO 3046-1:2002 Reciprocating internal combustion engines: Performance: Part 1: Declarations of power, fuel and lubricating oil consumptions, and test methods: Additional requirements for engines for general use. URL: https://www.iso.org/standard/28330.html.
- IUMI (2018). An analysis of the global marine insurance market 2018. Tech. rep. Hamburg: International Union of Marine Insurance, p. 28.
- Jankovsky, Zachary K., Matthew R. Denman, and Tunc Aldemir (2018). "Dynamic event tree analysis with the SAS4A/SASSYS-1 safety analysis code". In: Annals of Nuclear Energy 115, pp. 55–72. ISSN: 18732100. DOI: 10.1016/ j.anucene.2018.01.001. URL: https://doi.org/10.1016/j.anucene. 2018.01.001.
- Jardine, Andrew K.S., Daming Lin, and Dragan Banjevic (Oct. 2006). A review on machinery diagnostics and prognostics implementing condition-based maintenance. DOI: 10.1016/j.ymssp.2005.09.012.
- Javed, Nauman and Tilman Wolf (2012). "Automated sensor verification using outlier detection in the Internet of things". In: Proceedings - 32nd IEEE International Conference on Distributed Computing Systems Workshops, ICDCSW 2012, pp. 291–296. DOI: 10.1109/ICDCSW.2012.78.

- Jiang, Ge, Hongjie Yuan, Peichang Li, and Peng Li (2018). "A new approach to fuzzy dynamic fault tree analysis using the weakest n-dimensional t-norm arithmetic". In: *Chinese Journal of Aeronautics* 31.7, pp. 1506–1514. ISSN: 10009361. DOI: 10.1016/j.cja.2018.04.014. URL: https://doi.org/10. 1016/j.cja.2018.04.014.
- Joshi, Gunjan and Himanshu Joshi (2014). "FMEA and Alternatives v/s Enhanced Risk Assessment Mechanism". In: International Journal of Computer Applications 93.14, pp. 33–37. DOI: 10.5120/16284-6047.
- Kabir, Sohag (2017). "An overview of fault tree analysis and its application in model based dependability analysis". In: *Expert Systems with Applications* 77, pp. 114–135. ISSN: 09574174. DOI: 10.1016/j.eswa.2017.01.058. URL: http://dx.doi.org/10.1016/j.eswa.2017.01.058.
- Kabir, Sohag, Martin Walker, Yiannis Papadopoulos, Erich Rüde, and Peter Securius (2016). "Fuzzy temporal fault tree analysis of dynamic systems". In: *International Journal of Approximate Reasoning* 77, pp. 20–37. ISSN: 0888613X. DOI: 10.1016/j.ijar.2016.05.006. URL: http://dx.doi.org/10.1016/j.ijar.2016.05.006.
- Kang, Jichuan, Liping Sun, and C. Guedes Soares (2019). "Fault Tree Analysis of floating offshore wind turbines". In: *Renewable Energy* 133, pp. 1455–1467.
 ISSN: 18790682. DOI: 10.1016/j.renene.2018.08.097. URL: https://doi.org/10.1016/j.renene.2018.08.097.
- Karkouch, Aimad, Hajar Mousannif, Hassan Al Moatassime, and Thomas Noel (Sept. 2016). Data quality in internet of things: A state-of-the-art survey. DOI: 10.1016/j.jnca.2016.08.002.
- Kelly, Anthony (2006). Strategic Maintenance Planning. 1st. Oxford: Butterworth-Heinemann, p. 304. ISBN: 9780080478999.

- Khare, Vikas, Savita Nema, and Prashant Baredar (2019). "Reliability analysis of hybrid renewable energy system by fault tree analysis". In: *Energy and Envi*ronment 30.3, pp. 542–555. ISSN: 20484070. DOI: 10.1177/0958305X18802765.
- Kim, Kyusung, Charles M Ball, and Emmanuel O Nwadiogbu (2009). Method For Fault Diagnosis of a Turbine Engine.
- King, R D, C Feng, and A Sutherland (1995). "Statlog: comparison of classification algorithms on large real- world problems." In: Applied Artificial Intelligence 9.3, pp. 288–333. DOI: https://doi.org/10.1080/08839519508945477.
- Kirk, Matthew (2017). Thoughtful machine learning. Ed. by Mike Loukides and Shannon Cutt. Sebastopol: O'Reily, p. 205. ISBN: 9781449374068.
- Kobbacy, Khairy A.H (2008). Complex System Maintenance Handbook. 1st. New Jersey: Springer. ISBN: 9781848000100.
- Korb, Kevin B and Ann E Nicholson (2010). Bayesian artificial intelligence, second edition. 2nd. New York: CRC Press. Chap. Second, pp. 1–452. ISBN: 9781439815922. DOI: 10.1201/b10391.
- Korczewski, Zbigniew (2016). "Exhaust gas temperature measurements in diagnostics of turbocharged marine internal combustion engines Part II dynamic measurements". In: *Polish Maritime Research* 23.1, pp. 68–76. ISSN: 20837429. DOI: 10.1515/pomr-2016-0010.
- Kotsiantis, S B, D Kanellopoulos, and P E Pintelas (2006). "Data preprocessing for supervised learning". In: International Journal of Computer Science 1.2, pp. 111-117. ISSN: 1306-4428. DOI: 10.1080/02331931003692557. URL: http: //citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.132.5127& rep=rep1&type=pdf.
- Kuboki, N. and S. Takata (2019). "Selecting the optimum inspection method for preventive maintenance". In: *Proceedia CIRP* 80, pp. 512–517. ISSN: 22128271.
 DOI: 10.1016/j.procir.2019.01.090. URL: https://doi.org/10.1016/j.procir.2019.01.090.

- Kvamstad-lervold, Beate (2017). 9 Paths Into the Maritime Future. Tech. rep. February. Trondheim: SINTEF.
- Lambrou, Maria (2017). "Shipping 4.0: Technology Stack and Digital Innovation Challenges". In: IAME 2017 Conference June, pp. 1–20.
- Lang, Kyle M. and Todd D. Little (2016). "Principled Missing Data Treatments". In: Prevention Science April 2016, pp. 1–11. ISSN: 13894986. DOI: 10.1007/ s11121-016-0644-5.
- Langseth, Helge and Luigi Portinale (2007). "Bayesian networks in reliability". In: Reliability Engineering and System Safety 92.1, pp. 92–108. ISSN: 09518320. DOI: 10.1016/j.ress.2005.11.037.
- Law, Averill M. (2009). "How to build valid and credible simulation models". In: Proceedings - Winter Simulation Conference, pp. 24–33. ISSN: 08917736. DOI: 10.1109/WSC.2009.5429312.
- Lazakis, I., Y. Raptodimos, and T. Varelas (2018a). "Predicting ship machinery system condition through analytical reliability tools and artificial neural networks". In: Ocean Engineering 152, pp. 404–415. ISSN: 00298018. DOI: 10.1016/j.oceaneng.2017.11.017.
- Lazakis, Iraklis (2011). "Establishing an innovative and integrated reliability and criticality based maintenance strategy for the maritime industry by". PhD thesis. Glasgow: University of Strathclyde, p. 428. URL: http://digitool.lib. strath.ac.uk/webclient/StreamGate?folder_id=0&dvs=1542894747511~453& usePid1=true&usePid2=true.
- Lazakis, Iraklis, Christos Gkerekos, and Gerasimos Theotokatos (2018). "Investigating an SVM-driven, one-class approach to estimating ship systems condition". In: Ships and Offshore Structures 14.5, pp. 432–441. ISSN: 17445302.
 DOI: 10.1080/17445302.2018.1500189. URL: https://doi.org/10.1080/17445302.2018.1500189.

- Lazakis, Iraklis and Aykut Ölçer (2015). "Selection of the best maintenance approach in the maritime industry under fuzzy multiple attributive group decision-making environment". In: Proceedings of the Institution of Mechanical Engineers Part M: Journal of Engineering for the Maritime Environment 230.2, pp. 297–309. ISSN: 20413084. DOI: 10.1177/1475090215569819.
- Lazakis, Iraklis, Yannis Raptodimos, and T Varelas (2018b). "Predicting ship machinery system condition through analytical reliability tools and artificial neural networks". In: Ocean Engineering 152.November 2017, pp. 404–415. ISSN: 00298018. DOI: 10.1016/j.oceaneng.2017.11.017. URL: https: //doi.org/10.1016/j.oceaneng.2017.11.017.
- Lazakis, Iraklis, Osman Turan, and Seref Aksu (2010a). "Improving ship maintenance: A criticality and reliability approach". In: 11th International Symposium on Practical Design of Ships and Other Floating Structures, PRADS 2010 2, pp. 1411–1420.
- (Oct. 2010b). "Increasing ship operational reliability through the implementation of a holistic maintenance management strategy". In: *Ships and Offshore Structures* 5.4, pp. 337–357. ISSN: 17445302. DOI: 10.1080/17445302.2010.
 480899. URL: http://www.tandfonline.com/doi/abs/10.1080/17445302.
 2010.480899.
- Lepore, Antonio, Marco Seabra dos Reis, Biagio Palumbo, Ricardo Rendall, and Christian Capezza (2017). "A comparison of advanced regression techniques for predicting ship CO 2 emissions". In: *Quality and Reliability Engineering International* 33.6, pp. 1281–1292. ISSN: 10991638. DOI: 10.1002/qre.2171.
- Li, Zhixiong, Xinping Yan, Chengqing Yuan, Jiangbin Zhao, and Zhongxiao Peng (Mar. 2011). "Fault detection and diagnosis of a gearbox in marine propulsion systems using bispectrum analysis and artificial neural networks". In: *Journal* of Marine Science and Application 10.1, pp. 17–24. ISSN: 16719433. DOI: 10.

1007/s11804-011-1036-7. URL: http://link.springer.com/10.1007/ s11804-011-1036-7.

- Lin, Boliang, Jianping Wu, Ruixi Lin, Jiaxi Wang, Hui Wang, and Xuhui Zhang (2019). "Optimization of high-level preventive maintenance scheduling for high-speed trains". In: *Reliability Engineering and System Safety* 183.November 2018, pp. 261–275. ISSN: 09518320. DOI: 10.1016/j.ress.2018.11.028.
- Lin, Yufei, Maoyin Chen, and Donghua Zhou (2013). "Online probabilistic assessment of operating safety for multi-mode engineering systems based on Bayesian methods". In: 2013 25th Chinese Control and Decision Conference, CCDC 2013 119, pp. 4821–4825. ISSN: 09518320. DOI: 10.1109/CCDC.2013.
 6561807. URL: http://dx.doi.org/10.1016/j.ress.2013.05.018.
- Lindstad, Haakon, Gunnar S. Eskeland, Harilaos Psaraftis, Inge Sandaas, and Anders H. Strømman (Dec. 2015). "Maritime shipping and emissions: A threelayered, damage-based approach". In: Ocean Engineering 110, pp. 94–101. ISSN: 00298018. DOI: 10.1016/j.oceaneng.2015.09.029.
- Little, Roderick and Donald Rubin (2002). Statistical Analysis with Missing Data. Ed. by David J. Balding, Peter Bloomfield, Noel A. C. Cressie, Nicholas I. Fisher, Iain M. Johnstone, J. B. Kadane, Louis M. Ryan, David W. Scott, Adrian F. M. Smith, and Jozef L. Teugels. 2nd. New Jersey: John Wiley & Sons. ISBN: 0471183865.
- Liu, Ruonan, Boyuan Yang, Enrico Zio, and Xuefeng Chen (2018). "Artificial intelligence for fault diagnosis of rotating machinery: A review". In: *Mechanical Systems and Signal Processing* 108, pp. 33–47. ISSN: 10961216. DOI: 10.1016/j.ymssp.2018.02.016. URL: https://doi.org/10.1016/j. ymssp.2018.02.016.
- Liu, Zepeng and Long Zhang (2019). "A review of failure modes, condition monitoring and fault diagnosis methods for large-scale wind turbine bearings". In: *Measurement*, p. 107002. ISSN: 0263-2241. DOI: 10.1016/j.measurement.

2019.107002. URL: https://doi.org/10.1016/j.measurement.2019. 107002.

Lloyd's Register (2007). ShipRight. London.

- Logan, K (2015). Integrating data with CBM-How predictive analytics can ensure propulsion system health. New Jersey.
- Longford, Nicholas T. (2005). Missing Data and Small-Area Estimation. 1st. New York: Springer, pp. -. ISBN: 0387987185.
- Lorencin, Ivan, Nikola Andelic, Vedran Mrzljak, and Zlatan Car (2019). "Multilayer perceptron approach to condition-based maintenance of marine COD-LAG propulsion system components". In: *Pomorstvo* 33.2, pp. 181–190. ISSN: 13320718. DOI: 10.31217/p.33.2.8.
- Louzada, Francisco, Anderson Ara, and Guilherme B. Fernandes (2016). "Classification methods applied to credit scoring: Systematic review and overall comparison". In: Surveys in Operations Research and Management Science 21.2, pp. 117–134. ISSN: 18767354. DOI: 10.1016/j.sorms.2016.10.001. URL: http://dx.doi.org/10.1016/j.sorms.2016.10.001.
- Loyola-Gonzalez, Octavio (2019). "Black-box vs. White-Box: Understanding their advantages and weaknesses from a practical point of view". In: *IEEE Access* 7, pp. 154096–154113. ISSN: 21693536. DOI: 10.1109/ACCESS.2019.2949286.
- Lu, P. J., M. C. Zhang, T. C. Hsu, and J. Zhang (2001). "An evaluation of engine faults diagnostics using artificial neural networks". In: *Journal of Engineering* for Gas Turbines and Power 123.2, pp. 340–346. ISSN: 07424795. DOI: 10. 1115/1.1362667.
- Lughofer, Edwin and Moamar Sayed-Mouchaweh (2019). Predictive maintenance in dynamic systems. Ed. by Edwin Lughofer and Moamar Sayed-Mouchaweh. 1st. London: Springer, pp. 1–23. ISBN: 9783030056452. DOI: 10.1007/978-3-030-05645-2{_}1.

- Ma, Jianping and Jin Jiang (2011). "Applications of fault detection and diagnosis methods in nuclear power plants: A review". In: *Progress in Nuclear Energy* 53.3, pp. 255–266. ISSN: 01491970. DOI: 10.1016/j.pnucene.2010.12.001. URL: http://dx.doi.org/10.1016/j.pnucene.2010.12.001.
- Maimon, Oded and Lior Rokach (n.d.). The Data Mining and Knowladge Discovery Handbook. Ed. by Oded Maimon and Lior Rokach. 1st. Tel-Aviv: Springer. ISBN: 9780387244358.
- MAN B&W (2014). Influence of Ambient Temperature Conditions. Tech. rep. Copenhagen: MAN Diesel and Turbo, p. 20.
- (20017). Operational and Maintenance Manual. Augsburg.
- Manzini, Riccardo, Hoang Pham, Alberto Regattieri, and Emilio Ferrari (2010). Maintenance for Industrial Systems, pp. 189–204. ISBN: 9781848825741. DOI: 10.1007/978-1-84882-575-8.
- Mares, Radu Emil, Mariana Nagy, and Ioan Radu (2020). "Using the event tree analysis in the investigation of a work accident". In: MATEC Web of Conferences 305, p. 00088. DOI: 10.1051/matecconf/202030500088.
- Martinez-Guerra, Rafael and Juan Luis Mata-Machuca (2013). Understanding Complex Systems Fault Detection and Diagnosis in Nonlinear Systems A Differential and Algebraic Viewpoint. Ed. by Rafael Martinez-Guerra and Juan Luis Mata-Machuca. London: Springer, p. 143. ISBN: 978-3-319-03046-3. DOI: 10.1007/978-3-319-03047-0. URL: http://www.springer.com/series/ 5394.
- Martinez-Luengo, Maria, Mahmood Shafiee, and Athanasios Kolios (Feb. 2019). "Data management for structural integrity assessment of offshore wind turbine support structures: data cleansing and missing data imputation". In: Ocean Engineering 173, pp. 867–883. ISSN: 00298018. DOI: 10.1016/j.oceaneng. 2019.01.003. URL: https://www.sciencedirect.com/science/article/ pii/S0029801818308576#pdfjs.action=download.

- McKee, K K, G L Forbes, I Mazhar, R Entwistle, and I Howard (2014). "Engineering Asset Management 2011: Proceedings of the sixth world congress on engineering asset management". In: ed. by Jay Lee, Jun Ni, Jagnathan Sarangapani, and Joseph Mathew. 1st. London: Springer. Chap. A Review o, pp. 259–272. ISBN: 9781447149927. DOI: 10.1007/978-1-4471-4993-4.
- McKnight, Patrick, Katherine McKnight, Souray Sidani, and Aurelio Figueredo (2007). Missing Data: A Gentle Introduction. New York: Guildford Press.
- Miller, Clayton, Zoltán Nagy, and Arno Schlueter (Jan. 2018). A review of unsupervised statistical learning and visual analytics techniques applied to performance analysis of non-residential buildings. DOI: 10.1016/j.rser.2017.05. 124.
- Mittal, Ankush and Ashraf Kassim (2007). Bayesian Network Technologies: Applications and Graphical Models. 1st. New York: IGI Publishing, p. 368. ISBN: 9781599041414. DOI: 10.4018/978-1-59904-141-4.
- Mobley, K, L Higgins, and D Wikoff (2008). Maintenance Engineering Handbook. Ed. by R K Mobley. 7th. McGraw Hill, p. 1244. ISBN: 0071641017. DOI: 10. 1036/0071546464.
- Mohammed, J. Zaki and Meira Jr. Wagner (2014). Data Mining and Analysis: Fundamental Concepts and Algorithms. Ed. by J. Zaki Mohammed. 1st. New York: Cambridge University Press, p. 562. ISBN: 0219-1377. DOI: 10.1145/ 3054925. URL: http://www.worldcat.org/oclc/890721171%5Cnhttps: //www.chapters.indigo.ca/en-ca/books/data-classificationalgorithms-and-applications/9781466586758-item.html%0Ahttp: //dx.doi.org/10.1007/s10115-014-0808-1%0Ahttp://dl.acm.org/ citation.cfm?doid=3071073.305492.
- Mohanty, Amiya R. (2015). *Machinery Condition Monitoring*. 1st. Taylor & Francis. ISBN: 9781466593053.

- Mokashi, A. J., J. Wang, and A. K. Vermar (Sept. 2002). "A study of reliability-centred maintenance in maritime operations". In: *Marine Policy* 26.5, pp. 325–335. ISSN: 0308597X. DOI: 10.1016/S0308-597X(02)00014-3.
- Moubray, John (1991). Reliability-Centred Maintenance II. 2nd. Oxford: Butterworth-Heinemann, p. 452. ISBN: 9780750633581.
- Mrad, Ali Ben, Veronique Delcroix, Mohamed Amine Maalej, Sylvain Piechowiak, and Mohamed Abid (2012). "Communications in Computer and Information Science". In: Advances in Computational Intelligence. Ed. by Salvatore Greco, Bernadette Bouchon-Meunier, Giulianella Coletti, Mario Fedrizzi, Benedetto Matarazzo, and Ronald Yager. Catania: Springer. Chap. Uncertain, p. 626. ISBN: 978-3-642-31717-0. DOI: 10.1007/978-3-642-31718-7.
- Mukherjee, Amrit, Pratik Goswami, Lixia Yang, Ziwei Yan, and Mahmoud Daneshmand (2020). "Dynamic clustering method based on power demand and information volume for intelligent and green IoT". In: *Computer Communications* 152.November 2019, pp. 119–125. ISSN: 1873703X. DOI: 10.1016/j.comcom. 2020.01.026. URL: https://doi.org/10.1016/j.comcom.2020.01.026.
- Müller, Andreas C. and Sarah Guido (2015). Introduction to Machine Learning with Python and Scikit-Learn. Ed. by Dawn Schanafelt. 1st. Sebastopol: O'Reily, p. 394. ISBN: 9781449369415. URL: http://kukuruku.co/hub/ python/introduction-to-machine-learning-with-python-andscikitlearn.
- Murphy, A. J., A. J. Norman, K. Pazouki, and D. G. Trodden (2015). "Thermodynamic simulation for the investigation of marine Diesel engines". In: Ocean Engineering 102, pp. 117–128. ISSN: 00298018. DOI: 10.1016/j.oceaneng.2015. 04.004. URL: http://dx.doi.org/10.1016/j.oceaneng.2015.04.004.
- Musman, Ahmadi Hamdan and Rosmaini Ahmad (2018). "Critical component identification and autonomous maintenance activities determination using fuzzy analytical hierarchy process method". In: *International Journal of In*-

dustrial and Systems Engineering 28.3, pp. 360–378. ISSN: 17485045. DOI: 10.1504/IJISE.2018.089745.

- Naik, Jyotirmayee, Ranjeeta Bisoi, and P. K. Dash (Dec. 2018). "Prediction interval forecasting of wind speed and wind power using modes decomposition based low rank multi-kernel ridge regression". In: *Renewable Energy* 129, pp. 357-383. ISSN: 18790682. DOI: 10.1016/j.renene.2018.05. 031. URL: https://www.sciencedirect.com/science/article/pii/ S0960148118305524.
- Narayanagounder, Sellappan and Karuppusami Gurusami (2009). "A new approach for prioritization of failure modes in design FMEA using ANOVA".
 In: World Academy of Science, Engineering and Technology 37, pp. 524–531.
 ISSN: 2010376X.
- Necoara, I., B. De Schutter, T. Van Den Boom, and H. Hellendoorn (2008). "Model predictive control for uncertain max-min-plus-scaling systems". In: *International Journal of Control* 81.5, pp. 701–713. ISSN: 00207179. DOI: 10. 1080/00207170601094404.
- Niu, Gang (2016). Data-driven technology for engineering systems health management. Ed. by Gang Niu. 1st. Beijing: Springer, p. 357. ISBN: 9789811020322.
 DOI: 10.1007/978-981-10-2032-2.
- Nordmann, Rainer and Martin Aenis (2004). "Fault Diagnosis in a Centrifugal Pump Using Active Magnetic Bearings". In: International Journal of Rotating Machinery 10.3, pp. 183–191. ISSN: 1023-621X. DOI: 10.1155/ s1023621x04000193.
- Nounou, Mohamed, Hazem Nounou, Majdi Mansouri, Mohamed Faouzi Harkat, Ayman Al-khazraji, and Mansour Hajji (June 2018). "Wavelet optimized EWMA for fault detection and application to photovoltaic systems". In: Solar Energy 167, pp. 125–136. ISSN: 0038092X. DOI: 10.1016/j.solener.

2018.03.073. URL: https://linkinghub.elsevier.com/retrieve/pii/ S0038092X18303050.

- Nourian, R. and S. Meysam Mousavi (2019). "Design and implementation of an expert system for periodic and emergency control under uncertainty: A case study of city gate stations". In: *Journal of Natural Gas Science and Engineering* 66.September 2018, pp. 306–315. ISSN: 18755100. DOI: 10.1016/ j.jngse.2019.04.007. URL: https://doi.org/10.1016/j.jngse.2019. 04.007.
- Nourian, R., S. Meysam Mousavi, and S. Raissi (2019). "A fuzzy expert system for mitigation of risks and effective control of gas pressure reduction stations with a real application". In: *Journal of Loss Prevention in the Process Industries* 59.September 2018, pp. 77–90. ISSN: 09504230. DOI: 10.1016/j.jlp.2019.03.003. URL: https://doi.org/10.1016/j.jlp.2019.03.003.
- NTSB (2015). Fire Aboard Containership Gunde Maersk. Tech. rep. Washington D.C: National Transportation Safety Board, p. 6. DOI: 10.4135/9781452241067. n23.
- OCIMF (2004). Tanker Management and Self Assessment a Best-Practice Guide for Ship Operators. Tech. rep. London: OCIMF. URL: http://www.sbsp.co. kr/skin/board/sub_tmsa/TMSA.pdf.
- Olive, David J (2005). *Linear Regression*. 1st. Cham: Springer, p. 499. ISBN: 9783319552507. DOI: 10.1007/0-387-27255-0{_}4.
- Oxford Economics Ltd (2017). The economic value of the EU shipping industry. Tech. rep. February. Oxford Economics, p. 24. URL: http://www.ecsa.eu/ images/Studies/150220%20European%20Shipping%20Update.pdf.
- Özcan, Evren Can, Sultan Ünlüsoy, and Tamer Eren (2017). "A combined goal programming – AHP approach supported with TOPSIS for maintenance strategy selection in hydroelectric power plants". In: *Renewable and Sustainable Energy Reviews* 78.February, pp. 1410–1423. ISSN: 18790690. DOI: 10.1016/

j.rser.2017.04.039. URL: http://dx.doi.org/10.1016/j.rser.2017. 04.039.

- Papagiannakis, R. G. and D. T. Hountalas (2004). "Combustion and exhaust emission characteristics of a dual fuel compression ignition engine operated with pilot diesel fuel and natural gas". In: *Energy Conversion and Management* 45.18-19, pp. 2971–2987. ISSN: 01968904. DOI: 10.1016/j.enconman. 2004.01.013.
- Pearl, Judea (1985). Bayesian networks: A model of self-activated memory for evidential reasoning. Tech. rep. Los Angeles: University of California Los Angeles, pp. 1–14.
- (1988). Prpbabilistic Reasoning in Intelligent Systems. Ed. by Michael B Morgan. 2nd. San Francisco: Morgan Kaufmann, p. 576. DOI: 10.1016/b978-0-08-051489-5.50002-3.
- Peeters, J. F.W., R. J.I. Basten, and T. Tinga (2018). "Improving failure analysis efficiency by combining FTA and FMEA in a recursive manner". In: *Reliability Engineering and System Safety* 172.December 2017, pp. 36–44. ISSN: 09518320. DOI: 10.1016/j.ress.2017.11.024. URL: https://doi.org/10. 1016/j.ress.2017.11.024.
- Pigott, Therese D. (2003). "A Review of Methods for Missing Data". In: Educational Research and Evaluation 7.4, pp. 353–383. ISSN: 1380-3611. DOI: 10.1076/edre.7.4.353.8937.
- Polotski, Vladimir, Jean Pierre Kenne, and Ali Gharbi (Dec. 2019). "Optimal production and corrective maintenance in a failure-prone manufacturing system under variable demand". In: *Flexible Services and Manufacturing Journal* 31.4, pp. 894–925. ISSN: 19366590. DOI: 10.1007/s10696-019-09337-8.
- Pordanjani, Iraj Rahimi, Yunfei Wang, and Wilsun Xu (June 2013). "Identification of Critical Components for Voltage Stability Assessment Using Channel Components Transform". In: *IEEE Transactions on Smart Grid* 4.2, pp. 1122–

1132. ISSN: 1949-3053. DOI: 10.1109/TSG.2013.2242498. URL: http://ieeexplore.ieee.org/document/6485019/.

Prajapati, Ashok, James Bechtel, and Subramaniam Ganesan (2012). "Condition based maintenance: A survey". In: Journal of Quality in Maintenance Engineering 18.4, pp. 384–400. ISSN: 13552511. DOI: 10.1108/13552511211281552.

PTC Windchill (2019). Getting Started Guide. Boston.

- Raczko, Edwin and Bogdan Zagajewski (Jan. 2017). "Comparison of support vector machine, random forest and neural network classifiers for tree species classification on airborne hyperspectral APEX images". In: *European Journal of Remote Sensing* 50.1, pp. 144–154. ISSN: 22797254. DOI: 10.1080/ 22797254.2017.1299557. URL: https://www.tandfonline.com/doi/full/ 10.1080/22797254.2017.1299557.
- Radionov, A. A., S. A. Evdokimov, A. A. Sarlybaev, and O. I. Karandaeva (2015). "Application of subtractive clustering for power transformer fault diagnostics". In: *Procedia Engineering* 129, pp. 22–28. ISSN: 18777058. DOI: 10.1016/j.proeng.2015.12.003.
- Rahmah, Nadia and Imas Sukaesih Sitanggang (2016). "Determination of Optimal Epsilon (Eps) Value on DBSCAN Algorithm to Clustering Data on Peatland Hotspots in Sumatra". In: *IOP Conference Series: Earth and Environmental Science*. Vol. 31. 1, p. 12012. DOI: 10.1088/1755-1315/31/1/012012.
- Raiyan, Asif, Subir Das, and M. Rafiqul Islam (2017). "Event tree analysis of marine accidents in Bangladesh". In: *Proceedia Engineering* 194, pp. 276–283.
 ISSN: 18777058. DOI: 10.1016/j.proeng.2017.08.146. URL: http://dx. doi.org/10.1016/j.proeng.2017.08.146.
- Rakopoulos, C. D., K. A. Antonopoulos, D. C. Rakopoulos, D. T. Hountalas, andE. G. Giakoumis (2006). "Comparative performance and emissions study of a direct injection Diesel engine using blends of Diesel fuel with vegetable oils or

bio-diesels of various origins". In: *Energy Conversion and Management* 47.18-19, pp. 3272–3287. ISSN: 01968904. DOI: 10.1016/j.enconman.2006.01.006.

- Ramzali, Nahid, Mohammad Reza Miri Lavasani, and Jamal Ghodousi (2015). "Safety barriers analysis of offshore drilling system by employing Fuzzy event tree analysis". In: Safety Science 78, pp. 49–59. ISSN: 18791042. DOI: 10. 1016/j.ssci.2015.04.004. URL: http://dx.doi.org/10.1016/j.ssci. 2015.04.004.
- Ranachowski, Z. and A. Bejger (2005). "Fault diagnostics of the fuel injection system of a medium power maritime diesel engine with application of acoustic signal". In: Archives of Acoustics 30.4, pp. 465–472. ISSN: 01375075.
- Raptodimos, Yiannis (2018). Combination of Reliability Tools and Artificial Intelligence in a Hybrid Condition Monitoring Framework for Ship Machinery Systems. Glasgow.
- Raptodimos, Yiannis and Iraklis Lazakis (2018). "Using artificial neural networkself-organising map for data clustering of marine engine condition monitoring applications". In: *Ships and Offshore Structures* 13.6, pp. 649–656. ISSN: 17445302. DOI: 10.1080/17445302.2018.1443694. URL: https://doi.org/ 10.1080/17445302.2018.1443694.
- Rausan, Marvin and Arnljot Hoyland (2004). System Reliability Theory: Models, Statistical Methods, and Applications. Ed. by Walter Shewhart and Samuel Wilks. 2nd. Hoboken: Wikey & Sons, p. 672. ISBN: 978-0-471-47133-2.
- Relex Software Corporation (2003). Reliability: A Practitioner's Guide. Tech. rep. Greensburg: Relex Software Corporation, p. 350.
- Renjith, Shini, A. Sreekumar, and M. Jathavedan (2020). "Performance evaluation of clustering algorithms for varying cardinality and dimensionality of data sets". In: *Materials Today: Proceedings*, pp. 1–7. ISSN: 22147853. DOI: 10.1016/j.matpr.2020.01.110. URL: https://doi.org/10.1016/j. matpr.2020.01.110.

Riascos, Luis Alberto M., Marcelo G. Simoes, and Paulo E. Miyagi (Feb. 2007). "A Bayesian network fault diagnostic system for proton exchange membrane fuel cells". In: *Journal of Power Sources* 165.1, pp. 267–278. ISSN: 0378-7753. DOI: 10.1016/J.JPOWSOUR.2006.12.003. URL: https://www.sciencedirect. com/science/article/pii/S0378775306025262.

Rizvi, M. H. (1983). Hot-deck procedures: Introduction.

- Roth, PL L (1994). "Missing data: A conceptual review for applied psychologists". In: Personnel Psychology 47.3, pp. 537-560. ISSN: 1744-6570. DOI: 10.1111/ J.1744-6570.1994.TB01736.X. URL: http://onlinelibrary.wiley.com/ doi/10.1111/j.1744-6570.1994.tb01736.x/abstract.
- Royston, Patrick (2004). "Multiple imputation of missing values". In: The Stata Journal 4.3, pp. 227–241. ISSN: 1536867X.
- Royston, Patrick and Ian White (2015). "Multiple Imputation by Chained Equations (MICE): Implementation in Stata ". In: Journal of Statistical Software 45.4. DOI: 10.18637/jss.v045.i04.
- Rubin, D B (1976). "Inference and missing data (with discussion)". In: *Biometrika* 63.July, pp. 581–592.
- Ruggeri, F, F Faltin, and R Kenett (2007). "Bayesian networks". In: Wiley Interdisciplinary Reviews: Computational Statistics 1.3, pp. 307-315. ISSN: 19395108. DOI: 10.1002/wics.48. URL: http://doi.wiley.com/10.1002/ wics.48.
- Ruijters, Enno and Mariëlle Stoelinga (2015). "Fault tree analysis: A survey of the state-of-the-art in modeling, analysis and tools". In: *Computer Science Review* 15, pp. 29–62. ISSN: 15740137. DOI: 10.1016/j.cosrev.2015.03.001. URL: http://dx.doi.org/10.1016/j.cosrev.2015.03.001.
- Ruilin, Zhang and Ian S. Lowndes (Nov. 2010). "The application of a coupled artificial neural network and fault tree analysis model to predict coal and gas

outbursts". In: International Journal of Coal Geology 84.2, pp. 141–152. ISSN: 01665162. DOI: 10.1016/j.coal.2010.09.004.

- Salim, Siti Hajar, Saiful Amri Mazlan, and Siti Aisyah Salim (2019). "Preliminary Study on Decision Making Factors to Replace Medical Equipment in Hospital". In: Journal of Social Transformation and Regional Development 1.1, pp. 12–17. DOI: 10.30880/jstard.2019.01.01.002.
- Salmasnia, Ali, Maryam Kaveie, and Mohammadreza Namdar (Apr. 2018). "An integrated production and maintenance planning model under VP-T2 Hotelling chart". In: *Computers and Industrial Engineering* 118, pp. 89–103. ISSN: 03608352. DOI: 10.1016/j.cie.2018.02.021.
- Saltelli, Andrea (2004). Sensitivity analysis in practice: a guide to assessing scientific models (Google eBook), p. 219. ISBN: 0470870931. URL: http://books. google.com/books?id=NsAVmohPNpQC&pgis=1.
- Sari, Adel Haghani Abandan (2013). Data-driven design of fault diagnosis systems. Ed. by Adel Haghani Abandan Sari. Vol. 9783658058. Rostock: Springer, pp. 1–136. ISBN: 9783658058074. DOI: 10.1007/978-3-658-05807-4. URL: www.springer-vieweg.de.
- Sarih, Houda, Ayeley Tchangani, Kamal Medjaher, and Eric Pere (Jan. 2018).
 "Critical components identification based on experience feedback data in the framework of PHM". In: *IFAC-PapersOnLine* 51.11, pp. 429–434. ISSN: 24058963.
 DOI: 10.1016/j.ifacol.2018.08.336.
- Saxena, Amit, Mukesh Prasad, Akshansh Gupta, Neha Bharill, Om Prakash Patel, Aruna Tiwari, Meng Joo Er, Weiping Ding, and Chin Teng Lin (2017). "A review of clustering techniques and developments". In: *Neurocomputing* 267, pp. 664–681. ISSN: 18728286. DOI: 10.1016/j.neucom.2017.06.053. URL: http://dx.doi.org/10.1016/j.neucom.2017.06.053.
- Sayed-Mouchaweh, Moamar (2018). Fault diagnosis of hybrid dynamic and complex systems. Ed. by Moamar Sayed-Mouchaweh. Douai: Springer, pp. 1–286.

ISBN: 9783319740140. DOI: 10.1007/978-3-319-74014-0. URL: https: //doi.org/10.1007/978-3-319-74014-0.

- Schlechtingen, Meik and Ilmar Ferreira Santos (2011). "Comparative analysis of neural network and regression based condition monitoring approaches for wind turbine fault detection". In: *Mechanical Systems and Signal Processing* 25.5, pp. 1849–1875. ISSN: 08883270. DOI: 10.1016/j.ymssp.2010.12.007. URL: http://dx.doi.org/10.1016/j.ymssp.2010.12.007.
- Schlechtingen, Meik and Ilmar Ferreira Santos (2014). "Wind turbine condition monitoring based on SCADA data using normal behavior models. Part 2: Application examples". In: Applied Soft Computing Journal 14.PART C, pp. 447– 460. ISSN: 15684946. DOI: 10.1016/j.asoc.2013.09.016. URL: http: //dx.doi.org/10.1016/j.asoc.2013.09.016.
- Schlechtingen, Meik, Ilmar Ferreira Santos, and Sofiane Achiche (2013). "Wind turbine condition monitoring based on SCADA data using normal behavior models. Part 1: System description". In: *Applied Soft Computing Journal* 13.1, pp. 259–270. ISSN: 15684946. DOI: 10.1016/j.asoc.2012.08.033. URL: http://dx.doi.org/10.1016/j.asoc.2012.08.033.
- Schubert, Erich, Jörg Sander, Martin Ester, Hans Peter Kriegel, and Xiaowei Xu (2017). "DBSCAN revisited, revisited: Why and how you should (still) use DBSCAN". In: ACM Transactions on Database Systems 42.3. ISSN: 15574644. DOI: 10.1145/3068335.
- Selim, Shokri Z. and M. A. Ismail (1984). "Soft clustering of multidimensional data: a semi-fuzzy approach". In: *Pattern Recognition* 17.5, pp. 559–568. ISSN: 00313203. DOI: 10.1016/0031-3203(84)90054-2.
- Selvik, J. T. and T. Aven (Feb. 2011). "A framework for reliability and risk centered maintenance". In: *Reliability Engineering and System Safety* 96.2, pp. 324–331. ISSN: 09518320. DOI: 10.1016/j.ress.2010.08.001.

- Shafiee, Mahmood, Evenye Enjema, and Athanasios Kolios (Mar. 2019). "An Integrated FTA-FMEA Model for Risk Analysis of Engineering Systems: A Case Study of Subsea Blowout Preventers". In: Applied Sciences 9.6, p. 1192. ISSN: 2076-3417. DOI: 10.3390/app9061192. URL: https://www.mdpi.com/ 2076-3417/9/6/1192.
- Shafiee, Mahmood, Ashraf Labib, Jhareswar Maiti, and Andrew Starr (2019). "Maintenance strategy selection for multi-component systems using a combined analytic network process and cost-risk criticality model". In: Proceedings of the Institution of Mechanical Engineers, Part O: Journal of Risk and Reliability 233.2, pp. 89–104. ISSN: 17480078. DOI: 10.1177/1748006X17712071.
- Shah, Anoop D., Jonathan W. Bartlett, James Carpenter, Owen Nicholas, and Harry Hemingway (Mar. 2014). "Comparison of random forest and parametric imputation models for imputing missing data using MICE: A CAL-IBER study". In: American Journal of Epidemiology 179.6, pp. 764–774. ISSN: 14766256. DOI: 10.1093/aje/kwt312. URL: https://academic.oup.com/ aje/article-lookup/doi/10.1093/aje/kwt312.
- Sharma, Anil, G. S. Yadava, and S. G. Deshmukh (2011). "A literature review and future perspectives on maintenance optimization". In: *Journal of Quality* in Maintenance Engineering 17.1, pp. 5–25. ISSN: 13552511. DOI: 10.1108/ 13552511111116222.
- Sherwin, David (2000). "A review of overall models for maintenance management". In: Journal of Quality in Maintenance Engineering 6.3, pp. 138–164. ISSN: 13552511. DOI: 10.1108/13552510010341171.
- Shin, Jinsoo, Hanseong Son, Rahman Khalil Ur, and Gyunyoung Heo (2015). "Development of a cyber security risk model using Bayesian networks". In: *Reliability Engineering and System Safety* 134, pp. 208–217. ISSN: 09518320. DOI: 10.1016/j.ress.2014.10.006. URL: http://dx.doi.org/10.1016/j. ress.2014.10.006.

- Shreve, Dennis (2003). Integrated Condition Monitoring Technologies. Auburn. DOI: 10.1016/0963-8695(91)90501-s.
- Silva, Andre A., Shalabh Gupta, Ali M. Bazzi, and Arthur Ulatowski (July 2018). "Wavelet-based information filtering for fault diagnosis of electric drive systems in electric ships". In: *ISA Transactions* 78, pp. 105–115. ISSN: 00190578. DOI: 10.1016/j.isatra.2017.08.013. URL: https://linkinghub. elsevier.com/retrieve/pii/S001905781730544X.
- Singh, Ranteshwar, Ashish M. Gohil, Dhaval B. Shah, and Sanjay Desai (2013).
 "Total productive maintenance (TPM) implementation in a machine shop: A case study". In: *Proceedia Engineering*. Vol. 51. Elsevier Ltd, pp. 592–599.
 ISBN: 9781627486330. DOI: 10.1016/j.proeng.2013.01.084.
- Smith, R and K Mobley (2008). Rules of Thumb for Maintenance and Reliability Engineers. 1st. Oxford: Butterworth-Heinemann, p. 336. ISBN: 978-0-7506-7862-9. DOI: https://doi.org/10.1016/B978-0-7506-7862-9.X5001-1.
- Sodré, J R and S M C Soares (n.d.). Comparison of Engine Power Correction Factors for Varying Atmospheric Conditions. Tech. rep.
- Spreafico, Christian, Davide Russo, and Caterina Rizzi (2017). "A state-of-the-art review of FMEA/FMECA including patents". In: Computer Science Review 25, pp. 19–28. ISSN: 15740137. DOI: 10.1016/j.cosrev.2017.05.002. URL: http://dx.doi.org/10.1016/j.cosrev.2017.05.002.
- Stenström, Christer, Per Norrbin, Aditya Parida, and Uday Kumar (May 2016). "Preventive and corrective maintenance – cost comparison and cost-benefit analysis". In: Structure and Infrastructure Engineering 12.5, pp. 603-617. ISSN: 1573-2479. DOI: 10.1080/15732479.2015.1032983. URL: http:// www.tandfonline.com/doi/full/10.1080/15732479.2015.1032983.
- Stopford, Martin (2018). *Maritime Economics*. 3rd. Oxon: Taylor & Francis. ISBN: 0-203-89174-0.

- Sullivan, Danielle and Rebecca Andridge (2015). "A hot deck imputation procedure for multiply imputing nonignorable missing data: The proxy patternmixture hot deck". In: *Computational Statistics and Data Analysis* 82, pp. 173– 185. ISSN: 01679473. DOI: 10.1016/j.csda.2014.09.008. URL: http: //dx.doi.org/10.1016/j.csda.2014.09.008.
- Sullivan, G. P., R. Pugh, A. P. Melendez, and W. D. Hunt (2010a). Operations & Maintenance Best Practices: A Guide to Achieving Operational Efficiency. Tech. rep. August 2010. Oak Ridge: US Department of Energy, p. 321. DOI: 10.2172/1034595. URL: http://www1.eere.energy.gov/femp/pdfs/omguide_complete.pdf.
- (2010b). Operations & Maintenance Best Practices. Tech. rep. 3. Washington
 DC: US Department of Energy, p. 21.
- Taher, Atabak, Iraklis Lazakis, and Osman Turan (2014). "Integration of Business and Technical Aspects of Reliability and Maintenance". In: July, pp. 1–9.
- Takata, S., F. Kimura, F. J.A.M. Van Houten, E. Westkämper, M. Shpitalni, D. Ceglarek, and J. Lee (2004). "Maintenance: Changing role in life cycle management". In: *CIRP Annals - Manufacturing Technology* 53.2, pp. 643– 655. ISSN: 00078506. DOI: 10.1016/S0007-8506(07)60033-X.
- Tan, Pang-Ning, Michael Steinbach, and Vipin Kumar (2006). Introduction to data mining. 1st. Pearson Education, p. 796. ISBN: 9789332518650. DOI: 10. 1016/0022-4405(81)90007-8.
- Taylor, James W. and Jooyoung Jeon (June 2018). "Probabilistic forecasting of wave height for offshore wind turbine maintenance". In: *European Journal of Operational Research* 267.3, pp. 877–890. ISSN: 03772217. DOI: 10.1016/j.ejor.2017.12.021.
- Taylor, Publisher and Donald B Rubin (1996). "Multiple Imputation after 18 + Years". In: Journal of the American Statistical Association 91.434, pp. 473– 489.
- Ten-Wolde, Mike and Adel A. Ghobbar (2013). "Optimizing inspection intervals
 Reliability and availability in terms of a cost model: A case study on railway carriers". In: *Reliability Engineering and System Safety* 114.1, pp. 137–147. ISSN: 09518320. DOI: 10.1016/j.ress.2012.12.013.
- Thang, Tran Manh and Juntae Kim (2011). "The anomaly detection by using DBSCAN clustering with multiple parameters". In: 2011 International Conference on Information Science and Applications, ICISA 2011, pp. 1–5. DOI: 10.1109/ICISA.2011.5772437.
- The Bahamas Maritime Authority (2013). Carnival Triumph: Joint repot of the investigation into an engine room fire. Tech. rep. London: The Bahamas Maritime Authority, p. 104. URL: https://www.facebook.com/Carnival/posts/ 10151338849524584.
- Theotokatos, Gerasimos, Cong Guan, Hui Chen, and Iraklis Lazakis (2018). "Development of an extended mean value engine model for predicting the marine two-stroke engine operation at varying settings". In: *Energy* 143, pp. 533–545. ISSN: 03605442. DOI: 10.1016/j.energy.2017.10.138. URL: https://doi.org/10.1016/j.energy.2017.10.138.
- Theotokatos, Gerasimos and Vasileios Tzelepis (2015). "A computational study on the performance and emission parameters mapping of a ship propulsion system". In: Proceedings of the Institution of Mechanical Engineers Part M: Journal of Engineering for the Maritime Environment 229.1, pp. 58–76. ISSN: 20413084. DOI: 10.1177/1475090213498715.
- Tian, Zhigang, Tongdan Jin, Bairong Wu, and Fangfang Ding (May 2011). "Condition based maintenance optimization for wind power generation systems under continuous monitoring". In: *Renewable Energy* 36.5, pp. 1502–1509. ISSN: 09601481. DOI: 10.1016/j.renene.2010.10.028.
- Tixier, J., G. Dusserre, O. Salvi, and D. Gaston (2002). "Review of 62 risk analysis methodologies of industrial plants". In: *Journal of Loss Prevention in the*

Process Industries 15.4, pp. 291–303. ISSN: 09504230. DOI: 10.1016/S0950-4230(02)00008-6.

- Trachi, Youness, Elhoussin Elbouchikhi, Vincent Choqueuse, Mohamed El Hachemi Benbouzid, and Tianzhen Wang (2017). "A Novel Induction Machine Fault Detector Based on Hypothesis Testing". In: *IEEE Transactions on Industry Applications* 53.3, pp. 3039–3048. ISSN: 00939994. DOI: 10.1109/TIA. 2016.2625769.
- Tsitsilonis, Konstantinos Marios and Gerasimos Theotokatos (Dec. 2018). "A novel systematic methodology for ship propulsion engines energy management". In: Journal of Cleaner Production 204, pp. 212–236. ISSN: 09596526. DOI: 10.1016/j.jclepro.2018.08.154. URL: https://linkinghub. elsevier.com/retrieve/pii/S0959652618324934.
- UNCTAD (2017). *Review of Maritime Transport 2017*. Ed. by Deniz Barki and Lucy Délèze-Black. Geneva: United Nations, p. 114. ISBN: 9789211129229.
- (2018a). "50 Years of Review of Maritime Transport, 1968-2018: Reflecting on the past, exploring the future". In: 50 Years of Review of Maritime Transport, 1968-2018: Reflecting on the past, exploring the future 812, 86p.
- (2018b). Review of Maritime Transport 2018. Ed. by United Nations. 1st. New York: United Nations Publications, p. 116. ISBN: 978-92-1-047241-8. URL: https://unctad.org/en/PublicationsLibrary/rmt2018_en.pdf.
- UNFCCC (2014). Shipping, World Trade and the Reduction of CO 2 Emissions. Tech. rep. International Chamber of Shipping. URL: http://www. ics-shipping.org/docs/default-source/resources/policy-tools/ shipping-world-trade-and-the-reduction-of-co2-emissionsEE36BCFD2279. pdf?sfvrsn=20.
- Venkatesh, J (2003). An Introduction to Total Productive Maintenance (TPM). Tech. rep. 5. Como: The Plant Maintenance Resource Center, p. 327. DOI:

10.1038/nrg1090. URL: http://www.plant-maintenance.com/articles/
tpm_intro.shtml%20What.

- Verma, Ajit Kumar, Srividya Ajit, and Durga Rao Karanki (2010). Reliability and Safety Engineering: Second Edition. Ed. by Hoang Pham. 1st. Vol. 0. Springer Series in Reliability Engineering. London: Springer London, pp. 1– 571. ISBN: 9781447162698. DOI: 10.1007/978-1-4471-6269-8. URL: http: //link.springer.com/10.1007/978-1-84996-232-2.
- Vesely, Bill (2002). Fault Tree Analysis (FTA): Concepts and Applications. Washington D.C. URL: http://s3.amazonaws.com/academia.edu.documents/ 35382134/ftacourse.pdf?AWSAccessKeyId=AKIAIWOWYYGZ2Y53UL3A&Expires= 1489453586 & Signature = beeOzymfqTNQXIoigc81XOWTvYo % 3D & response content - disposition = inline % 3B % 20filename % 3DMission _ Success _ Starts_With_Safety_Faul.
- Vorkapic, Aleksandar, Predrag Kralj, and Dragan Martinovic (2017). "The analysis of the maintenance systems of a LPG carrier's liquefaction system main components". In: *Pomorstvo* 31.1, pp. 3–9. ISSN: 13320718. DOI: 10.31217/ p.31.1.2.
- Waeyenbergh, Geert and Liliane Pintelon (June 2002). "A framework for maintenance concept development". In: International Journal of Production Economics 77.3, pp. 299–313. ISSN: 09255273. DOI: 10.1016/S0925-5273(01) 00156-6.
- Wang, Hongzhou (2002). "A survey of maintenance policies of deteriorating systems". In: European Journal of Operational Research 139.3, pp. 469–489. ISSN: 12100552. DOI: 10.1007/springerreference{_}23399.
- Wang, Jinxin, Zhongwei Wang, Viacheslav Stetsyuk, Xiuzhen Ma, Fengshou Gu, and Wenhui Li (2019). "Exploiting Bayesian networks for fault isolation: A diagnostic case study of diesel fuel injection system". In: ISA Transactions 86,

pp. 276-286. ISSN: 00190578. DOI: 10.1016/j.isatra.2018.10.044. URL: https://doi.org/10.1016/j.isatra.2018.10.044.

- Wang, Wenbin, B. Hussin, and Tim Jefferis (2012). "A case study of condition based maintenance modelling based upon the oil analysis data of marine diesel engines using stochastic filtering". In: *International Journal of Production Economics* 136.1, pp. 84–92. ISSN: 09255273. DOI: 10.1016/j.ijpe.2011. 09.016. URL: http://dx.doi.org/10.1016/j.ijpe.2011.09.016.
- Wang, Y, C Deng, J Wu, Y Wang, and Y Xiong (2014). "A corrective maintenance scheme for engineering equipment". In: *Engineering Failure Analysis* 36, pp. 269–283. ISSN: 13506307. DOI: 10.1016/j.engfailanal.2013.10.006. URL: http://dx.doi.org/10.1016/j.engfailanal.2013.10.006.
- Wang, Zhanwei, Zhiwei Wang, Suowei He, Xiaowei Gu, and Zeng Feng Yan (2017).
 "Fault detection and diagnosis of chillers using Bayesian network merged distance rejection and multi-source non-sensor information". In: Applied Energy 188, pp. 200–214. ISSN: 03062619. DOI: 10.1016/j.apenergy.2016.11.130.
- Weimer, James, José Araujo, Mani Amoozadeh, Seyed Alireza Ahmadi, Henrik Sandberg, and Karl Henrik Johansson (2013). "Parameter-Invariant Actuator Fault Diagnostics in Cyber-Physical Systems with Application to Building Automation". In: Lecture Notes in Control and Information Sciences. Vol. 449 LNCIS. Springer Verlag, pp. 179–196. ISBN: 9783319011585. DOI: 10.1007/ 978-3-319-01159-2{_}10.
- White, Ian R., Patrick Royston, and Angela M. Wood (2011). "Multiple imputation using chained equations: Issues and guidance for practice". In: *Statistics* in Medicine 30.4, pp. 377–399. ISSN: 02776715. DOI: 10.1002/sim.4067.
- Whiteley, Michael, Sarah Dunnett, and Lisa Jackson (2016). "Failure Mode and Effect Analysis, and Fault Tree Analysis of Polymer Electrolyte Membrane Fuel Cells". In: International Journal of Hydrogen Energy 41.2, pp. 1187–

1202. ISSN: 03603199. DOI: 10.1016/j.ijhydene.2015.11.007. URL: http: //dx.doi.org/10.1016/j.ijhydene.2015.11.007.

- Wiggelinkhuizen, Edwin, Theo Verbruggen, Henk Braam, Luc Rademakers, Jianping Xiang, and Simon Watson (2008). "Assessment of condition monitoring techniques for offshore wind farms". In: Journal of Solar Energy Engineering, Transactions of the ASME 130.3, pp. 0310041–0310049. ISSN: 01996231. DOI: 10.1115/1.2931512.
- Woodyard, Doug (2009). Manire Diesel Engines and Gas Turbines. Ed. by Butterworth-Heineman. 9th. Oxford: Elsevier, p. 905. ISBN: 9780750689847.
- Wu, Junjie (2012). "Advances in K-means Clustering". In: Advances in K-means Clustering: A Data Mining Thinking, pp. 155–175. ISSN: 2190-5053. DOI: 10. 1007/978-3-642-29807-3. URL: http://link.springer.com/book/10. 1007%252F978-3-642-29807-3%0Ahttp://link.springer.com/book/10. 1007/978-3-642-29807-3.
- Xu, Wilsun, Iraj Rahimi Pordanjani, Yunfei Wang, and Ebrahim Vaahedi (2012).
 "A network decoupling transform for phasor data based voltage stability analysis and monitoring". In: *IEEE Transactions on Smart Grid* 3.1, pp. 261–270.
 ISSN: 19493053. DOI: 10.1109/TSG.2011.2163175.
- Yang, Li, Zhi sheng Ye, Chi Guhn Lee, Su fen Yang, and Rui Peng (2019). "A twophase preventive maintenance policy considering imperfect repair and postponed replacement". In: *European Journal of Operational Research* 274.3, pp. 966–977. ISSN: 03772217. DOI: 10.1016/j.ejor.2018.10.049. URL: https://doi.org/10.1016/j.ejor.2018.10.049.
- Yazdi, Mohammad, Farzaneh Nikfar, and Mahnaz Nasrabadi (2017). "Failure probability analysis by employing fuzzy fault tree analysis". In: International Journal of Systems Assurance Engineering and Management 8.s2, pp. 1177– 1193. ISSN: 09764348. DOI: 10.1007/s13198-017-0583-y.

- Yu, Shyr Shen, Shao Wei Chu, Chuin Mu Wang, Yung Kuan Chan, and Ting Cheng Chang (2018). "Two improved k-means algorithms". In: Applied Soft Computing Journal 68, pp. 747–755. ISSN: 15684946. DOI: 10.1016/j.asoc. 2017.08.032. URL: http://dx.doi.org/10.1016/j.asoc.2017.08.032.
- Yuan, Zhi, Nima Khakzad, Faisal Khan, and Paul Amyotte (2015). "Risk analysis of dust explosion scenarios using bayesian networks". In: *Risk Analysis* 35.2, pp. 278–291. ISSN: 15396924. DOI: 10.1111/risa.12283.
- Zaher, A. S. and S. D.J. McArthur (2007). "A multi-agent fault detection system for wind turbine defect recognition and diagnosis". In: 2007 IEEE Lausanne POWERTECH, Proceedings, pp. 22–27. DOI: 10.1109/PCT.2007.4538286.
- Zaher, A., S. D.J. McArthur, D. G. Infield, and Y. Patel (2009). "Online wind turbine fault detection through automated SCADA data analysis". In: Wind Energy 12.6, pp. 574–593. ISSN: 10954244. DOI: 10.1002/we.319.
- Zhan, Yulong, Zhu Bin Shi, Theingi Shwe, and Xiao Zhong Wang (2007). "Fault diagnosis of marine main engine cylinder cover based on vibration signal". In: *Proceedings of the Sixth International Conference on Machine Learning and Cybernetics, ICMLC 2007.* Vol. 2. IEEE, pp. 1126–1130. ISBN: 142440973X. DOI: 10.1109/ICMLC.2007.4370313. URL: http://ieeexplore.ieee.org/ document/4370313/.
- Zhan, Yulong, Zhubin Shi, and Mlngming Liu (2007). "The application of support vector machines (SVM) to fault diagnosis of marine main engine cylinder cover". In: *IECON Proceedings (Industrial Electronics Conference)*. IEEE, pp. 3018–3022. ISBN: 1424407834. DOI: 10.1109/IECON.2007.4459891. URL: http://ieeexplore.ieee.org/document/4459891/.
- Zhang, Guizhen and Vinh V. Thai (2016). "Expert elicitation and Bayesian Network modeling for shipping accidents: A literature review". In: Safety Science 87, pp. 53–62. ISSN: 18791042. DOI: 10.1016/j.ssci.2016.03.019. URL: http://dx.doi.org/10.1016/j.ssci.2016.03.019.

- Zhang, Min-Ling and Zhi-Hua Zhou (2007). "ML-KNN: A lazy learning approach to multi-label learning". In: *Pattern Recognition* 40, pp. 2038–2048. DOI: 10. 1016/j.patcog.2006.12.019.
- Zhang, Shichao (2012). "Nearest neighbor selection for iteratively kNN imputation". In: Journal of Systems and Software 85.11, pp. 2541-2552. ISSN: 01641212. DOI: 10.1016/j.jss.2012.05.073. URL: http://dx.doi. org/10.1016/j.jss.2012.05.073.
- Zhang, Shichao, Debo Cheng, Zhenyun Deng, Ming Zong, and Xuelian Deng (2018). "A novel kNN algorithm with data-driven k parameter computation". In: *Pattern Recognition Letters* 109, pp. 44–54. ISSN: 01678655. DOI: 10.1016/ j.patrec.2017.09.036. URL: https://doi.org/10.1016/j.patrec.2017. 09.036.
- Zhang, W and W Wang (2014). "Cost modelling in maintenance strategy optimisation for infrastructure assets with limited data". In: *Reliability Engineering* and System Safety 130, pp. 33–41. ISSN: 09518320. DOI: 10.1016/j.ress. 2014.04.025.
- Zhao, Yang, Jin Wen, and Shengwei Wang (July 2015). "Diagnostic Bayesian networks for diagnosing air handling units faults Part II: Faults in coils and sensors". In: Applied Thermal Engineering 90, pp. 145–157. ISSN: 13594311. DOI: 10.1016/j.applthermaleng.2015.07.001.
- Zhao, Yang, Jin Wen, Fu Xiao, Xuebin Yang, and Shengwei Wang (Jan. 2017).
 "Diagnostic Bayesian networks for diagnosing air handling units faults part I: Faults in dampers, fans, filters and sensors". In: *Applied Thermal Engineering* 111, pp. 1272–1286. ISSN: 13594311. DOI: 10.1016/j.applthermaleng.2015. 09.121.
- Zhou, Xiaojun, Lifeng Xi, and Jay Lee (Apr. 2007). "Reliability-centered predictive maintenance scheduling for a continuously monitored system subject to

degradation". In: *Reliability Engineering and System Safety* 92.4, pp. 530–534. ISSN: 09518320. DOI: 10.1016/j.ress.2006.01.006.

Appendix A

FTA Gates and Events

A.1 FTA Gates

AND Gate

The AND gate is used to indicate that the output occurs if and only if all the input events occur. The output of an AND gate can be the top event or any intermediate event. The input events can be basic events, intermediate events (outputs of other gates) or a combination of both (PTC Windchill 2019).

OR Gate

The OR gate is used to indicate that the output occurs if and only if at least one of the input events occur. The output of an OR gate can be the top event or any intermediate event. The input events can be basic events, intermediate events or a combination of both. There should be at least two inputs to an OR gate (PTC Windchill 2019).

Voting Gate

The Voting (M/n) gate is used to indicate that the output occurs if and only if M

out of the n input events occur. The output occurs when at least m input events occur. When M = 1, the Voting gate behaves like an OR gate. The output of a Voting gate can be a top event or an intermediate event. The input events can be basic events, intermediate events or a combination of both (PTC Windchill 2019).

Inhibit Gate

The Inhibit gate is used to indicate that the output occurs when the input events (l1 and l2) occur and the input condition (C) is satisfied. The output of an Inhibit gate can be a top event or an intermediate event. The input events can be basic events, intermediate events or a combination of both (PTC Windchill 2019).

Exclusive OR Gate

The Exclusive OR (XOR) gate is used to indicate that the output occurs if and only if one of the two input events occurs and the other input event does not occur. An XOR gate can only have two inputs. The output of an Exclusive OR gate can be the top event or an intermediate event. The input events can be basic events, intermediate events or a combination of both. The presence of an XOR gate may give rise to non-coherent trees, where the non-occurrence of an event causes the top event to occur (PTC Windchill 2019).

NOT Gate

The NOT gate is used to indicate that the output occurs when the input event does not occur. The presence of a NOT gate may give rise to non-coherent trees, where the non-occurrence of an event causes the top event to occur. There is only one input to a NOT gate (PTC Windchill 2019).

NOR Gate

The NOR gate functions like a combination of an OR gate and a NOT gate. The NOR gate is used to indicate that the output occurs when all the input events are absent. The output of a NOR gate can be the top event or an intermediate event. The input events can be basic events, intermediate events or a combination of both. The presence of a NOR gate may give rise to non-coherent trees, where the lack of one or more events causes the top event to occur (PTC Windchill 2019).

NAND Gate

The NAND gate functions like a combination of an AND gate and a NOT gate. The NAND gate is used to indicate that the output occurs when at least one of the input events is absent. The output of a NAND gate can be the top event or an intermediate event. The input events can be basic events, intermediate events or a combination of both. The presence of a NAND gate may give rise to non-coherent trees, where the non-occurrence of an event causes the top event to occur (PTC Windchill 2019).

Priority AND Gate

The Priority AND (PAND) gate is used to indicate that the output occurs if and only if all input events occur in a particular order. The order is the same as that in which the inputs events are connected to the PAND gate from left to right. The PAND gate is a dynamic gate, which means that the order of the occurrence of input events is important to determining the output.

The output of a PAND gate can be the top event or an intermediate event. The inputs can be basic events or outputs of any AND gate, OR gate, or dynamic gate, which includes the SPARE gate, PAND gate, sequence-enforcing (SEQ) gate and functional dependency (FDEP) gate. (These gates should have the inputs from basic events or other AND gates and OR gates.) The items that enter a PAND gate need to fail in temporal order from left to right to trigger the event. The PAND gate also supports a single input. When only a single input exists, then occurrence of that input will trigger the event (PTC Windchill 2019).

Functional Dependency Gate

The Functional Dependency (FDEP) gate is used to indicate that all dependent basic events are forced to occur whenever the trigger event occurs. The separate occurrence of any of the dependent basic events has no effect on the trigger event. The FDEP gate has one trigger event and can have one or more dependent events. All dependent events are either basic events or spare events. The trigger event can be a terminal event or output of any AND gate, OR gate or dynamic gate, which includes the SPARE gate, PAND gate, Sequence-Enforcing gate (SEQ) and FDEP gate.

Dependent events are repeated events that are present in other parts of the fault tree. The FDEP gate is a dynamic gate, which means the temporal order of the occurrence of events is important to analyse this gate. Generally, the output of the FDEP gate is not that important; however, it is equivalent to the status of its trigger event.

The FDEP gate can also be used to set the priorities for SPARE gates. For example, if multiple spares are connected to a FDEP gate, after the occurrence of the trigger event, all spares that are connected to the FDEP gate will fail. Upon failure of these spares, the next available good spares in those SPARE gates will replace the failed spares. If there exists a conflict in choosing the next available spare between multiple SPARE gates, the priority will be based on the order of the connection of these spares in the FDEP gate from left to right (PTC Windchill 2019).

Sequence Enforcing Gate

The Sequence-Enforcing (SEQ) gate forces events to occur in a particular order.

The input events are constrained to occur in the left-to-right order in which they appear under the gate. That means that the left-most event must occur before the event on its immediate right, which must occur before the event on its immediate right is allowed to occur. The SEQ gate is used to indicate that the output occurs if and only if all input events occurs, when the input events are constraint to occur in a particular order.

The SEQ gate is a dynamic gate, which means the occurrence of the inputs follows a sequential order. In other words, an event connected to a SEQ gate will be initiated immediately after occurrence of its immediate left event. Therefore, if the left-most input is a basic event, then the SEQ gate works like a cold SPARE gate. The SEQ gate can be contrasted with the PAND gate in that the PAND gate detects whether events occur in a particular order (but the events can occur in any order), whereas the SEQ gate allows the events to occur only in the specified order. The first input (left-most input) to a SEQ gate can be a terminal event or outputs of any AND gate, OR gate or dynamic gate, which includes the SPARE gate, PAND gate, FDEP gate or SEQ gate). Only basic events are allowed for all other inputs (PTC Windchill 2019).

SPARE Gate

The SPARE gate is used to model the behaviour of spares in the system. The SPARE gate is used to indicate that the output occurs if and only if all input spare events occur. All inputs of a SPARE gate are spare events. A SPARE gate can have multiple inputs. The first event (left-most event) is known as the primary input, and all other inputs are known as alternative inputs. The primary event is the one that is initially powered on, and the alternative inputs and are initially in standby mode.

After a failure, the active/powered unit that is the first available spare from left to right will be chosen to be active. If all units are failed, then the spare will be considered as failed (output occurred). Depending on the dormancy factor of spares, spares can fail even in standby mode.

If the dormancy factor of all spares connected to a SPARE gate are 0, then the spare acts like a cold spare. If the dormancy factor of all spares connected to a SPARE gate is 1, then the spare acts like a hot spare. If the dormancy factor of all spares connected to a SPARE gate are the same (and are between 0 and 1), then the spare acts like a warm spare. If the dormancy factors of its inputs are different, then it handles generalised situations. The SPARE gate is a dynamic gate, which means the temporal order of the occurrence of events is important to analyse this gate (PTC Windchill 2019).

A.2 FTA Events

Basic Event

A basic event is either a component level event that is not further resolved. A basic event is at the lowest level in a tree branch and terminates a fault tree path. Component level events can include hardware or software failures, human errors and sub-system failures (PTC Windchill 2019).

House Event

A house event is used to represent an event that is normally expected to occur. A house event can be turned on or off. When a house event is turned on (TRUE), that event is presumed to have occurred, and the probability of that event is set to 1. When a house event is turned off (FALSE), that event is presumed not to have occurred, and the probability is set to 0. House events are useful in making parts of a fault tree functional or non-functional. House events are also referred to as trigger events or switching events (PTC Windchill 2019).

House Event

A conditional event is used to indicate specific conditions or restrictions that apply to any logic gate, although they are most often used with Inhibit gates (PTC Windchill 2019).

Undeveloped Event

An undeveloped event is used if further resolution of that event does not improve the understanding of the problem or if further resolution is not necessary for proper evaluation of the fault tree. It is similar to a basic event, but is shown as a different symbol to signify that it could be developed further but that the analysis has not yet been done or need not be done for the sake of the analysis in question. Undeveloped events may changed to some other event type and broken down into associated gates and events if it is later deemed necessary (PTC Windchill 2019).

Spare Event

A spare event is used to specify spares in dynamic fault trees. Spare events are similar to basic events in functionality; however, they allow only rates as inputs. The dormancy factor of the spare indicates the ratio of failure rate in the spare mode and the failure rate in the operational mode. Spare events can have a spares pool, which represents the number of identical instances of that event. For example, if a spares pool of an event is two, there are two identical spare components of that spare event. Spare events are restricted to use as either spares to SPARE gates or as dependent events to Functional Dependency gates (PTC Windchill 2019).

Appendix B

Ship Components Cost

B.1 Costs

Component	Cost (US Dollars)
BWTS	90000
CO2 System	90000
Central Cooler 1	30000
Central Cooler 2	30000
Central Cooler 3	30000
High Duty Compressor 1	17000
High Duty Compressor 2	17000
Auxiliary Cooler 1	15000
Auxiliary Cooler 2	15000
Cargo Pumps	15000
FO Purifier 1	15000
FO Purifier 2	15000
FO Purifier 3	15000

Table B.1: Repair and replacement costs for the components of vessel "A"

Continued on next page

Component	Cost (US Dollars)
Low Duty Compressor 1	15000
Low Duty Compressor 2	15000
Ballast Pump 1	15000
Ballast Pump 2	15000
Ballast Pump 3	15000
LO Purifier 1	14000
LO Purifier 2	14000
LO Purifier 3	14000
LO Purifier 4	14000
Bilge Pumps	12000
Steering Gear Cylinder Ram	12000
Dry Powder System	12000
Fire Pump 1	12000
Fire Pump 2	12000
MGE LO Pump 1	10000
MGE LO Pump 2	10000
MGE LO Pump 3	10000
MGE LO Pump 4	10000
Boiler Burner 1	8000
Boiler Burner 2	8000
IGG Burner	8000
Gas Combustion Unit	8000
MGE 1 LO Pump	6500
MGE 2 LO Pump	6500
MGE 3 LO Pumps	6500

Table B.1 – Continued from previous page

Component	Cost (US Dollars)
MGE 4 Lubricating Pumps	6500
Blower 1	6000
Blower 2	6000
MGE 1 Cyl. Head	5500
MGE 2 Cylinder Head	8000
MGE 3 Cylinder Head	8000
MGE 4 Cylinder Head	12000
Boiler Feedwater Pump 1	5000
Boiler Feedwater Pump 2	5000
FO Feed Pump 1	5000
FO Feed Pump 2	5000
FO Feed Pump 3	5000
FO Feed Pump 4	5000
High Duty Heater	5000
LNG Vaporiser	5000
Reduction Gear LO Pump 1	5000
Reduction Gear LO Pump 2	5000
FO Circulating Pump 1	4000
FO Circulating Pump 2	4000
FO Circulating Pump 3	4000
FO Circulating Pump 4	4000
Forcing Vaporiser	4000
Fuel Gas Pump 1	4000
Fuel Gas Pump 2	4000
IGG Fuel Pump	4000

Table B.1 – Continued from previous page

Component	Cost (US Dollars)
HFO Transfer Pump 1	4000
HFO Transfer Pump 2	4000
MGE 1 Coolers	4000
MGE 3 Coolers	4000
MGE 4 Coolers	4000
MGE 2 Coolers	4000
Pilot Fuel Feed Pump 1	4000
Pilot Fuel Feed Pump 2	4000
MGE 1 Rod	3500
MGE 2 Rod	3500
MGE 3 Rod	3500
MGE 4 Rod	3500
FW Booster Pump 1	3000
FW Booster Pump 2	3000
LO Transfer Pump 1	3000
LO Transfer Pump 2	3000
MGE 1 Cylinder Liner	3000
MGE 1 Piston	3000
MGE 2 Cylinder Liner	3500
MGE 2 Piston	3000
MGE 3 Cylinder Liner	3500
MGE 3 Pistons	3000
MGE 4 Cylinder Liner	3000
MGE 4 Piston	3000
SWC Pump 1	3000

Table B.1 – Continued from previous page

Component	Cost (US Dollars)
SWC Pump 2	3000
SWC Pump 3	3000
Boiler Water Circulating Pumps	3000
Aux SWC Pump 1	2500
Aux SWC Pump 2	2500
Boiler Fuel Pump 1	2500
Boiler Fuel Pump 2	2500
FWC Pump 1	2500
FWC Pump 2	2500
LO Purifier Supply Pump 1	2500
LO Purifier Supply Pump 2	2500
LO Purifier Supply Pump 3	2500
LO Purifier Supply Pump 4	2500
MDO Transfer Pump	2500
MGE 1 FO Injection Pump	2500
MGE 2 FO Injection Pump	2500
MGE 3 FO Injection Pump	2500
MGE 4 FO Injection Pump	2500
MGO Transfer Pump	2500
Steering Gear Oil Pump	2500
Spray Pump 1	2500
Spray Pump 2	2500
Spray Pump 3	2500
Spray Pump 4	2500
Steering Gear Control And Repeat Back Leaver	2000

Table B.1 – Continued from previous page

Continued on next page

Component	Cost (US Dollars)
FO Purifier Supply Pump 1	2000
FO Purifier Supply Pump 2	2000
FO Purifier Supply Pump 3	2000
Steering Gear Remote Controller	2000
Steering Gear Angle Transmitter	1500
IGG Safety Equipment	1000
Fuel Injector 1	300
Fuel Injector 2	300
MGE 1 Main Bearing	200
MGE 2 Main Bearing	200
MGE 3 Main Bearing	200
MGE 4 Main Bearing	200
Boiler Refractory Area 1	200
Boiler Refractory Area 2	200
MGE 1 FO Injector	150
MGE 2 FO Injector	300
MGE 3 FO Injector	300
MGE 4 FO Injector	150

Table B.1 – Continued from previous page

Appendix C

Critical Equipment Selection

C.1 MTBFs

Inert Gas Generation	
Component	MTBF (Hours)
Blower	30240
Fuel Pump	12264
Burner	6132
Safety Equipment	15120

Table C.1: Table of MTBFs used for the inert gas generation system

Table C.2: Table of MTBFs used for the cargo equipment system

Cargo Equipment	
BF (Hours)	
1540	
1540	
1512	
2190	
2190	

Bilge, Fire and Ballast	
Component	MTBF (Hours)
Ballast Pumps	2190
BWTS	6048
Fire Pumps	2190
Bilge Pumps	2190
CO2 System	6132
Dry Powder System	6132

Table C.3: Table of MTBFs used for the bilge, fire and ballast systems

Table C.4: Table of MTBFs used for the steering gear system

Steering Gear	
Component	MTBF (Hours)
Cylinder Ram	61320
Oil Pump	12264
Control and Repeat Back Lever	30660
Angle Transmitter	18396
Remote Controller	18396

Table C.5: Table of MTBFs used for the steam generation system

Steam Generation	
Component	MTBF (Hours)
Pilot Fuel Injector	3100
Fuel Pump	3100
Water Circulating Pump	3000
Feed-water Pump	3000
Burner	6132
Refractory Area	1226

Main Generating Engines	
Component	MTBF (Hours)
Piston Rod	61320
Cylinder Head	61320
Fuel Oil Injector	12264
Main Bearing	61320
Cylinder Liner	61320
Piston	30660
Cooling Pumps	3000
Coolers	2190
Lubricating Pumps	12264
Fuel Oil Injection Pumps	2200

Table C.6: Table of MTBFs used for the main generating engines system

Table C.7: Table of MTBFs used for the gas fuel system

Gas Fuel	
Component	MTBF (Hours)
Fuel Gas Pumps	3100
Low Duty Compressor	2190
Forcing Vaporiser	1512
Gas Combustion Unit	12264

Table C.8: Table of MTBFs used for the fuel oil feed system

Fuel Oil Feed	
Component	MTBF (Hours)
Fuel Oil Feed Pump	2200
Fuel Oil Circulating Pump	2200
Pilot Fuel Feed Pump	3100

Table C.9: Table of MTBFs used for the fuel oil purification system

Fuel Oil Purification	
Component	MTBF (Hours)
Fuel Oil Purifiers	8400
Fuel Oil Purifiers Supply Pump	2200

Fuel Oil Transfer		
Component MTBF (Hours)		
Heavy Fuel Oil Transfer Pumps	2200	
Marine Diesel Oil Transfer	3100	
Pumps		
Marine Gas Oil Transfer Pumps	3100	

Table C.10: Table of MTBFs used for the fuel oil transfer system

Table C.11: Table of MTBFs used for the lubricating oil purification system

Lubricating Oil Purification		
Component	MTBF (Hours)	
Lubricating Oil Purifiers	12000	
Lubricating Oil Purifier Supply	12264	
Pumps		

Table C.12: Table of MTBFs used for the lubricating oil service system

Lubricating Oil Service		
Component	MTBF (Hours)	
Lubricating Oil Transfer Pumps	12264	
Reduction Gear Lubricating Oil	12264	
Pumps		
MGE Lubricating Oil Pumps	12264	

Table C.13: Table of MTBFs used for the central MGE cooling system

Central MGE Cooling	
Component	MTBF (Hours)
Sea Water Cooling Pumps	2190
Central Cooler	2190

Table C.14: Table of MTBFs used for the auxiliary cooling system

Auxiliary Cooling	
Component	MTBF (Hours)
Fresh Water Cooling Pumps	3000
Sea Water Cooling Pumps	2190
Fresh Water Booster Pumps	3000
Auxiliary Cooler	2190

C.2 IMs

Table C.15: Obtained I^B for the components of vessel "A"

Component	I^B
MGE 4 Cylinder Head	3.0307
MGE 4 Cylinder Liner	3.0307
MGE 4 Main Bearing	3.0307
MGE 4 Piston	3.0307
MGE 4 Rod	3.0307
MGE 1 FO Injection Pump	2.8700
MGE 2 FO Injection Pump	2.8700
MGE 3 FO Injection Pump	2.8700
MGE 4 FO Injection Pump	2.8700
MGE 1 FO Injector	2.8000
MGE 2 FO Injector	2.8000
MGE 3 FO Injector	2.8000
MGE 4 FO Injector	2.8000
FO Feed Pump 1	2.7061
FO Feed Pump 2	2.7061
FO Feed Pump 3	2.7061
FO Feed Pump 4	2.7061
MGE 1 Cyl. Head	2.7061
MGE 1 Cylinder Liner	2.7061
MGE 1 Main Bearing	2.7061
MGE 1 Piston	2.7061
MGE 1 Rod	2.7061
MGE 2 Cylinder Head	2.7061

Component	I^B
MGE 2 Cylinder Liner	2.7061
MGE 2 Main Bearing	2.7061
MGE 2 Piston	2.7061
MGE 2 Rod	2.7061
MGE 3 Cylinder Head	2.7061
MGE 3 Cylinder Liner	2.7061
MGE 3 Main Bearing	2.7061
MGE 3 Pistons	2.7061
MGE 3 Rod	2.7061
MGE 1 LO Pump	2.0200
MGE 2 LO Pump	2.0200
MGE 3 LO Pumps	2.0200
MGE 4 Lubricating Pumps	2.0200
IGG Burner	1.8525
IGG Fuel Pump	1.8525
IGG Safety Equipment	1.8525
Steering Gear Control And Repeat Back Leaver	1.0000
Steering Gear Cylinder Ram	1.0000
Boiler Feedwater Pump 1	1.0000
Boiler Feedwater Pump 2	1.0000
Steering Gear Oil Pump	1.0000
Boiler Water Circulating Pumps	1.0000
BWTS	0.8750
Cargo Pumps	0.6318
High Duty Heater	0.6318

Table C.15 – Continued from previous page

Component	I^B
LNG Vaporiser	0.6318
Bilge Pumps	0.5392
CO2 System	0.3296
Dry Powder System	0.3296
Central Cooler 1	0.2661
Central Cooler 2	0.2661
Central Cooler 3	0.2661
SWC Pump 1	0.2661
SWC Pump 2	0.2661
SWC Pump 3	0.2661
High Duty Compressor 1	0.2316
High Duty Compressor 2	0.2316
Boiler Burner 1	0.2288
Boiler Burner 2	0.2288
Boiler Refractory Area 1	0.2288
Boiler Refractory Area 2	0.2288
FO Purifier Supply Pump 1	0.2198
FO Purifier Supply Pump 2	0.2198
FO Purifier Supply Pump 3	0.2198
Aux SWC Pump 1	0.2142
Aux SWC Pump 2	0.2142
Auxiliary Cooler 1	0.2142
Auxiliary Cooler 2	0.2142
FW Booster Pump 1	0.1960
FW Booster Pump 2	0.1960

Table C.15 – Continued from previous page

Component	I^B
FWC Pump 1	0.1960
FWC Pump 2	0.1960
LO Transfer Pump 1	0.1671
LO Transfer Pump 2	0.1671
Reduction Gear LO Pump 1	0.1671
Reduction Gear LO Pump 2	0.1671
Forcing Vaporiser	0.1327
Gas Combustion Unit	0.1327
Fire Pump 1	0.1277
Fire Pump 2	0.1277
Ballast Pump 1	0.1176
Ballast Pump 2	0.1176
Ballast Pump 3	0.1176
Spray Pump 1	0.0688
Spray Pump 2	0.0688
Spray Pump 3	0.0688
Spray Pump 4	0.0688
MDO Transfer Pump	0.0637
MGO Transfer Pump	0.0637
Blower 1	0.0603
Blower 2	0.0603
Steering Gear Angle Transmitter	0.0529
Steering Gear Remote Controller	0.0529
Low Duty Compressor 1	0.0487
Low Duty Compressor 2	0.0487

Table C.15 – Continued from previous page

Component	I^B
HFO Transfer Pump 1	0.0481
HFO Transfer Pump 2	0.0481
Fuel Gas Pump 1	0.0366
Fuel Gas Pump 2	0.0366
Boiler Fuel Pump 1	0.0210
Boiler Fuel Pump 2	0.0210
Fuel Injector 1	0.0210
Fuel Injector 2	0.0210
FO Purifier 1	0.0208
FO Purifier 2	0.0208
FO Purifier 3	0.0208
Pilot Fuel Feed Pump 1	0.0171
Pilot Fuel Feed Pump 2	0.0171
FO Circulating Pump 1	0.0080
FO Circulating Pump 2	0.0080
FO Circulating Pump 3	0.0080
FO Circulating Pump 4	0.0080
MGE 1 Coolers	0.0050
MGE 3 Coolers	0.0050
MGE 4 Coolers	0.0050
MGE 2 Coolers	0.0050
LO Purifier 1	0.0011
LO Purifier 2	0.0011
LO Purifier 3	0.0011
LO Purifier 4	0.0011

Table C.15 – Continued from previous page

Component	I^B
LO Purifier Supply Pump 1	0.0010
LO Purifier Supply Pump 2	0.0010
LO Purifier Supply Pump 3	0.0010
LO Purifier Supply Pump 4	0.0010
MGE LO Pump 1	0.0010
MGE LO Pump 2	0.0010
MGE LO Pump 3	0.0010
MGE LO Pump 4	0.0010

Table C.15 – *Continued from previous page*

Table C.16: Obtained I^{CR} for the components of vessel "A"

Component	I^{CR}
MGE 1 FO Injection Pump	0.1800
MGE 2 FO Injection Pump	0.1800
MGE 3 FO Injection Pump	0.1800
MGE 4 FO Injection Pump	0.1800
MGE 1 FO Injector	0.1500
MGE 2 FO Injector	0.1500
MGE 3 FO Injector	0.1500
MGE 4 FO Injector	0.1500
MGE 1 LO Pump	0.1300
MGE 2 LO Pump	0.1300
MGE 3 LO Pumps	0.1300
MGE 4 Lubricating Pumps	0.1300

Component	I^{ICR}
FO Feed Pump 1	0.0874
FO Feed Pump 2	0.0874
FO Feed Pump 3	0.0874
FO Feed Pump 4	0.0874
LNG Vaporiser	0.0545
Cargo Pumps	0.0538
Boiler Feedwater Pump 1	0.0505
Boiler Feedwater Pump 2	0.0505
Boiler Water Circulating Pumps	0.0505
IGG Burner	0.0497
Steering Gear Cylinder Ram	0.0475
High Duty Heater	0.0413
Bilge Pumps	0.0352
Steering Gear Oil Pump	0.0290
IGG Fuel Pump	0.0259
BWTS	0.0238
IGG Safety Equipment	0.0211
Central Cooler 1	0.0174
Central Cooler 2	0.0174
Central Cooler 3	0.0174
SWC Pump 1	0.0174
SWC Pump 2	0.0174
SWC Pump 3	0.0174
MGE 4 Cylinder Liner	0.0173
MGE 1 Piston	0.0155

Table C.16 – Continued from previous page

Continued on next page

Component	I^{ICR}
MGE 2 Cylinder Liner	0.0155
MGE 3 Cylinder Liner	0.0155
High Duty Compressor 1	0.0151
High Duty Compressor 2	0.0151
FO Purifier Supply Pump 1	0.0143
FO Purifier Supply Pump 2	0.0143
FO Purifier Supply Pump 3	0.0143
Aux SWC Pump 1	0.0140
Aux SWC Pump 2	0.0140
Auxiliary Cooler 1	0.0140
Auxiliary Cooler 2	0.0140
Forcing Vaporiser	0.0115
FW Booster Pump 1	0.0099
FW Booster Pump 2	0.0099
FWC Pump 1	0.0099
FWC Pump 2	0.0099
CO2 System	0.0088
Dry Powder System	0.0088
MGE 4 Cylinder Head	0.0087
MGE 4 Main Bearing	0.0087
MGE 4 Piston	0.0087
MGE 4 Rod	0.0087
Fire Pump 1	0.0083
Fire Pump 2	0.0083
MGE 1 Cyl. Head	0.0078

Table C.16 – Continued from previous page

Component	I^{ICR}
MGE 1 Cylinder Liner	0.0078
MGE 1 Main Bearing	0.0078
MGE 1 Rod	0.0078
MGE 2 Cylinder Head	0.0078
MGE 2 Main Bearing	0.0078
MGE 2 Piston	0.0078
MGE 2 Rod	0.0078
MGE 3 Cylinder Head	0.0078
MGE 3 Main Bearing	0.0078
MGE 3 Pistons	0.0078
MGE 3 Rod	0.0078
Ballast Pump 1	0.0077
Ballast Pump 2	0.0077
Ballast Pump 3	0.0077
MGE 1 Coolers	0.0062
MGE 3 Coolers	0.0062
MGE 4 Coolers	0.0062
MGE 2 Coolers	0.0062
Boiler Burner 1	0.0061
Boiler Burner 2	0.0061
Spray Pump 1	0.0059
Spray Pump 2	0.0059
Spray Pump 3	0.0059
Spray Pump 4	0.0059
Steering Gear Control And Repeat Back Leaver	0.0050
Continued on next page	

Table C.16 – Continued from previous page

Component	I^{ICR}
Boiler Refractory Area 1	0.0032
Boiler Refractory Area 2	0.0032
Low Duty Compressor 1	0.0032
Low Duty Compressor 2	0.0032
HFO Transfer Pump 1	0.0031
HFO Transfer Pump 2	0.0031
MDO Transfer Pump	0.0031
MGO Transfer Pump	0.0031
LO Transfer Pump 1	0.0023
LO Transfer Pump 2	0.0023
Reduction Gear LO Pump 1	0.0023
Reduction Gear LO Pump 2	0.0023
Gas Combustion Unit	0.0019
Fuel Gas Pump 1	0.0018
Fuel Gas Pump 2	0.0018
Boiler Fuel Pump 1	0.0010
Boiler Fuel Pump 2	0.0010
Fuel Injector 1	0.0010
Fuel Injector 2	0.0010
Pilot Fuel Feed Pump 1	0.0008
Pilot Fuel Feed Pump 2	0.0008
FO Circulating Pump 1	0.0005
FO Circulating Pump 2	0.0005
FO Circulating Pump 3	0.0005
FO Circulating Pump 4	0.0005

Table C.16 – Continued from previous page
Component	I^{ICR}
Steering Gear Angle Transmitter	0.0005
Steering Gear Remote Controller	0.0005
FO Purifier 1	0.0004
FO Purifier 2	0.0004
FO Purifier 3	0.0004
Blower 1	0.0003
Blower 2	0.0003
LO Purifier 1	0.0000
LO Purifier 2	0.0000
LO Purifier 3	0.0000
LO Purifier 4	0.0000
LO Purifier Supply Pump 1	0.0000
LO Purifier Supply Pump 2	0.0000
LO Purifier Supply Pump 3	0.0000
LO Purifier Supply Pump 4	0.0000
MGE LO Pump 1	0.0000
MGE LO Pump 2	0.0000
MGE LO Pump 3	0.0000
MGE LO Pump 4	0.0000

Table C.16 - Continued from previous page

Appendix D

Fault Detection

D.1 Data Checking Results



Figure D.1: Data checking result for ME scavenging air temperature



Figure D.2: Data checking result for ME scavenging air pressure



Figure D.3: Data checking result for ME cylinder 1 EG temperature



Figure D.4: Data checking result for ME cylinder 2 EG temperature



Figure D.5: Data checking result for ME cylinder 3 EG temperature



Figure D.6: Data checking result for ME cylinder 4 EG temperature



Figure D.7: Data checking result for ME cylinder 5 EG temperature